Investigating the relationship between changes in physical activity levels and injury risk among children and adolescents

Chinchin Wang, MSc

Department of Epidemiology, Biostatistics, and Occupational Health

Faculty of Medicine

McGill University

Montréal, Québec, Canada

August 2024

A thesis submitted to McGill University in partial fulfillment of the requirements of the degree of Doctorate of Philosophy in Epidemiology

© Chinchin Wang, 2024

Table of Contents

Abstract	v
Resumé	vi
Acknowledgements	viii
Contribution to Original Knowledge	x
Contribution of Authors	xi
List of Tables	xiii
List of Figures	xiv
List of Abbreviations and Acronyms	xvi
Chapter 1: Introduction	
Organization of Thesis	
Chapter 2: Literature Review	
Preface	
Measurement of load and physical activity	
A note on terminology	
Internal and external load	
Measures of load in epidemiology	
Quantifying the relationship between physical activity and injury	
Absolute activity and injury risk	
The fitness-fatigue model	
The acute:chronic workload ratio	
Variations of the ACWR	
Current state of evidence	
Chapter 3: Overview of Methods	
Preface	

Measures of injury incidence	
Injury effect measures	
Modelling injury data	
Injury probability distributions	
Generalized linear models	
Generalized additive models	
Cox proportional hazards model	
Estimating injury risks and risk ratios using regression	
Accounting for repeated measures	
Summarizing physical activity data	
Flexible weighted cumulative exposure methods	
Chapter 4: Manuscript 1	
Preface	
Principles of musculoskeletal sport injuries for epidemiologists: a review.	
Chapter 5: Manuscript 2	
Preface	
Time-varying associations between physical activity and injury risk among	g children 74
Supplementary Material	
Chapter 6: Manuscript 3	
Preface	
Immortal time bias: a common problem in observational studies of training	g load and injury . 99
Chapter 7: Manuscript 4	
Preface	
The target trial framework for determining the effect of changes in training	
using observational data: a methodological commentary	

Chapter 8: Manuscript 5	
Preface	
The effect of changes in planned participation on injury risk in adolescent ice he	ockey players:
a target trial emulation	
Supplementary Material	
Chapter 9: Discussion	
Preface	
Summary of Findings	
Significance of Findings	
Limitations	
Measurement of physical activity	
Selection bias	
Grouping acute and overuse injuries	
Initial versus subsequent injuries	
Unmeasured confounding and exchangeability assumption	
Positivity assumption	
Accounting for repeated measures	
Future Directions	
Conclusion	
References	

Abstract

Physical activity is a key component of maintaining a healthy lifestyle, and is particularly important for the healthy development of children and adolescents. However, physical activity also has an inherent risk of injury. There exists limited research and few guidelines on the impact of increasing physical activity levels on injury risk among children and adolescents. While the evidence base is larger for adults, existing studies are limited by sample size and methodological issues. Further, few studies have explicitly aimed to estimate the causal effect of changes in activity on injury risk. The objectives of this manuscript-based thesis are (1) to assess the relationship between changes in physical activity levels and injury risk among children and adolescents, and (2) to inform the methodology for future research in this area. Manuscript 1 is a narrative review of the pathophysiological and epidemiologic principles underlying musculoskeletal sport injuries. It is targeted towards epidemiologists without formal training in this substantive area. Manuscript 2 explores how past physical activity levels are associated with incident injury risk in schoolchildren using flexible weighted cumulative exposure methods. It shows that activity performed ten or more weeks ago has a protective association with current injury risk, while activity performed in two to nine weeks ago has a detrimental association. Manuscript 3 is a methodological commentary on immortal time bias in observational studies of changes in activity and injury risk. It shows how conventional approaches for measuring activity result in bias, and provides recommendations for avoiding these biases. Manuscript 4 is a methodological commentary that discusses how observational data can be used to estimate the causal effect of changes in activity on injury risk through the application of the target trial framework. Manuscript 5 applies the target trial framework to adolescent ice hockey players and assesses the intention-to-treat effect of increasing participation duration by varying amounts on injury risk. It employs flexible non-linear modelling approaches to show that injury risk increases consistently with increases in planned participation duration among adolescents. Together, these manuscripts advance the methodology used for assessing the relationship between changes in physical activity levels and injury risk, with the ultimate goal of generating valid recommendations to promote physical activity among children and adolescents while minimizing risk of injury.

Resumé

L'activité physique constitue un élément clé du maintien d'un mode de vie sain et est particulièrement importante pour le développement sain des enfants et des adolescents. Cependant, l'activité physique comporte également un risque inhérent de blessure. Les recherches sur l'impact de l'augmentation des niveaux d'activité physique sur le risque de blessure chez les enfants et les adolescents sont restreintes, et peu de lignes directrices existent à ce sujet. Même si la base de données probantes est plus vaste pour les adultes, les études existantes sont limitées par la taille des l'échantillons et des problèmes méthodologiques. De plus, peu d'études ont visé explicitement à estimer l'effet causal des changements d'activité sur le risque de blessure. Les objectifs de cette thèse sont (1) d'évaluer la relation entre les changements dans les niveaux d'activité physique et le risque de blessure chez les enfants et adolescents, et (2) d'éclairer la méthodologie des recherches futures dans ce secteur. Le manuscrit 1 est une revue narrative des principes physiopathologiques et épidémiologiques qui sous-tendent les blessures sportives musculo-squelettiques. Il s'adresse aux épidémiologistes sans formation formelle dans ce domaine particulier. Le manuscrit 2 explore la manière dont l'activité physique antérieure est associée au risque de blessure accidentelle chez les écoliers à l'aide de méthodes flexibles d'exposition cumulative pondérée. Cela montre qu'une activité antérieure effectuée il y a dix semaines ou plus a une association protectrice par rapport au risque de blessure actuel, tandis qu'une activité menée au cours des deux à neuf semaines précédentes a une association préjudiciable. Le manuscrit 3 est un commentaire méthodologique sur le biais de temps immortel dans les études observationnelles sur les changements d'activité et le risque de blessure. Il montre comment les approches conventionnelles de mesure de l'activité entraînent des biais et fournit des recommandations pour éviter ces biais. Le manuscrit 4 est un commentaire méthodologique qui explique comment les données d'observation peuvent être utilisées pour estimer les associations causales entre les changements d'activité et le risque de blessure grâce à l'application du cadre d'essai cible. Le manuscrit 5 applique le cadre d'essai cible à des joueurs de hockey sur glace qui sont adolescents et évalue l'effet de l'intention de traiter en augmentant la durée de participation prévue sur le risque de blessure. Il utilise des approches de modélisation non linéaires flexibles pour montrer que le risque de blessure augmente de manière cohérente par suite d'une augmentation de la durée de participation prévue chez les adolescents.

Ensemble, ces manuscrits font progresser la méthodologie utilisée pour évaluer la relation entre les changements dans les niveaux d'activité physique et le risque de blessure, dans le but ultime de générer des recommandations valides pour promouvoir l'activité physique chez les enfants et les adolescents tout en minimisant le risque de blessure.

Acknowledgements

Firstly, I thank my PhD co-supervisors, Ian Shrier and Jay Kaufman. Together, they have been instrumental in my development as an epidemiologist and independent researcher. Ian has been my supervisor for the past 6 years, spanning my MSc and PhD. He has contributed to my understanding of the physiological and clinical aspects of physical activity, sport, and injury. He was also the first to encourage me to try and qualify for the Boston Marathon, and has provided valuable guidance regarding injury management and prevention in my personal life. Jay was my thesis committee member during my MSc, and I am grateful he took on a supervisory role for my PhD. He has contributed greatly to my understanding of epidemiologic methods and causal inference.

I thank my thesis committee member, Russell Steele, who has acted in this capacity for the past 6 years. Russ has provided immense statistical guidance and has contributed greatly to developing my methodological skills.

I thank the current and former students in my research group. Tyrel Stokes and Jorge Trejo Vargas provided valuable statistical guidance for my research. Sabrina Yusuf and Hussein Samhat played important roles in validating data used in this thesis.

I thank my collaborators and co-authors, including Michal Abrahamowicz and Marie-Ève Beauchamp from the Research Institute of the McGill University Health Centre, Niels Wedderkopp and Eva Jespersen from the University of Southern Denmark, and Carolyn Emery, Paul Eliason, and Jean-Michel Galarneau from the University of Calgary.

I thank the faculty and administrative staff in the Department of Epidemiology, Biostatistics, and Occupational Health at McGill University, and at the Centre for Clinical Epidemiology and Lady Davis Institute for Medical Research. I also thank the professors and teaching assistants for the various courses I took during my PhD, who have contributed greatly to my understanding of different aspects of epidemiology.

Je remercie Lise Hardy pour son aide avec la version française de mon résumé de thèse.

I acknowledge scholarship funding from Fonds de recherche du Québec - Santé and the Canadian Institutes of Health Research.

Finally, I thank my family and friends for their continued support within and outside my studies.

Contribution to Original Knowledge

The contents of this thesis constitute an original contribution to the field of sport injury epidemiology, providing novel insights into (1) the relationship between changes in physical activity and injury among children and adolescents, and (2) the methodology that can be applied for causal inference in this area.

Manuscript 1 is a narrative review on the definitions, mechanisms, and healing of sport injuries from an epidemiological and biological perspective. It was written with the goal of providing a concise introductory overview for epidemiologists interested in advancing this field of research. To our knowledge, there are no other reviews with this purpose in the literature.

Manuscript 2 is the first study to apply flexible weighted cumulative exposure methods in the field of sport injury epidemiology, and is one of only two studies to apply flexible non-linear methods to describe patterns of sport or physical activity. It is also the first study to specifically assess how patterns of previous physical activity are associated with injury risk in children.

Manuscript 3 is the first manuscript to explicitly define an issue where individuals with small observed changes in activity levels appear to have an increased risk of injury as a type of immortal time bias. It is also the first to provide concrete strategies to prevent or mitigate this bias within this field of research.

Manuscript 4 is a methodological commentary that describes the application of the target trial framework to questions of activity and injury. It is the first manuscript to propose the use of the target trial framework in this area of research, and the first to describe the application of any such framework to estimate the causal effect of changes in activity on injury risk.

Manuscript 5 is one of few studies to examine the relationship between increases in activity and injury risk in adolescent ice hockey. It is also the first study to apply the target trial framework to a research question concerning the relationship between changes in activity and injury risk.

Contribution of Authors

Manuscript 1: Chinchin Wang, Steven D. Stovitz, Jay S. Kaufman, Russell Steele, Ian Shrier

C.W. and I.S. conceptualized the review. C.W. drafted the initial manuscript. As clinicians working in the field of sport injury epidemiology, S.D.S. and I.S. provided important clinical knowledge relating to the concepts covered in the manuscript. As epidemiologists without substantive expertise in the field, J.S.K. and R.S. provided important feedback regarding clarity for non-experts. All authors were involved in critically reviewing and revising the manuscript. All authors approved the final manuscript for submission.

Manuscript 2: Chinchin Wang, Michal Abrahamowicz, Marie-Eve Beauchamp, Jay S. Kaufman, Russell Steele, Eva Jespersen, Niels Wedderkopp, Ian Shrier

C.W., M.A., M-E.B., J.S.K., R.S., and I.S. conceptualized the use of flexible weighted cumulative exposure methods to summarize physical activity patterns. M.A. is the original developer of these methods, and M-E.B. has contributed greatly to the development of statistical packages and code to utilize these methods. C.W. conceptualized the study. C.W., M.A., M-E.B., J.S.K., R.S., and I.S. designed the study. C.W. conducted analyses, with guidance from M.A. and M-E.B. C.W. drafted the initial manuscript, and M.A. and M-E.B. contributed to drafting text related to statistical methodology. N.W. is the principal investigator for the parent study that generated the data for this study. E.J. and N.W. were involved in data collection for the parent study. All authors were involved in critically reviewing and revising the manuscript. All authors approved the final manuscript for submission.

Manuscript 3: Chinchin Wang, Jay S. Kaufman, Ian Shrier

C.W. conceptualized this commentary based on discussions with I.S. C.W. drafted the initial manuscript. All authors were involved in critically reviewing and revising the manuscript. All authors approved the final manuscript for submission.

Manuscript 4: Chinchin Wang, Jay S. Kaufman, Russell Steele, Ian Shrier

C.W. conceptualized the application of the target trial framework to studies on activity and injury, based on discussions with R.S. and I.S. C.W. conceptualized and drafted the initial manuscript. All authors were involved in critically reviewing and revising the manuscript. All authors approved the final manuscript for submission.

Manuscript 5: Chinchin Wang, Paul Eliason, Jean-Michael Galarneau, Carolyn A. Emery, Sabrina Yusuf, Russell J. Steele, Jay S. Kaufman, Ian Shrier

C.W. conceptualized this study. C.W. designed the study and conducted analyses. R.J.S. and I.S. contributed to study design, and R.J.S. provided important guidance regarding analyses. C.A.E. is the principal investigator for the parent study that generated the data for this study. P.E., and J-M.G. are part of the research team that collected and validated data for the parent study. C.W., S.Y., and I.S. also contributed to data validation. C.W. drafted the initial manuscript. All authors were involved in critically reviewing and revising the manuscript. All authors approved the final manuscript for submission.

List of Tables

Chapter 4: Manuscript 1
Table 1. Common categorizations of sport injuries in sport injury epidemiology 45
Chapter 5: Manuscript 2
Table 1. Characteristics of study participants from a Danish cohort of schoolchildren,
2008-2014 (N=1,667)
Table 2. Injury hazard ratios for different patterns of activity over the previous 20
weeks
Table S1. Bayesian Information Criterion for weighted cumulative exposure models
with different user-specified time windows and interior knots
Chapter 8: Manuscript 5
Table 1. Target trial specification and emulation to estimate the intention-to-treat effect
of changing participation duration on injury risk among adolescent ice hockey players
without recent injuries
Table 2. Characteristics of participants and player-trials included in a target trial
emulation to estimate the intention-to-treat effect of changing participation duration on
injury risk among adolescent ice hockey players without recent injuries

List of Figures

Chapter 4: Manuscript 1
Figure 1. Relationship between a musculoskeletal tissue's load capacity, load, and injury.
Chapter 5: Manuscript 2
Figure 1. Weight function for the association between past physical activity and the
current hazard of first injury in children92
Figure 2. Weight functions for the separate associations between past physical activity
and the current hazard of first injury in children, stratified by sex
Figure S1. Exponentially weighted moving average (EWMA) function for the association
between previous physical activity and injury97
Chapter 6: Manuscript 3
Figure 1. Immortal time bias in the measurement of load
Figure 2. Cloning and censoring method to prevent immortal time bias in studies of load
and injury
Chapter 7: Manuscript 4
Figure 1. Immortal time bias in the measurement of load
Chapter 8: Manuscript 5
Figure 1. Participant flow chart for the emulation of a target trial assessing the effect of
changing participation duration on injury risk among adolescent ice hockey players
without recent injuries
Figure 2. Intention-to-treat effect of changing participation duration on injury risk among
adolescent ice hockey players without recent injuries
Figure 3. Intention-to-treat effect of changing participation duration on injury risk among
adolescent ice hockey players without recent injuries, by league bodychecking status. 165
Figure 4. Intention-to-treat effect of changing participation duration on concussion risk
among adolescent ice hockey players without recent injuries
Figure S1. Schematic displaying the specification of the intervention in a target trial
estimating the intention-to-treat effect of changing participation duration on injury risk.

Figure S2. Directed acyclic graph for the relationship between planned changes in	
participation and injury among adolescent ice hockey players 1	68
Figure S3. Risk ratios for the intention-to-treat effect of changing participation duration	l
on injury risk among adolescent ice hockey players without recent injuries 1	69

List of Abbreviations and Acronyms

AICAkaike Information CriterionBICBayesian Information CriterionCIConfidence intervaldfDegrees of freedomDLNMDistributed lag non-linear modelDOMSDelayed onset muscle sorenessEWMAExponentially weighted moving averageGAMGeneralized additive modelGEEGeneralized additive mixed modelGEMGeneralized estimating equationsGLMGeneralized linear model
CIConfidence intervaldfDegrees of freedomDLNMDistributed lag non-linear modelDOMSDelayed onset muscle sorenessEWMAExponentially weighted moving averageGAMGeneralized additive modelGEEGeneralized additive mixed modelGEEGeneralized not in the solutionGLMGeneralized linear model
dfDegrees of freedomDLNMDistributed lag non-linear modelDOMSDelayed onset muscle sorenessEWMAExponentially weighted moving averageGAMGeneralized additive modelGAEGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
DLNMDistributed lag non-linear modelDOMSDelayed onset muscle sorenessEWMAExponentially weighted moving averageGAMGeneralized additive modelGAMMGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
DOMSDelayed onset muscle sorenessEWMAExponentially weighted moving averageGAMGeneralized additive modelGAMMGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
EWMAExponentially weighted moving averageGAMGeneralized additive modelGAMMGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
GAMGeneralized additive modelGAMMGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
GAMMGeneralized additive mixed modelGEEGeneralized estimating equationsGLMGeneralized linear model
GEEGeneralized estimating equationsGLMGeneralized linear model
GLM Generalized linear model
GLMM Generalized linear mixed model
HR Hazard ratio
IOC International Olympic Committee
IRF Injury report form
ITT Intention to treat
LRT Likelihood ratio test
PH Proportional hazards
PP Per protocol
RCT Randomized controlled trial
RPE Rating of perceived exertion
RR Risk ratio
sRPE Session RPE
WCE Weighted cumulative exposure

1 Chapter 1: Introduction

Physical activity plays a crucial role in maintaining healthy lifestyles.¹ Among children and
adolescents, physical activity promotes healthy bone growth and development, improves
cardiovascular health, and prevents obesity and its related morbidities.² Organized sport
participation is a common avenue for children and adolescents to obtain regular physical activity,
and provides additional psychosocial benefits including improved socialization and mental
health.³

8 Despite its benefits, physical activity and sport participation have an inherent risk of injury.⁴

9 Consequences of injury are wide-ranging, and may include time lost from participation in sport

10 or physical activity due to recovery needs or reduced enthusiasm, impaired development, and the

11 need for surgical intervention.^{5,6} Sport-related injuries are also an economic burden,^{7,8} with

12 physical activity and sport participation accounting for over 50,000 emergency room visits in

13 Canada between 2007 and 2010.⁹ As such, reducing injuries while maintaining or increasing

14 physical activity is an important goal for children and adolescents, parents, coaches, and

15 clinicians.

16 Current guidelines recommend that children and adolescents obtain a minimum threshold of physical activity for health, with further activity resulting in greater health benefits.¹⁰ There are 17 18 limited research and no guidelines on the risks of injury associated with increasing physical 19 activity levels among children and adolescents. Among adults, the International Olympic 20 Committee (IOC) currently recommends that athletes increase their activity by less than 30% in a given week to minimize injury risk.¹¹ However, these recommendations are based off studies 21 with small sample sizes and methodological issues, and are unlikely to be valid.^{12,13} There 22 23 remains an evidence gap in the relationship between changes in physical activity levels and 24 injury risk, particularly among children.

25 The objectives of this thesis are (1) to assess the relationship between changes in physical

26 activity levels and injury risk among children and adolescents, and (2) to inform the

27 methodology for future research in this area. Our overarching goal is to inform recommendations

17

- 28 for promoting increased physical activity while minimizing injury risk among children and
- adolescents.

30 Organization of Thesis

31 Chapter 2 reviews the relationship between physical activity and injury, including current 32 evidence in the literature and their limitations. Chapter 3 reviews statistical methods relevant to 33 this thesis. Chapter 4 contains Manuscript 1, a narrative review on the biological and 34 epidemiologic principles underlying musculoskeletal sport- and physical activity-related injuries. 35 Chapter 5 contains Manuscript 2, a research study that employs flexible non-linear methods to 36 determine the association between previous physical activity patterns and current injury risk in schoolchildren. Chapter 6 contains Manuscript 3, a methodological commentary on immortal 37 38 time bias in observational studies of changes in activity and injury risk. Chapter 7 contains 39 Manuscript 4, a methodological commentary on the potential application of the target trial 40 framework to estimate causal associations between change in activity and injury risk using 41 observational data. Chapter 8 contains Manuscript 5, a research study that applies the target trial 42 framework to adolescent ice hockey players to determine the effect of increasing planned 43 participation on injury risk. Chapter 9 discusses the overall findings of the thesis, limitations, and future directions. 44

45

46 Chapter 2: Literature Review

47 **Preface**

48 In this chapter, I provide a summary overview of the relationship between physical activity and

49 injury. A more in-depth review of musculoskeletal sport injuries, their mechanisms, and various

50 definitions and categorizations used in epidemiologic research is provided in Chapter 4

51 (Manuscript 1).

52 Mechanism of physical activity-related and sport injuries

53 Physical activity and sport exposes individuals to various forces. The sum of these forces acting 54 on a tissue is referred to as load (or tissue load).¹⁴ Although tissues within the human body have 55 a certain tolerance for load, loads beyond a tissue's load capacity will result in tissue damage.^{14,15} 56 Significant tissue damage that can be visualized without a microscope or that results in physical 57 symptoms are generally considered injuries.^{14,15}

The musculoskeletal system is primarily responsible for human movement and the ability to be physically active.¹⁶ Musculoskeletal tissues, which include bones, muscles, ligaments, and tendons, typically adapt and strengthen in response to loads that are close to or minimally above their load capacities, given sufficient recovery time.^{17,18} However, without sufficient recovery, microdamage will accumulate and load capacity will decrease, resulting in injury even at normal loads (known as "overuse" injuries).^{14,15,19} Further, sudden large loads will cause immediate tissue damage and injury (known as "acute" injuries).^{14,15}

Injuries can also occur to other organ systems during physical activity. Most notable among children and adolescents are neurological injuries, including concussions.²⁰ The biological mechanism for neurological injuries are similar to musculoskeletal injuries, where excessive loads will lead to tissue damage.²¹ Neurological injuries generally occur due to trauma or stretching,²⁰ and as such are considered acute injuries.²²

70 Fatigue and injury

71 Fatigue is a complex phenomenon that within this review, is defined as the decreased capacity of an individual to perform maximal physical activity due to prolonged physical exertion.²³ Fatigue 72 73 can occur with unaccustomed or strenuous physical activity, and has both physical and mental aspects.^{23,24} On a physiological level, fatigue is often associated with changes in cellular states 74 75 that decrease the capacity of tissues and organs to sustain the original level of physical activity using typical resources.²³ Muscle fatigue in particular refers to a reduction in the maximal force a 76 muscle can exert.^{23,25} Mental fatigue is a psychological state caused by prolonged cognitive 77 activity.^{23,24} Mental fatigue reduces an individual's cognitive and physical capacities, and is often 78 associated with feelings of tiredness.^{23,24} Muscle and mental fatigue are reversible with sufficient 79 80 rest and recovery,²³ and are not considered injuries.

Both muscle and mental fatigue are causal risk factors for injury through various mechanisms.
Muscle fatigue is caused by the accumulation of tissue microdamage, resulting in a decreased
load capacity and increased susceptibility to overuse injuries.^{14,17} Both muscle and mental fatigue
can impair balance, proprioception, spatial awareness, and reaction time,^{23,26–30} increasing
susceptibility to acute injuries (musculoskeletal or neurological) through falls or collisions.

86 Measurement of load and physical activity

Although it is the load on tissues that affects physical adaptations and injury, in practice, tissue loads are rarely measured.³¹ Direct measurements of tissue load require invasive and expensive methods,³¹ and are thus infeasible for epidemiologic studies. Tissue load can be approximated using external force, which refers to the forces applied to the surface of the human body.^{14,31} However, the measurement of external force requires specialized sensors that are generally infeasible on a population level.^{14,31} Instead, load is typically quantified using indirect measures.

93 A note on terminology

I defined "load" earlier as the sum of the forces that an individual is exposed to during physical
activity and sport. However, sport medicine researchers often use "load" as a short-form for
"workload" or "training load".³² Training load has been described as the input variable that is

manipulated to elicit a desired training response.^{33–35} Workload is another common term that is 97 often used interchangeably with training load.³² In practice, "load", "workload", and "training 98 load" are generic terms used to cover a broad range of exposure variables related to sport and 99 physical activity that may act as indirect measures of tissue load.^{32,35} While I generally use the 100 terms "physical activity" or "activity" to represent my exposures of interest, depending on the 101 102 context and target audience for a particular manuscript, I may use "load" in the same sense. For 103 the remainder of this thesis, the term "load" is used to refer to a general exposure variable related 104 to sport or physical activity rather than tissue load itself, unless otherwise stated.

105 Internal and external load

106 Load measures can be described as "external" or "internal.³³ External loads generally encompass

107 measures that individuals are physically exposed to, while internal loads encompass

108 psychophysiological measures relating to how individuals respond to activity.^{33,35} Individuals

109 exposed to the same external load may have different internal loads.³³

110 Measures of load in epidemiology

111 Measures of load usually relate to one or several components of physical activity: frequency,

112 intensity (rate of energy expenditure), time (duration), and type.³⁶ These components can be

113 combined into a single measure of energy expenditure.³⁶ The gold standard for measuring energy

114 expenditure is the doubly labeled water method;³⁷ however, it has a high cost and cannot separate

115 out the individual components of physical activity.^{36,38} As such, it is rarely applied in

116 epidemiologic research.³⁸

117 In epidemiologic studies, load is often defined simply by the frequency or duration of activity.

118 These are measures of external load, and can be general (e.g. number of activity sessions per

119 week, total duration of activity per week), or specific to the sporting context (e.g. distance

120 covered, weight lifted).³³ Frequency, duration, and type of physical activity are relatively

121 straightforward to assess through self-report or direct observation, although self-reported

122 measures can be limited by measurement error and recall or response biases.^{39,40}

123 The intensity of activity also impacts tissue load.^{14,40} Intensity can be assessed using devices or

124 through self-report. The rate of oxygen consumption is an internal physiological measure of the

125 intensity of activity that is sometimes used to assess endurance activities.³¹ While oxygen

126 consumption can be measured directly, it is normally approximated by heart rate.^{41,42} Heart rate

127 monitors are relatively low cost, portable, and non-invasive devices that can often measure

128 duration of activity as well as heart rate.^{41,42}

129 Accelerometers are another low cost and portable device used to assess physical activity.

130 Accelerometers measure the human body's acceleration in multiple planes, which can be used to

131 predict intensity and energy expenditure.⁴¹ They are also able to measure frequency and duration,

132 and can differentiate between some types of activity (e.g. running vs. cycling).^{41,43} However,

133 accelerometers are not necessarily reliable,⁴⁴ and are not able to accurately detect all types of

134 movements.⁴⁵ Accelerometers have been employed in several large-scale population studies,

135 including national surveys in Canada and the US.⁴³

136 Despite the benefits of device-based measures of physical activity, their associated costs can be prohibitive, particularly for large-scale longitudinal studies.⁴⁶ Further, device reliability can be 137 138 affected by human error (e.g. forgetting to wear the device or accidentally altering the device position).^{44,46} Intensity can be measured subjectively using ratings of perceived exertion (RPE). 139 RPE is self-reported, typically using a numerical scale, and is a measure of internal load.^{47,48} 140 141 RPE has been found to be correlated with physiological measures like heart rate and oxygen consumption.⁴⁸ Session RPE (sRPE) is a modified measure calculated by multiplying RPE by 142 duration of activity.⁴⁹ Perceived exertion can vary greatly between individuals,^{48,50} and is 143 affected by age, sex, and expertise as well as psychological factors like anxiety and stress.⁵⁰ 144 Further, activity duration can influence RPE,⁵¹ as individuals typically perceive greater exertion 145 during prolonged activities due to fatigue.⁴⁸ This complicates the interpretation of sRPE, as 146

147 duration is accounted for twice in its calculation.⁵¹

148 **Quantifying the relationship between physical activity and injury**

149 Absolute activity and injury risk

150 It is generally accepted that large absolute amounts of physical activity result in increased injury

151 risk, regardless of the measure.^{52,53} As injuries occur when tissue load exceeds the load

- 152 capacity,^{14,15} it follows that larger loads result in higher injury risks.
- 153 Increased frequency and/or duration of activity can result in increased risk of injury not only
- through increased loading on tissues, but through increased exposure time at risk and fatigue.^{4,54}
- 155 With increased exposure time, there is greater opportunity for inciting events that result in tissue
- 156 loads exceeding load capacity, such as a fall or rapid movement.⁵⁵ Further, increased activity
- 157 frequency and duration increases the risk of fatigue and resultant injuries.²³
- 158 To minimize injury risk, individuals must minimize fatigue and exposure to tissue loads near or
- above their load capacity. Because fatigue and load capacity are functions of tissue strength and
- 160 preparedness from previous activity,^{14,23} studying current activity relative to previous activity can
- 161 provide more insight into the relationship between physical activity and injury.

162 The fitness-fatigue model

In the 1970's, Banister et al. proposed a "fitness-fatigue" model to relate training patterns to athletic performance.⁵⁶ In this model, performance is determined by a combination of fitness, or beneficial physiological effects of long-term training, and fatigue, or negative physiological effects of shortterm training.^{56–58} As tissue adaptations take time, activity performed in the past is regarded to improve fitness whereas activity performed in the present or very recently is regarded to cause fatigue.⁵⁶ Performance is optimal when fitness is high and fatigue is low.^{56–58}

169 The acute:chronic workload ratio

- 170 In 2014, Gabbett et al. proposed a model called the "acute:chronic workload ratio (ACWR)" to
- relate training patterns to injury risk.^{59,60} Similar to the "fitness-fatigue" model, past activity
- 172 ("chronic load") represents a proxy for fitness, or the activity to which an individual is
- accustomed to, whereas current or recent activity ("acute load") represents a proxy for

fatigue.^{59,60} The ACWR is calculated by dividing the acute load by the chronic load.^{59,60} Injury risk was proposed to increase as the acute load outweighs the chronic load (ACWR > 1).⁵⁹

Gabbett et al. defined the acute load as the load in the present week (1-week period), and the chronic load as the unweighted weekly average of load in the present and previous 3 weeks (4week period).⁵⁹ The ACWR was originally applied to cricket fast bowlers, with load quantified as (1) the number of balls bowled per week, and (2) sRPE.⁵⁹ The authors found that bowlers with ACWRs greater than 2 had a 3.3-times greater injury risk in the subsequent week than bowlers with ACWRs between 0.5 and 0.99, but no association between the ACWR and injury in the current week.⁵⁹

183 Subsequently, Gabbett et al. also applied the ACWR in studies of rugby and Australian football, 184 quantifying load as distance covered while running. They developed a general model for the 185 relationship between the ACWR and injury risk using data from these three studies. This model 186 identified ACWRs between 0.8 and 1.3 as being associated with the lowest risk of injury in the 187 subsequent week, and ACWRs below 0.8 or above 1.3 as being associated with increased risk of injury.⁶⁰ It had several methodological limitations, including arbitrary discretization of the 188 ACWR prior to modelling, sparse data, and not accounting for repeated measures.¹² Further, it 189 190 combined different measures of load, with results unlikely to be applicable to a wide range of 191 sporting contexts. Even so, this model was used to generate training recommendations for 192 athletes across all team sports in the 2016 "International Olympic Committee (IOC) consensus statement on load in sport and risk of injury".¹¹ 193

194 Variations of the ACWR

195 **Coupled versus uncoupled measure**

In the original formulation of the ACWR, the acute load was defined as the load over the current week, whereas the chronic load was the average weekly load over the current week and previous weeks.⁵⁹ This creates a "coupled" measure, where the acute load is included in the numerator and denominator, and is a proportion rather than a true measure of change.¹² The coupled measure with an acute window of 1 week and chronic window of 4 weeks is capped at ACWR=4 as the acute load increases towards infinity and the chronic load decreases towards zero,^{12,61} and 202 is thus limited in its ability to differentiate injury risks at high acute loads.^{12,62} An alternative

- ²⁰³ "uncoupled" measure¹² defines the chronic load as the average weekly load over the 3 weeks
- 204 prior to the acute load week, so that the acute load is not included in the chronic load calculation.
- 205 The use of uncoupled acute and chronic loads allows for simpler calculations and interpretations
- 206 of changes in load.¹²

207 Acute and chronic time windows

- 208 Although the original time windows of 1 week (acute) and 4 weeks (chronic) were chosen
- arbitrarily,⁵⁹ they have since been used in the majority of studies employing the ACWR.^{63–65}
- 210 Some studies have explored chronic time windows between 2 and 8 weeks, and found
- 211 differences in model fit.^{62,66–71} Other studies have explored acute time windows of 2 weeks,⁶⁹ or
- have calculated ACWRs using daily data (e.g., 3 day acute load and 21 day chronic load).^{71,72}
- 213 Although multiple authors have recommended that the choice of time windows be decided based
- on the sporting context and schedule,^{63,64,71} there remains no clear guidance on identifying the
- 215 relevant time windows.

216 Unweighted versus weighted averages

217 The chronic load is defined as an unweighted weekly average in the original formulation of the ACWR.^{11,60} The use of an unweighted average obscures daily variations in activity, and assumes 218 that activity performed in each week is equally associated with injury risk.⁷³ This is unlikely to 219 be the case, as the physiological effects of activity on tissues and performance decay over 220 time.^{74,75} In 2017, Williams et al. proposed the use of an exponentially weighted moving average 221 (EWMA) to calculate acute and chronic loads.⁷⁶ The EWMA assigns exponentially decreasing 222 weights for activity performed in each previous day, such that activity performed furthest in the 223 224 past should have the lowest weight.⁷⁶ In actuality, while weights decrease exponentially, activity 225 performed on the day furthest in the past takes on a much higher weight than more recent activity 226 due to the EWMA's mathematical formulation.¹²

- 227 Although we previously explored a modified EWMA that assigned decreasing weights to data
- summarized weekly,⁶² the EWMA is normally applied only when daily data are available.
- 229 Several studies have found that ACWRs measured with the EWMA had greater associations with

injury than ACWRs measured with a daily rolling average,^{64,70,77} while others found no
 difference between the two methods.⁷⁸

A proposed alternative weighting approach is to use non-linear functions based on daily or weekly data to represent weighted cumulative acute and chronic loads.⁷⁹ In simulations, a distributed lag non-linear model was found to better fit the relationship between relative changes in load and injury risk than the coupled unweighted ACWR (7 day acute window and 28 day chronic window).⁸⁰ In Manuscript 2, we explore applying flexible weighted cumulative exposure methods to summarize activity.

238 Current or lagged injury risk

239 Studies have assessed associations between the ACWR and injury in the acute time window (e.g. 240 current day or week), as well as those occurring in a subsequent time period (most often the subsequent week).⁶³ The assessment of injury during the acute time window ignores temporality 241 between activity and injury.⁸¹ Individuals who get injured early in the acute time window are 242 243 likely to have systematically lower loads than those who remain uninjured. This systematic 244 difference can lead to an observed increase in current injury risk at low ACWRs as reported by the IOC consensus statement,¹¹ even if low ACWRs are causally protective against injury.^{12,52} 245 246 This bias is covered in more detail in Chapter 7 (Manuscript 3).

247 Employing a lag period for injury avoids this bias. Some authors have suggested that the use of a 248 lag period is also important because spikes in activity can predispose individuals to higher injury 249 risks for up to 4 weeks.^{63,69} However, the use of a lag period ignores the principle that a current 250 tissue load beyond the load capacity is the impetus for injury. It also ignores variations in current 251 activity that may influence injury risk.¹² This is of particular concern when acute and chronic loads are measured using weekly blocks.¹² For instance, an athlete with a low ACWR in the 252 253 current week might become injured upon increasing their load in the subsequent week (i.e. have 254 a high ACWR in the subsequent week). This would result in an apparent increased injury risk in 255 the subsequent week at low ACWRs in the current week, even if low ACWRs are causally 256 protective against injury.

257 Current state of evidence

Since its introduction, the ACWR has been employed in numerous studies assessing injury risk in both team^{63,64} and individual sports.^{82–85} In fact, the number of PubMed-listed publications employing the ACWR has been increasing exponentially year-to-year as of 2020.⁸⁶ Study populations have ranged in performance levels (i.e. recreational, amateur, elite, professional) and age groups.^{63,64,87} Further, the ACWR model presented in the IOC consensus statement has been employed to monitor training and generate recommendations across a variety of sporting contexts.^{88–90}

265 Multiple authors have raised criticisms of the ACWR. These relate mainly to the methodological 266 approaches applied to ACWR studies, including those that led to the IOC consensus model (e.g. 267 discretization, small sample sizes, not accounting for repeated measures, no control for confounding, p-hacking).^{12,91,92} Despite being used to make training recommendations, requiring 268 269 causal interpretations of findings, the majority of studies have been associational, without consideration of causal assumptions.^{12,92,93} To our knowledge, there are no studies that have 270 271 employed causal frameworks to study the effect of the ACWR (or load more generally) on injury risk.⁹³ The varying results obtained under different ACWR variations (i.e. load definition, time 272 273 window, coupled vs. uncoupled measure, unweighted vs. weighted average, injury lag period) has also been cited as a reason against its use.^{13,94} Some authors have also suggested that the use 274 275 of a ratio to express changes in activity in itself is inappropriate because the ACWR does not scale consistently across its range of values (i.e. inaccurate normalization).^{95–97} Despite 276 277 recommendations to develop new conceptual models for the relationship between physical activity and injury,⁹⁴ the ACWR remains one of the only such models and is by far the most 278 279 commonly used by both researchers and the general public.⁷⁹

280 Chapter 3: Overview of Methods

281 **Preface**

In this chapter, I provide a brief overview of the statistical methods used in this thesis, as well asother relevant methods.

284 Measures of injury incidence

Injury incidence is normally expressed as a risk or a rate. Injury risk is defined as the number of

injured individuals divided by the number of individuals at risk over a specified period of time.⁹⁸

287 Injury rate is defined as the number of injuries divided by the total person-time at risk.⁹⁸ Rates

288 might be expressed as the number of injuries per game, activity session, or minutes active.⁹⁹

289 One could also express injury incidence as a time-to-event or hazard (instantaneous event

290 rate).¹⁰⁰ This requires analyzing data using survival analyses, and requires extensions to account

for recurrent or subsequent injuries.^{100,101} While these measures can provide important insights

for reducing injury incidence, they are often less interpretable¹⁰⁰ and prone to selection bias.¹⁰²

293 Finally, injury incidence can be expressed as an odds. Injury odds is defined as the probability of

injury occurring (injury risk) divided by the probability of injury not occurring (1 - injury

risk).¹⁰³ Injury odds are a less intuitive measure than injury risks or rates for the general public,

but are the basis for common statistical procedures (i.e. logistic regression).¹⁰³ As such, they are

297 often employed in the literature.

298 Injury rates may be the preferred measure for comparing injury incidence between different mechanisms (e.g. sports), as they account for differences in exposure time.^{98,104} However, injury 299 risks are generally more intuitive for individual decision-making.⁹⁸ For example, suppose that 300 301 soccer has an injury rate of 0.10 injuries per hour of participation, while hockey has an injury 302 rate of 0.12 injuries per hour of participation. Suppose that a typical practice session in children's 303 soccer lasts 60 minutes while a practice session in children's hockey lasts 45 minutes. While we 304 might conclude that hockey is a more dangerous sport given its higher injury rate, parents might 305 conclude that soccer is a more dangerous sport given that the risk of their child getting injured

within a practice session is greater in soccer (10% injury risk per session in soccer compared to9% in hockey).

The overarching goal for athletes, coaches, clinicians, and parents is to increase activity while minimizing risk of injury. Increases in activity generally involve an increase in the exposure time at risk. Unlike injury risks, comparisons of injury rates do not directly contrast injury likelihood under different activity patterns. The exception is when only the intensity of activity is increased. If we consider the effects of increases in intensity that occur without changing activity frequency or duration, the injury risk will be equivalent to the injury rate over the same period of time.

314 Most load metrics involve a time component (e.g., duration of activity, number of activity 315 sessions, sRPE). This must be taken into consideration when comparing injury risks under 316 different loads. For instance, an individual might be interested in the risk of increasing their 317 activity from 1 hour to 2 hours. Absent a causal effect of changing activity on injury, the injury 318 risk for 2 hours of activity would be expected to be double the risk for 1 hour of activity simply 319 because the exposure time is doubled. However, the injury rate under these two situations would 320 be expected to be equal. If the injury risk for 2 hours of activity was 3-fold that of the risk for 1 321 hour of activity, the risk of injury associated with increasing activity duration would be greater 322 than expected. Therefore, we would conclude a causal effect of increasing activity duration on injury if the necessary assumptions for causal inference are met.⁹⁹ 323

324 Injury effect measures

Ratios are often used to compare injury incidences between groups.⁹⁸ Although risk ratios, rate ratios, odds ratios, and hazard ratios all represent different measures of association, they will approximate each other if the risk of the outcome is rare or when measured over a short period of time.^{105–107} Injury incidences can also be compared between groups using differences,⁹⁸ which take into account the absolute magnitude of injury incidence.¹⁰⁸

Risk ratios are collapsible, meaning that in the absence of confounding, the weighted average of stratified risk ratios will equal the marginal risk ratio. Similarly, adjustment for a variable that is not a confounder will not change the value of the risk ratio.^{109,110} Odds ratios, rate ratios, and hazard ratios are non-collapsible, and as such are more difficult to interpret causally than risk
 ratios.^{102,109–111}

335 Modelling injury data

336 Injury probability distributions

Normally, the injury outcome is assumed to follow a binomial distribution for inferences on
injury risks or odds, and a Poisson or negative binomial distribution for inferences on injury rates
or counts.^{103,112} These assumptions should be checked before conducting analyses, as they may
not always hold. Injury hazards are most commonly modelled using the Cox proportional
hazards (PH) model, which does not make assumptions about the distribution of survival times
(time-to-injury).¹¹³

343 Generalized linear models

344 Generalized linear models (GLMs) are a class of models where the outcome variable is assumed 345 to follow an exponential family distribution which is related by a linear or non-linear link function to a linear combination of the explanatory variables.¹¹⁴ GLMs include logistic models, 346 which use a logit link function, Poisson models, which use a log link function, and log-binomial 347 models, which also use a log link function.¹¹⁴ GLMs impose a single functional form for the 348 349 relationship between the explanatory variables and link-transformed outcome variable across 350 their entire range.¹¹⁵ GLMs encompass the most commonly used models in epidemiology, and are simple to interpret and computationally efficient.¹¹⁶ However, they may result in erroneous 351 inferences if the assumptions regarding functional form are not met.¹¹⁶ 352

353 Generalized additive models

354 Generalized additive models (GAMs) are an extension of GLMs which employ locally smooth

355 functions rather than imposing a single functional form for the relationship between the

- 356 explanatory and outcome variables.¹¹⁵ These smooth functions can be estimated non-
- 357 parametrically, and do not have to be specified.¹¹⁴ GAMs can be used with any of the link
- 358 functions employed under GLMs.¹¹⁵ As such, they are a flexible, non-linear alternative to

- 359 GLMs.¹¹⁵ GAMs are not necessarily uniformly better than GLMs. They are more
- 360 computationally intensive than GLMs, more difficult to interpret, and may result in

361 overfitting.¹¹⁷

We employ GAMs to model the relationship between changes in hockey participation and injuryrisk in Manuscript 5.

364 **Cox proportional hazards model**

The Cox PH model is the most commonly used model for survival analyses.¹¹³ It is a semi-365 366 parametric model, in that it does not make distributional assumptions about survival times.¹¹⁵ This makes it popular over parametric models such as the exponential or Weibull model.¹¹⁸ 367 368 However, the Cox PH model does specify the form in which the explanatory variables affect the 369 hazard rate (a multiplicative or "proportional" relationship so that the relative hazards are 370 constant over time, known as the proportional hazards assumption).^{113,115} Violations of this assumption can be handled by including time-varying covariates within the Cox PH model,¹¹⁹ or 371 stratification by variables that do not satisfy the assumption.¹²⁰ While explanatory variables are 372 generally assumed to be linearly related to the log hazard in a Cox PH model,¹¹³ the form of the 373 374 explanatory variables can be expressed flexibly with non-linear functions.¹¹⁵

375 Cox PH models are limited in that while the magnitude of hazard ratios tends to vary over time, simply fitting a Cox PH model will provide an average hazard ratio over time (ignoring period-376 specific changes in hazard ratios).¹⁰² Further, they have a built-in selection bias where the 377 calculation of hazard ratios at a particular time point is limited to those who have survived up to 378 379 that time.¹⁰² This is of particular concern for research questions involving physical activity and 380 injury risk, because those who remain uninjured will accumulate more observed physical activity 381 than those who become injured earlier on. An alternative survival method that may be more 382 amenable for causal inference is the Aalen additive hazards model; however, it is much less utilized in the literature.^{121,122} 383

We employ the Cox PH model to assess the association between previous activity and time-to-first injury in Manuscript 2.

386 Estimating injury risks and risk ratios using regression

Binary data are often analyzed using logistic regression.¹²³ Logistic regression involves
modelling the log odds of an outcome, and is used to estimate odds ratios.¹²³ However, injury
odds and odds ratios are rarely the incidence measure of interest.¹⁰³ Further, injury is not
necessarily a rare outcome, and as such, injury odds and odds ratios may not be good
approximators of injury risks and risk ratios.¹⁰³

- 392 Log-binomial regression yields estimates of risk ratios.¹²⁴ However, there are frequent issues
- 393 with fitting log-binomial models due to failed convergence.¹²⁴ A modified Poisson regression
- 394 can also be used to estimate risk ratios; however, expected probabilities may not be constrained
- 395 between 0 and 1.^{124,125}

396 Logistic regression can be used to derive risks and risk ratios through marginal

- 397 standardization.^{125,126} Briefly, one could fit a logistic model to the data, and calculate predicted
- 398 odds of the injury outcome for the study population under different treatment conditions.
- 399 Covariate values are set to their observed values, so that only the treatment variable changes. The
- 400 odds can be converted to risks, and risk ratios calculated by dividing the risks under the different
- 401 treatment conditions.¹²⁶ Bootstrapping can be used to calculate standard errors and confidence
- 402 intervals for the risks and risk ratios.^{125,126} Marginal standardization is implemented to calculate
- 403 risks and risk ratios in Chapter 8 (Manuscript 5).

404 Accounting for repeated measures

Injury is a recurrent event, in that it can occur multiple times in the same individual.¹⁰¹ Most 405 406 longitudinal studies relating to sport injuries collect repeated measurements of activity and injury within the same participants over time.^{127,128} Standard modelling approaches assume that each 407 observation (i.e. occurrence of injury) is independent.¹²⁷ However, observations within an 408 individual are often correlated more strongly than observations between individuals. Not 409 accounting for this correlation will result in overly precise and potentially biased estimates.^{101,128} 410 411 Further, individuals may have different risk factors that make them more or less susceptible to injury, and therefore different baseline injury risks.¹⁰¹ The influence of different covariates on 412 injury risk may also differ between individuals.¹⁰¹ Standard modelling approaches assume the 413

same baseline injury risk given a set of covariates for all individuals, and the same influence of
covariates on injury risk.¹⁰¹ Below, I briefly discuss several analytical techniques that can be
used to account for repeated measures.

417 Mixed effect models

418 Repeated measures can be accounted for by including a random effect (intercept and/or slope)

- 419 within a GLM or GAM. Random effects are able to vary across clusters (in this case individuals),
- 420 as opposed to fixed effects which are set to a particular value for all individuals.¹²⁹ Models that
- 421 allow for random effects are generally referred to as mixed effect models, and include
- 422 generalized linear mixed models (GLMMs) and generalized additive mixed models
- 423 (GAMMs).^{129,130} These models account for correlation within clusters in their standard errors.¹²⁹
- 424 When the probability of injury is modelled using a standard fixed effect model (GLM or GAM),
- 425 individuals are assumed to have the same baseline injury risk, reflected in a fixed intercept.¹⁰¹
- 426 The influence of covariates on injury risk are also assumed to be the same across all individuals
- 427 and injuries, reflected in fixed slopes.¹⁰¹ Random intercepts on the individual-level allow the
- 428 baseline injury risk to vary between individuals, whereas random slopes allow the influence of
- 429 covariates to vary.¹⁰¹ Mixed effect models make distributional assumptions about their random
- 430 effects, and may generate biased estimates if the model is misspecified.¹³¹ Several packages exist
- 431 to implement GLMMs and GAMMs in standard statistical software.^{132,133} Issues with
- 432 convergence tend to occur as more random effects are included in a model.¹³⁴

433 Generalized estimating equations

434 Generalized estimating equations (GEE) account for repeated measures by estimating the correlation within individuals through an iterative process.¹³⁵ Unlike mixed effect models, GEE 435 436 employs robust estimation of standard errors, and as such, does not require a correctly specified error distribution.¹³¹ Whereas mixed effect models provide estimates conditional on the 437 individual, GEE is used to fit marginal or population average models.¹³¹ While a mixed effect 438 439 model might be used to estimate the average change in injury risk associated with a change in 440 activity level (high vs. low) within an individual, a population average model would estimate the change in the average injury risk for high versus low activity levels for the population.¹³¹ 441

442 Conditional estimates may of greater interest to individuals or clinicians looking to decrease their

- 443 own or their patient's risk of injury, whereas marginal estimates may be of more interest to
- 444 policymakers looking to decrease injuries within a population. Parameter estimates for mixed
- 445 effect models versus GEE typically differ when using logistic regression, with population
- 446 average estimates closer to the null value of $1.^{131}$ While GEE can be easily implemented with
- 447 GLMs using standard statistical software,¹³⁶ extensions to GAMs are limited.

448 Cluster bootstrapping

- 449 Cluster bootstrapping is an alternative approach to account for repeated measures, particularly
- 450 when the number of clusters is low.¹³⁷ Cluster bootstrapping is based off of standard
- 451 bootstrapping. Whereas standard bootstrapping involves drawing a random sample of
- 452 observations (i.e. rows of data) equal to the size of the dataset repeatedly with replacement,
- 453 cluster bootstrapping involves drawing a random sample of clusters (e.g. individuals) equal to
- the number of clusters with replacement.¹³⁷ All observations within the sampled clusters are
- 455 included when computing the statistics of interest, and confidence intervals are derived using
- 456 standard procedures (e.g. by taking the 2.5% and 95% percentiles of the distribution of the
- 457 sample statistic).^{137,138}A major limitation of cluster bootstrapping compared to mixed effect
- 458 models and GEE is its computational intensity.¹³⁷
- We employ cluster bootstrapping to account for repeated measures on the individual level inManuscripts 2 and 5.

461 Summarizing physical activity data

As discussed in Chapter 2, researchers often employ unweighted or weighted averages to
represent cumulative physical activity or load data, particularly those that occurred in the past
(i.e. chronic load). We explore using flexible weighted cumulative exposure (WCE) methods to
represent cumulative physical activity in Manuscript 2.

466 Flexible weighted cumulative exposure methods

467 WCE methods were originally proposed as a way to summarize past exposures (specifically doses) in pharmacoepidemiology.¹³⁹ Cumulative doses at any given time were calculated as a 468 weighted mean of past doses, with higher weights assigned to more recent doses, using a pre-469 specified weight function.¹⁴⁰ The cumulative dose could then be included as a time-varying 470 exposure in a Cox PH model.¹⁴⁰ Flexible WCE methods are an extension to this approach, where 471 472 the weight functions are estimated flexibly using cubic regression B-splines rather than imposing a specific functional form.¹⁴¹ The weight function is then included as the exposure of interest in a 473 Cox PH model, which allows for the adjustment of additional covariates.^{139,141} 474

475 Flexible WCE methods offer a data-driven approach to (1) weighting past exposures, and (2)

476 determining the relevant time window over which a cumulative exposure is associated with an

477 outcome.^{139,142} These methods may be particularly beneficial for assessing the effect of

478 cumulative loads on injury, as both the relative importance of loads at different time points and

the time window over which previous loads may affect current risk of injury are unknown.

Flexible WCE methods are easily implemented using the *WCE* package in R.¹⁴³ Briefly, the user specifies the time-varying exposure, outcome, and covariates, the time window of interest, and the number of knots used for the cubic B-spline regression. A higher number of knots offers increased flexibility of the weight function.¹³⁹ The Akaike Information Criterion (AIC) or Bayesian Information Criterion (BIC) can be used to compare the fit of models with different knots and time windows.¹⁴² Below, I outline some technical details on how the weight functions are estimated.

487 Weighted cumulative exposure

488 The weighted cumulative exposure at time *u*, for individual *i*, is defined as:

489
$$WCE_i(u) = \sum_{t=1}^{u} w(u-t)X_i(t),$$

490 where $X_i(t)$ is the exposure for individual *i* at time *t*, *u*-*t* is the time elapsed since the exposure 491 $X_i(t)$, and w(u-t) is an estimated weight assigned to exposure at time *t*, based on time elapsed 492 since exposure *u*-*t*. The estimated weights quantify the relative importance of exposures that 493 occurred *u*-*t* weeks prior for the hazard of outcome at time *u*, with positive weights indicating
494 increased hazard and negative weights indicating decreased hazard.^{140,144} Weighted past

495 exposures are summed for each past time *t*, from the start of a user-selected time window of

496 relevant past exposure up to time u. The resulting $WCE_i(u)$ is a time-varying exposure metric,

497 calculated at each time u during follow-up, until the time of event or censoring.¹⁴¹

The weight function is estimated flexibly using cubic regression B-splines, avoiding having tospecify *a priori* the form of the function. The weight function is defined as:

500
$$w(u-t) = \sum_{j=1}^{m+4(-2)} \theta_j B_j(u-t)$$

501 where *m* is the number of interior knots within the estimated spline function which determines its

flexibility and degrees of freedom, B_i are the m+4(-2) functions in the B-spline basis, and θ_i are

503 coefficients that are estimated from the data.¹⁴¹

504 The time window over which the WCE function is modelled, [u - a, u], must also be specified.

505 Exposures occurring before this interval (t < u - a) are thought to be too far in the past to be

506 etiologically relevant to the risk of outcome at time *u*. The function can be constrained so that

507 weights smoothly decrease to zero at either end of the time window.¹⁴¹

509 Chapter 4: Manuscript 1

510 **Preface**

511 Sport injury epidemiology is a growing research area concerned with the prevention, diagnosis, 512 and management of injuries sustained during sport and physical activity. It plays an important 513 role in reducing injury incidence among individuals of all ages and backgrounds. While some 514 researchers in this field are medical professionals or otherwise have an extensive understanding 515 of the biological and clinical basis of sport injuries, this is not a given. In fact, sport injury 516 epidemiology is a field that can benefit from skilled methodologists with other substantive 517 backgrounds. 518 The most common type of sport injuries, particularly among children, are those affecting the

The most common type of sport injuries, particularly among cinidren, are mose affecting the musculoskeletal system. This manuscript is a narrative review on musculoskeletal sport injuries targeted toward epidemiologists without a substantive background in this field. It was written with the goal of providing a concise introductory overview for future researchers and methodologists interested in advancing this field of research.

523 This manuscript has been published in *Injury Epidemiology* (<u>https://doi.org/10.1186/s40621-024-</u>
524 <u>00507-3</u>).

526 Principles of musculoskeletal sport injuries for epidemiologists: a

527 review

- 528 Chinchin Wang^{1,2}, Steven D. Stovitz³, Jay S. Kaufman¹, Russell Steele⁴, Ian Shrier^{2,5}
- ¹Department of Epidemiology, Biostatistics and Occupational Health, McGill University,

530 Montreal, Canada

- ²Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, Montreal,
 Canada
- ³Department of Family Medicine and Community Health, University of Minnesota, Minneapolis,
 USA
- ⁴Department of Mathematics and Statistics, McGill University, Montreal, Canada
- ⁵Department of Family Medicine, McGill University, Montreal, Canada
- 537

538 **Corresponding author:**

- 539 Ian Shrier MD, PhD
- 540 Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, McGill
- 541 University, 3755 Côte Ste-Catherine Road, Montreal, Quebec, Canada H3T 1E2
- 542 Email: ian.shrier@mcgill.ca
- 543 Phone Number: 1-514-229-0114

545 Abstract

546 Background: Musculoskeletal injuries are a common occurrence in sport. The goal of sport
547 injury epidemiology is to study these injuries at a population level to inform their prevention and
548 treatment.

549 **Main Body:** This review provides an overview of musculoskeletal sport injuries and the

550 musculoskeletal system from a biological and epidemiologic perspective, including injury

551 mechanism, categorizations and types of sport injuries, healing, and subsequent injuries. It is

552 meant to provide a concise introductory substantive background of musculoskeletal sport injuries

553 for epidemiologists who may not have formal training in the underlying anatomy and

554 pathophysiology.

555 **Conclusion:** An understanding of sport injuries is important for researchers in sport injury 556 epidemiology when determining how to best define and assess their research questions and 557 measures.

558

559 Keywords: sport injuries, musculoskeletal system, epidemiology, strains and sprains

560 Background

561 Sport and physical activity are crucial to maintaining a healthy lifestyle for people of all ages. 562 Their wide-ranging benefits include prevention of chronic diseases, reduced morbidity and 563 mortality, and improved mental health.¹ However, participation in sports and physical activity 564 can also result in injury. Referred to as "sport injuries", they occur most commonly to the 565 musculoskeletal system that allows the human body to move. Sport injuries can result in 566 morbidity, predispose to further injuries, and decrease subsequent activity due to time lost during recovery or reduced desire to be active.² By studying the epidemiology of sport injuries, we can 567 568 inform their prevention and management at a population level.

569 Epidemiologists conducting research in the sport injury field may not have formal training in the

570 anatomy and pathophysiology of sport injuries and the musculoskeletal system. This review aims

571 to provide a concise introductory overview of musculoskeletal sport injuries for epidemiologists,

572 covering 1) definition of a sport injury from biological and epidemiological perspectives; 2)

573 common categorizations of sport injuries and subsequent injuries; 3) a summary of the

574 musculoskeletal systems and the injuries occurring to specific tissues and organs; 4) principles of

575 healing and rehabilitation of sport injuries.

576 Main text

577 What is a sport injury?

An injury is generally considered a sport injury when it occurs in relation to participation in sport or physical activity. Sport injuries occur most commonly to the musculoskeletal system³ and as such, we focus on musculoskeletal sport injuries in this review. However, sport participation can also result in injuries to other organ systems such as the neurological system (e.g. concussions, spinal cord injuries and peripheral nerve injuries), cardiovascular system (e.g. arrhythmias), and other systems.^{4,5} In this section, we provide definitions of injury from both the biological and epidemiological perspectives.

585 **Biological perspective**

- 586 Biologically, injuries are broadly defined as tissue damage.⁶ When an individual performs
- 587 activity, their body is exposed to various forces. For simplicity, we will refer to the sum of
- 588 these forces as load.⁶ Load causes tissues to undergo deformation. Upon deformation, tissue
- 589 cells try to keep or restore themselves to their original state, causing an internal resistance
- 590 known as stress.⁶⁻⁸ Loads beyond a tissue's load capacity (or load tolerance) will cause
- 591 excessive stress leading to tissue damage (Figure 1A).^{6,8}
- 592 Tissue damage can occur to varying extents. Large loads beyond a tissue's load capacity will 593 cause immediate tissue damage. The amount of damage can usually be visualized without a 594 microscope, and may result in physical symptoms (e.g. pain) and limitations of tissue 595 function.^{6,8} Healing involves structural changes, where damaged tissue may be repaired or replaced with scar tissue.⁹ Smaller loads that are close to or minimally above a tissue's load 596 597 capacity can cause microdamage or microtrauma, referring to microscopic tears or cracks within a cellular membrane that can be viewed using traditional light microscopy.^{6,8,10–12} While 598 599 generally considered an injury in a biological sense, when microdamage is isolated, tissues are 600 usually able to restore tissue integrity without scar tissue, and without most of the physical 601 symptoms of injury.⁸ Given sufficient recovery time, structural adaptations in response to microdamage lead to increased tissue strength and load capacity (Figure 1B).^{13,14} However, 602 603 repeated microdamage without sufficient recovery can result in injury, limiting tissue function and resulting in physical symptoms and scar tissue formation (Figure 1C).^{6,8,9} 604

605 Damage can also be restricted to the internal structure of a tissue cell, which might be only 606 visible using electron microscopy.¹² This includes damage at the cell cytoskeleton level.^{15,16} 607 The cytoskeleton is responsible for cellular structure and stability, and plays crucial roles in 608 cell movement, division, and intracellular transport.¹⁶ Damage to the cytoskeleton might impair transport of important molecules, leading to decreased cell function.¹⁶ This damage may or 609 610 may not be considered an injury. For instance, concussion injuries usually involve a decrease in neuronal brain function without visible microscopic damage.¹⁷ While the underlying 611 612 mechanisms for the decreased function are not fully understood, this is consistent with the hypothesis of cell cytoskeleton damage.¹⁸ 613

- 614 Finally, fatigue is the loss of tissue strength upon repeated loading and associated
- 615 microdamage. While not considered an injury in itself, it results in decreased load capacity,

616 which can make a tissue more susceptible to injury.⁶ Muscle fatigue specifically refers to a

617 reduction in the maximal force or ability to maintain maximal force of the muscles due to

618 repeated use, and is reversible with rest.¹⁹ It occurs due to impairments in the contractile

619 proteins and structures that allow muscles to generate force, which can be considered as

620 damage at the cell cytoskeleton level.^{20,21}

621 Epidemiologic perspective

The International Olympic Committee consensus definition of injury for surveillance and epidemiologic studies is "tissue damage or other derangement of normal physical function due to participation in sports, resulting from rapid or repetitive transfer of kinetic energy".²² This is distinct from their definition of illness, which is a complaint or disorder where the primary mode does not involve the transfer of kinetic energy.²² Illnesses include disorders resulting from the loss of vital elements (e.g. dehydration),²² or from the external environment (e.g. heat stroke) during sport, among others.

Injuries often present as pain or other physical symptoms such as aches, soreness, stiffness, or deformities that affect normal physical function.^{23–25} As such, injuries are sometimes recorded as any patient-reported symptom or complaint of the musculoskeletal system due to physical activity.^{22,26} However, the perception of pain can differ between individuals based on age, sex, and level of activity. ^{23,27,28} Further, injuries can occur without pain (e.g., microdamage), and pain can be present independent of tissue damage.²³ Therefore, the same underlying biological damage in different individuals may not be similarly defined as an injury.

636 Some athletes and researchers use a more restricted definition of injury compared to the

637 consensus definition. Many athletes (particularly elite) consider pain and other physical

638 symptoms as a normal part of sport participation.^{25,29,30} They perceive an injury as a condition

that must preclude them from performing at their normal or optimal level, beyond the

640 experience of pain, which results in altered or missed participation from sport.^{25,29}

641 Common definitions of injury in epidemiologic studies are "any complaint", encompassing all

642 injuries whether the complaint is symptom- or performance-based; "medical attention injuries",

or injuries where medical attention was sought; and "time-loss injuries", or injuries causing the

- athlete to be unable to complete a current or future activity session. ²² Medical attention
- 645 injuries may or may not limit an athlete's ability to participate in sport, and not all time-loss
- 646 injuries may be reported, nor require or result in medical attention.²² The optimal definition for
- 647 a sport injury depends on the research question of interest. For example, researchers may be
- 648 interested in tissue damage as evidenced by microscopy, a medical diagnosis, a patient-
- 649 reported complaint, or time lost from sport participation.
- 650 Although diagnosing injuries is beyond the scope of this article, a positive diagnostic test for an
- 651 injury could occur due to 1) an actual injury; 2) a false positive test (i.e. test illustrates abnormal
- 652 morphology when it is normal); or 3) a misunderstanding of what is "abnormal" (i.e. test
- 653 accurately illustrates abnormal morphology which is not an actual injury). For example, over
- 654 30% of individuals over 50 years who have not had back pain (i.e. uninjured) will have disc
- 655 herniation on a magnetic resonance image.³¹ Epidemiologists must avoid conflating abnormal
- 656 tests with clinical diagnoses, especially in small, individual clinic-based studies.

657 Categorizations of musculoskeletal sport injuries

658 Musculoskeletal sport injuries are often categorized by characteristics such as their mode of

onset, severity, and anatomical and tissue location in research and surveillance. These

660 categorizations can be used to study the prevention or occurrence of specific groups of injuries.

661 Common categorizations are summarized in Table 1 and discussed further below, and in "The

662 musculoskeletal system" section.

Category	Categorization	Definition
Mode of onset	Acute vs. overuse	 <u>Acute</u>: Sudden onset related to a specific inciting event <u>Overuse</u>: Gradual onset, may or may not be related to a specific inciting event
	• Traumatic vs. atraumatic	 <u>Traumatic</u>: Related to a specific traumatic inciting event <u>Atraumatic</u>: Not related to a specific traumatic inciting event
Severity	Duration of time-loss	• Time from which sport participation is ceased until full return to participation
	Duration of symptoms	• Time from onset of symptoms (e.g., pain) until symptoms cease
	Acute vs. chronic	 <u>Acute</u>: symptoms present for a short period of time <u>Chronic</u>: symptoms present for a long period of time The period of time distinguishing acute vs chronic often varies according to tissue type and anatomical location
	• Severity of symptoms	Self-reported severity of symptoms, assessed using a scale or scoring system
	Functional consequences	• Self-reported or clinician assessed functional consequences of injury, assessed using a scale or scoring system
	Amount of tissue damage	 Higher degree injuries represent greater severity, although specific definitions may differ depending on the injury type and location <u>1st degree:</u> Least severe (e.g. minimal damage or tearing with minimal symptoms) <u>2nd degree:</u> Moderately severe (e.g. visible damage or partial tearing with symptoms) <u>3rd degree:</u> Most severe (e.g. complete tearing or rupture of tissue)
Anatomical location	Body part or region of injury	 <u>Upper extremity</u>: Shoulder, upper arm, elbow, forearm, wrist, hand <u>Lower extremity</u>: Hip, groin, thigh, knee, lower leg, ankle, foot, <u>Trunk</u>: Chest, abdomen, thoracic spine, lumbar spine <u>Head and neck</u>: Head, neck
Tissue type	Injuries by tissue type	 <u>Bone:</u> fracture <u>Muscle:</u> strain <u>Ligaments:</u> sprain <u>Tendon:</u> tendinopathy, tendinosis, tendinitis, partial or complete rupture a different meanings. We refer readers to the text for discussion of the limitations of these categorizations

Table 1. Common categorizations of sport injuries in sport injury epidemiology.

Different categorizations may use the same terms with different meanings. We refer readers to the text for discussion of the limitations of these categorizations.

665 Mode of onset

A common categorization of sport injuries is by their mode of onset or mechanism of injury. 666 Injuries are often categorized as acute versus overuse, although these definitions are not always 667 consistent. Acute injuries have a sudden onset related to a specific event.²² Biologically, this 668 669 occurs when an undamaged tissue is subjected to a sudden load beyond its load capacity.³² One 670 such injury is a broken bone (i.e. fracture) occurring upon a fall. Overuse injuries, on the other 671 hand, occur gradually due to repetitive loading and associated microdamage to a tissue, without a specific inciting event.²² For example, an athlete might develop considerable pain in 672 673 the calf muscle (i.e. strain) occurring gradually with repeated physical activity that impedes 674 further activity. The term "overuse" implies that these injuries occur due to excessive activity 675 beyond what the tissues are prepared for from previous activity and loading. Instead, we could 676 consider that these injuries occur from previous "underuse" of the tissues, whereby the 677 individual has not been sufficiently active previously and has not developed the load capacity within their tissues to handle this level of activity.³³ 678

Despite these distinctions, there are grey zones and limitations to this categorization. Some injuries with specific inciting events may have been due to underlying "overuse" (i.e. repeated loading and microdamage without sufficient recovery) rather than a sudden load. An injury may occur if the damaged tissue is subjected to a load that would normally be tolerated if not for the microdamage and reduced load capacity (Figure 1C).²² While these injuries occur acutely, their underlying mechanism is consistent with that of overuse.

685 Another closely-related categorization that is sometimes used in the literature is traumatic 686 (similar to and often used as a synonym for acute) or atraumatic (a.k.a. non-traumatic, similar 687 to overuse).³⁴ Like acute injuries, traumatic injuries have a sudden onset related to a specific 688 event. Atraumatic injuries occur without a specific inciting event. Injuries occurring alongside 689 a specific event but that are due to underlying microdamage are considered traumatic under 690 this categorization. For instance, a bone can be weakened due to repetitive activity and 691 associated microdamage without any symptoms. The bone might fracture due to a sudden 692 force, such as pushing off during a sprint race. This might be considered an acute injury

- 693 because it occurred alongside a specific inciting event, but atraumatic because there was no
- 694 direct trauma.



A. Musculoskeletal tissues become injured when exposed to loads greater than their load capacity

B. Adequate recovery after activity increases a tissue's load capacity



C. Injuries can result from normal loads when tissues are fatigued or inadequately recovered



696 Figure 1. Relationship between a musculoskeletal tissue's load capacity, load, and injury.

697 The red coloured bars represent load on a tissue, while the blue coloured bars represent the

tissue's load capacity. **A.** Tissues exposed to loads lower than their load capacity experience

microdamage (panel a), while tissues exposed to loads greater than their load capacity
 experience immediate tissue damage and injury (panel b). B. Given sufficient recovery,

701 microdamage from loads below a tissue's capacity (panel a) will result in strengthening and

increased load capacity (panel b). **C.** Tissues without sufficient recovery (panel a) have a

703 decreased load capacity (panel b) which can lead to immediate tissue damage and injury even

with normal loads (panel c).

705

706 Injury severity

Injuries can also be described by their severity. Severity may have different meanings for
different athletes (e.g., the need for medication or surgery),³⁵ but is often defined as the duration
of injury, particularly duration of time lost from participation.²² Time loss is assessed from the
date that an athlete begins to have altered or missed participation from sport (which may not be

the date of injury onset from a biological standpoint), until the date that they are fully able to

712 participate.²² Time loss can underestimate severity if an athlete returns to sport before the injury

is resolved, or overestimate severity if an athlete does not resume their normal participation upon

healing (e.g., because their fitness and performance is not at a competitive level).²²

715 A similar measure to injury duration is the duration of symptoms. While defined as the amount 716 of time that symptoms are present, it is sometimes dichotomized as acute versus chronic. Under 717 this categorization, acute injuries refer to recent injuries, while chronic injuries refer to injuries 718 where symptoms have been present for an extended period of time. For example, some 719 categorize back pain as chronic when symptoms have been present longer than 12 weeks, and as acute when symptoms have been present for less than 12 weeks.^{36,37} Some authors use chronic as 720 a synonym for overuse when describing mode of onset;³⁸ however, these are separate concepts.²² 721 722 Symptom duration may be longer than time loss duration if an athlete returns to full participation 723 with lingering symptoms, or shorter if an athlete does not resume their normal participation at the 724 time their symptoms cease.

725 Injury severity can also be described by the severity of symptoms, functional consequences, or a

composite score (i.e. patient-reported outcome measure), as self-reported by athletes^{20,24,37} or

assessed by a clinician. Severity can also be assessed through functional tests. These measures

are often assessed using scales or scoring systems.^{22,26,39}

Finally, strain (muscle) and sprain (ligament) injuries are often graded as 1st, 2nd, or 3rd degree

730 based on amount of tissue damage and/or clinical symptoms, with 1st degree being the least

731 severe and 3rd degree being the most severe.^{6,40,41} However, we note that certain injuries are

732 graded based on factors other than severity of symptoms or tissue damage. These categorizations

are reviewed in "The musculoskeletal system" section.

734 Injury anatomical location

735 Injuries are often grouped by their anatomical location for summary purposes. These groups may 736 be broad (e.g., upper extremity, lower extremity, trunk) or specific (e.g., knee, lower leg, ankle) depending on the context.⁴² While reporting specific anatomical locations is recommended for 737 738 injury surveillance programs,²² different injuries to the same location can have very different 739 clinical presentations and outcomes. Although specific locations are important for clinicians who 740 are recommending treatment, epidemiologists often focus on broader categories. These broader 741 categories result in larger sample sizes with potentially greater generalizability, although they have limitations when considering specific injuries.^{22,35} 742

743 The musculoskeletal system and its injuries by tissue type

The musculoskeletal system is made up of distinct tissues that function together to provide
shape, stability, and movement to the human body. It plays a central role in the ability for
humans to do sport and physical activity.⁴³

747 The major connective tissues of the musculoskeletal system are bone, cartilage, skeletal muscle, tendons, and ligaments. Bones are the structural basis for the human skeleton.⁴⁴ The attachment 748 749 points of adjacent bones are known as joints. Joints allow for movement of bones with respect to each other.⁴⁴ Joints contain cartilage, and are surrounded by ligaments, which are tissue bands 750 that physically attach bones to other bones.⁴⁵ Muscles pass over joints, and are attached to two 751 different bones by tissue bands called tendons.⁴⁴ Muscles can be thought of as elastics that 752 753 shorten (i.e., when contracting) and lengthen (i.e., when relaxing). When a muscle shortens, their 754 attachment sites to each bone come closer together. In general, this leads to movement at the 755 joint.44

The following section expands on each location and tissue, and summarizes the injuries that canoccur to them.

758 **Bones**

- 759 Bones make up the human skeleton, which provides structure and support for the body.⁴⁴ They
- 760 provide a rigid attachment point for muscles, which allow bones to move.⁴⁴ Bones also protect
- the internal organs, produce blood cells, and store and release minerals and fats.^{44,46}

762 Bone composition

Bones are comprised of two types of bone tissue: cortical, and cancellous. Cortical bone is dense and forms a protective outer layer around cancellous bone, which is spongey and responsible for absorbing the load transmitted to bones.^{6,10,44} The thickness of cortical bone and the relative distribution of cortical and cancellous bone tissue differs between different locations within a bone and between bones.¹⁰ Bones are surrounded by an outer fibrous sheath called the periosteum. The periosteum contains nerves and blood vessels, and plays a role in bone remodelling both regularly and after injury.^{44,47}

770 In the long bone of the limbs, the end of a bone is termed the epiphysis, and primarily consists of cancellous bone.⁴⁴ The main shaft of a bone is the diaphysis, and consists primarily of cortical 771 bone.⁴⁴ The metaphysis lies between the diaphysis and epiphyses, and consists primarily of 772 cancellous bone.⁴⁴ Between the metaphysis and the epiphyses is the epiphyseal plate (growth 773 774 plate) in children, which is a region of bone growth in long bones such as the tibia (shin bone) or femur (thigh bone).⁴⁸ While the epiphyseal plate is composed of cartilage during childhood and 775 adolescence, it calcifies into bone tissue after growth has completed.^{10,44} Damage to the 776 777 epiphyseal plate can result in the slowing or stopping of growth for the affected bone, which can lead to angular deformities or asymmetry in lengths of the lower limbs.^{49,50} 778

779 Bone injuries

Bones respond to load by thickening in mass and strengthening.⁴⁴ However, forces beyond the
load capacity may result in injury to the bone, called a fracture.¹⁰ Among athletes, fractures
generally occur after a sudden traumatic event such as a fall, contact with another athlete, or
contact with an object (e.g. ball travelling with speed).⁵¹

Stress fractures are distinct from regular fractures in that they do not arise from a single
traumatic event, but rather repeated exposure to load (i.e., overuse injury).^{8,52} Bones are normally

in a state of remodelling, where bone tissue breaks down (resorption) and is replaced by new
tissue.^{52,53} Remodelling increases in response to load and associated microdamage.^{52,53} Without
adequate recovery time, more bone tissue is resorbed than deposited, creating stress fractures that
appear as small cracks.^{52,53} While most stress fractures will heal with adequate rest and
rehabilitation,⁵⁴ some may require surgery.⁵⁵ Continued loading onto a stress fracture without
adequate recovery may result in a complete fracture.⁵⁵

792 Two special types of fractures that occur in children are greenstick and buckle fractures. Greenstick fractures only affect one side of the bone, creating a crack that does not extend 793 through the entire bone and causing bending of the bone rather than a full break.⁵⁶ This may be 794 due to increased cartilage content and compliance of young bones compared to adult bones, or 795 796 because the periosteum sheath surrounding a bone is more elastic in children than in adults, decreasing the likelihood of complete fractures.⁴⁷ Greenstick fractures occur most commonly 797 after a fall.⁵⁶ Buckle fractures, also referred to as torus fractures, occur due to compression of 798 cortical bone that creates a bulge on one side of the bone, without a full break.⁵⁷ These fractures 799 800 are common among children and are generally simple to treat through immobilization.⁵⁷

801 Muscles

Muscles are the core component of the human muscular system, which is responsible for generating force and movement.⁵⁸ While there are three types of muscle (smooth, cardiac, and skeletal), only skeletal muscles are responsible for movement of the human body.⁵⁸

805 Muscle composition

Skeletal muscle tissue is comprised of individual muscle fibres.⁴⁴ The patterns in which the
muscle fibres are organized help determine the strength and velocity of the muscular
contraction.^{8,44} Muscle is the tissue that is the most capable of strengthening in response to load.⁸
This occurs primarily through hypertrophy, in which muscle fibres increase their cross-sectional
area.⁸ Skeletal muscle tissue also contains structures that convey information about muscle
tension and position.⁵⁹

812 Muscle injuries

813 There are several types of muscle injuries. Muscle tissue will become damaged when the applied 814 load exceeds its load capacity, either through a single event or repetitive microdamage. These 815 injuries are referred to as strains (or pulled muscles by the general public) when caused by stretching or contraction forces.⁸ Strains are the most common sport injury.^{8,40} The amount of 816 817 damage can range from tearing of a few muscle fibres with minimal loss of strength (1st degree) to visible partial tearing of the muscle tissue (2^{nd} degree) to a complete tear/rupture of the muscle 818 (3rd degree).^{40,41} Damage to muscle tissue may also damage the structures within the tissue 819 820 responsible for sensing tension and position through the same mechanisms. For instance, a 821 stretching of the muscle will also stretch these structures. As such, balance and position sense are often disrupted in muscle strains.⁶⁰ 822

Muscles can also be injured by compressive forces that exceed the load capacity, typically through a direct blow to the muscle. These injuries are referred to as contusions.^{8,61} Contusions are often associated with rupture of blood vessels, causing internal bleeding that results in a bruise.^{8,61} Internal bleeding can lead to various clinical consequences, including some conditions that result in permanent disability.^{8,61}

One particular consequence of unaccustomed activity is delayed onset muscle soreness (DOMS), which is sometimes used as a proxy for injury.⁶² DOMS presents as soreness, stiffness, or pain that follows 1 to 3 days after unaccustomed activity.^{8,63} Although frequently studied in animal models, its mechanism is not entirely understood. It is unclear whether findings from studies on DOMS are generalizable to muscle injuries.¹²

833 Tendons

Tendons represent connective tissue that physically connects muscles to bones.⁴⁴ They enable the
 transmission of force from muscle to bone, and help stabilize joints.^{8,64}

836 **Tendon composition**

837 Tendons are composed of dense collagen fibres aligned in the same direction as the muscle
838 fibres.^{8,64} Tendons contain few elastic fibres,⁸ which causes them to experience only a small

change in length for a large amount of force compared to muscle.⁶⁴ The muscle-tendon interface
 contains receptors that sense and transmit information about forces within the tendon.⁶⁵

841 **Tendon injuries**

Tendons will stiffen when lengthened in response to increasing load but become damaged under excessive load.⁶⁴ Sudden injuries to the tendons are known as tears or ruptures. These can range in severity from partial tearing/rupture of the tissue to complete tearing/rupture.^{8,64}

Tendons can also experience overuse injuries.⁸ Repetitive loading and microdamage to tendons
result in chronic pain, known as tendinopathy.⁸ Tendinopathies include tendinosis in cases with
tissue degeneration, and tendinitis in cases with tendon inflammation.^{6,8,66}

848 Most musculoskeletal injuries occur at the junction where the muscle joins the tendon.⁶⁷

849 Although the tendon may be involved, these are generally considered muscle injuries. In general,

tendon injuries that are not close to the muscle-tendon junction often have poor blood supply,

resulting in longer recovery times for tendon-specific injuries compared to muscle injuries.^{68,69}

852 Joints

A joint is the point where two or more bones connect. Joints may provide stability or support movement depending on their type and composition.⁴⁴

855 Joint composition

856 There are three types of joints that differ in composition and function: fibrous, cartilaginous, and857 synovial joints.

Fibrous joints are fixed, generally immobile joints comprised of dense collagen rather than

cartilage, and are found in the skull among other locations.⁷⁰ Cartilaginous joints are joined by

860 fibrocartilage or hyaline cartilage (see "Cartilage" section) and are slightly mobile or immobile.⁷¹

861 The epiphyseal plate in long bones, which connects the diaphysis and epiphyses in childhood and

adolescence, is considered a cartilaginous joint.⁷¹ They are found where the right pelvis joins the

863 left pelvis, among other locations.⁷²

864 Synovial joints are the most common joints in the human body and include the major joints of the limbs (e.g. knee, elbow, shoulder).⁴⁴ They are mobile and are comprised of a joint cavity, 865 consisting of the ends of the bones that are covered by articular cartilage.^{73,74} The joint cavity is 866 867 surrounded by fibrous tissue known as a joint capsule, and lined with a synovial membrane that secretes fluid to keep the joint lubricated.^{44,74,75} The joint capsule seals the joint cavity, keeping 868 869 synovial fluid inside, and provides stability by limiting joint movements and preventing bones 870 from separating.⁷⁵ As synovial joints allow for movement, they are the most commonly 871 implicated in sport injuries.⁶

872 Joints can be categorized by the types of movement they allow. Joints can move in three planes: sagittal (longitudinal), frontal (coronal), and transverse (axial).⁵⁴ Movement in the sagittal plane 873 874 is seen from the side of the body (e.g. knee flexion and extension), movement in the frontal plane 875 is seen from the front of the body (e.g. hip abduction and adduction), and movement in the transverse plane is seen from above (e.g. hip rotation).⁵⁴ Hinge (uniaxial) joints are a type of 876 synovial joint where most movement occurs in a single plane, and are found in the elbow and 877 knee.^{76,77} Biaxial joints often experience movement in two planes, and include the 878 metacarpophalangeal (finger knuckle) joints.⁷⁸ Multiaxial joints often experience movement in 879 three planes, and include the hip and shoulder joints.^{79,80} 880

881 Joint injuries

Joint dislocations are a common injury that occur when the bones that connect at the joint are

displaced, resulting in immediate pain and limited range of motion.^{81,82} Dislocations typically

occur through a sudden traumatic force such as a collision or fall, and occur commonly in the

shoulder and elbow among athletes.^{81,82} Dislocations can cause damage to the ligaments,

cartilage, and bones.⁸¹ Dislocations are treated by physical manipulation of the joint back into its

normal location, followed by a recovery period often involving immobilization to heal tissue

damage.^{81,82} Recurrent dislocations in some joints (e.g. shoulder) are common among athletes.⁸¹

889 Subluxations are partial joint dislocations where the connecting bones do not completely

890 separate. Unlike full dislocations, subluxations sometimes spontaneously "relocate" to their

891 original position without physical manipulation.⁸³ Joint injuries also include specific injuries to

892 cartilage and ligaments. These injuries are covered in their respective sections.

893 Cartilage

894 Cartilage is an important connective tissue mainly present in joints.⁴⁴ Cartilage is weaker and

895 more flexible than bone. However, it is still weight-bearing and resilient.⁴⁴ There are three types

896 of cartilage: hyaline, fibrocartilage, and elastic.⁴⁴ Elastic cartilage, which is present in the ear and

⁸⁹⁷ larynx, is not considered a component of the musculoskeletal system.⁸⁴

898 Hyaline cartilage composition

899 Hyaline cartilage is the most common cartilage in the human body.⁸⁵ It does not contain any

900 nerves or blood vessels, and is limited in its ability to repair itself following damage.⁸⁶ It is found

901 inside joints covering the ends of adjacent bones, where it is referred to as articular cartilage.⁸⁵

902 Articular cartilage is a highly specialized tissue that reduces friction and provides a smooth 903 surface for movement at the joints.⁸⁶ It redistributes pressure across bones to minimize high 904 pressure point loads that could cause bone swelling and injury.⁸⁶ Its nutrition comes from 905 molecules dissolved in the normal joint fluid (synovial fluid). As the joint moves, synovial fluid 906 circulates and distributes nutrients.⁸⁶ When joint movement is restricted, for instance due to 907 injury or casting, cartilage nutrition is impaired.^{86,87} Immobilizing a joint is one method of 908 creating osteoarthritis in animals.⁸⁸

909 Hyaline cartilage injury

910 Similar to bone, hyaline cartilage can be injured through a single traumatic event.⁸⁹ Furthermore,

911 extensive damage to articular cartilage with insufficient repair leads to unequal redistribution of

912 forces within the joint. One possible consequence of articular damage is post-traumatic

913 osteoarthritis, a condition characterized by joint pain, dysfunction, and malalignment.⁸⁹ Although

914 articular cartilage itself is not visible on x-rays, insufficient repair may lead to a decreased

915 cartilage thickness, causing the two bones of a joint to appear much closer together than normal

916 on an x-ray. This is called joint space narrowing, and is an important sign for clinically

917 meaningful osteoarthritis.⁹⁰

918 Fibrocartilage composition

919 Fibrocartilage is a stronger and denser type of cartilage than hyaline cartilage.⁴⁴ Unlike hyaline

920 cartilage, it contains nerves and blood vessels, but only at its periphery.⁸ It is typically found in

921 larger joints, and functions to absorb and distribute forces more evenly across bones.⁸

922 Fibrocartilage tissue is present in the meniscus of the knee, and the labrum in the hip and

923 shoulder.^{8,44} It also forms part of the intervertebral discs that lie between the bones of the lumbar

924 spine.⁴⁴

925 Fibrocartilage Injury

926 Damage to fibrocartilage occurs in meniscal and labral injuries (tears). Acute meniscal tears

927 occur due to trauma to the knee, and can occur in isolation, or alongside injury to ligaments.⁹¹

928 Degenerative meniscal tears are more common with increasing age and increased loading (e.g.

929 weight-bearing activities).^{92,93} As we age, the meniscus becomes weaker and more susceptible to

930 tears with low loads. Labral tears in the hip and shoulder can also occur from trauma or

931 degeneration from repetitive loading.^{94–96} Hip labral tears are associated with certain types of

932 abnormal hip morphologies.⁹⁶

933 Ligaments

Ligaments represent connective tissue that physically connects bones, spanning a joint.⁴⁵ They 934 are often just local prominent thickenings of the joint capsule tissue that run from one bone to the 935 other, with a different tissue composition.⁷⁵ However, some ligaments exist inside or outside of 936 the joint capsule.⁷⁵ Their primary function is to stabilize joints and prevent excessive movement 937 at the joint.^{45,54} However, ligaments also play an important role in proprioception as they contain 938 939 nerve endings that convey information about joint position and movement that are necessary to 940 coordinate contractions by different muscles during movement.⁸ Ligaments stretch out in 941 response to low amounts of load, but will resist movement when pulled tight in response to further load, thus preventing further movement of the joint.^{45,54} However, ligaments will tear if 942 stretched too far, causing injury.45,64 943

944 Ligament composition

- 945 Ligaments are primarily composed of dense collagen fibres, with only small amounts of elastic
- 946 fibres.^{45,54} They are generally strong and stiff, with limited elasticity.⁵⁴ The degree of stiffness
- 947 differs by the relative composition of collagen versus elastic fibres and other components.⁹⁷ The
- 948 stiffness of a ligament also increases as the ligament is stretched.⁴⁵

949 Ligament injuries

- 950 While ligaments have a relatively high load tolerance, excessive load will cause damage and
- 951 injury to the ligaments and other joint structures.^{45,64} Injuries to ligaments are referred to as
- 952 sprains.⁶ These can range in severity from some tissue damage with minimal symptoms (1st
- 953 degree), partial tear of the ligament (2nd degree), to complete tearing/rupture and separation of
- 954 the damaged ends of the ligament (3^{rd} degree). Grade definitions and terminology can differ by
- 955 injury type. For instance, there are three ligaments that are implicated in a lateral ankle sprain.
- 956 Some categorizations use 3rd degree ankle sprain to refer to the complete tearing of each
- 957 ligament, while some consider a 3rd degree ankle sprain to mean all three ligaments are
- 958 damaged.^{98,99} Surgery may be recommended for some complete ligament tears but not others.^{100–}
- ¹⁰² Because damage to a ligament may damage local nerves, proprioception, balance and position
- 960 sense are often disrupted in sprains.⁸
- Nearly all ligament injuries occur due to a single event.⁶ However, ligaments and other joint
 structures may also experience microdamage when subjected to repetitive loading.^{45,97} This
 microdamage is usually asymptomatic, but may affect joint stability and predispose individuals
 to other injuries.⁸

965 Healing and rehabilitation of injury

966 Biological and clinical perspective

967 Injury healing occurs in three overlapping phases: 1) inflammation; 2) repair or regeneration; and

- 968 3) remodelling.¹⁰³ Briefly, inflammation causes damaged cells and tissues to degenerate.^{103,104}
- 969 Tissue repair or regeneration replaces damaged cells with new cells. Finally, the repaired or
- 970 regenerated tissue is remodelled to regain optimal strength and function in a process that can take
- 971 months to years.^{103,104} The specific healing process of an injury depends on the type of tissue that

972 is damaged and the degree of damage. Bone heals by a regenerative process, whereby the healed tissue is the same as the original bone tissue.⁸ Other musculoskeletal tissues heal by a 973 974 regenerative process when there is less severe tissue damage or microdamage. However, severe 975 tissue damage heals by a repair process, where the healed tissue is a scar tissue rather than the original tissue.^{8,105,106} While scar tissue is initially weak, its strength increases during the repair 976 and remodelling process until it is close to that of the original uninjured tissue.⁴⁰ Extensive scar 977 978 tissue formation due to tissue bleeding and inflammation may result in decreased tissue strength 979 and increased risk of recurrent injuries (discussed below).⁸

Tissue strength can be increased during the repair and remodelling process by applying progressive loads that cause microdamage and subsequent adaptation, but remain below the load capacity for injury.^{14,107–112} Overloading may lead to further tissue damage and disrupt the healing process. For instance, low loads during a recovery period may still be large enough to cause further tissue damage because of the reduced load capacity.¹⁴ Just as overloading may cause tissue damage, underloading or extensive immobilization may prevent tissues from strengthening and cause tissue atrophy.^{107,110,111,113}

987 Clinicians usually prescribe rehabilitation therapy for injuries. Rehabilitation refers to restoring the tissue to its preinjury state, and involves many components that are not always well described 988 in studies.¹¹⁴ Most injury rehabilitation programs start with reducing pain and preventing excess 989 990 bleeding and inflammation, reducing scar tissue formation. Additionally, injuries often result in 991 decreased range of motion, strength and proprioception. Therefore, exercises including those for 992 balance, strengthening, and stretching are often prescribed to specifically address these limitations.¹¹³ Other components of rehabilitation may include electronic modalities (e.g. 993 994 ultrasound), manual therapy (e.g. massage, mobilizations, manipulations), and prevention 995 education. Finally, patients are recommended to gradually return to participation in sport once 996 they are largely symptom-free and have regained adequate strength to minimize injury recurrence.¹¹³ 997

998 Epidemiological perspective

999 Injury healing can be assessed under different definitions, which will affect calculated injury 1000 durations. Ideally, an injury is considered healed when the athlete is able to return to their 1001 previous amount of activity without pain. When detailed data are available, healing date can be 1002 determined from clinical records or self-report of symptoms. Alternatively, researchers might use 1003 the date of last treatment for injury, under the assumption that treatment is only provided while 1004 an injury remains unhealed.¹¹⁵ However, the decision to stop treatment has subjectivity.¹¹⁵ As 1005 such, clinical data may not always minimize error or bias in the measurement of healing.

1006 Unfortunately, detailed data are often unavailable and researchers often operationalize the

1007 healing date of an injury as the date of full return to participation or play in sport.^{22,115} However,

1008 many athletes return to activity while they are still symptomatic, which could result in

1009 measurement error if utilizing return to play or even medical clearance date as the healing date.

1010 Epidemiologists should recognize that return to participation decisions can vary between

1011 athletes, coaches, and clinicians, and may not necessarily reflect biological healing.^{22,115}

1012 Subsequent and recurrent injuries

1013 Initial injuries may predispose to subsequent injuries due to muscle imbalances, deficits in

1014 strength and proprioception, or changes in biomechanics.¹¹⁶ Subsequent injuries to the same

1015 location account for a considerable proportion (10-25%) of all injuries.¹¹⁵

1016 Researchers must consider how to define and account for subsequent injuries, particularly in the 1017 longitudinal follow-up of athletes. Subsequent injuries generally refer to injuries that occur after 1018 an initial index injury. Subsequent injuries to a different body part are considered "subsequent 1019 new injuries".¹¹⁵ Subsequent injuries to the same body part but a different tissue type are called 1020 "local injuries". Finally, subsequent injuries to the same body part and tissue type are called 1021 "www.executionic."

1021 "recurrent injuries".

1022 Recurrent injuries can be exacerbations or re-injuries. An "exacerbation" is a worsening of an

1023 index injury that was not fully healed or recovered.^{22,115} A "re-injury" is a recurrent injury that

1024 occurs to the same location and tissue as an index injury that was fully healed or recovered.²²

1025 Recurrences are sometimes further categorized by the time that they occurred following healing

of the index injury (early: within 2 months; late: within 2 to 12 months; or delayed: more than 12
 months).¹¹⁵

1028 These categorizations have limitations. For instance, it can be unclear whether a subsequent injury is related or not related to the initial injury.¹¹⁷ Further, the definition of healing within an 1029 1030 epidemiologic study will affect whether an injury is considered a re-injury or an exacerbation. 1031 This will in turn affect the overall injury count, risk, rate, and length of time loss. Consider an individual who suffers an index injury on January 1st and returns to participation on January 10th 1032 1033 but continues to experience symptoms and receive medical treatment. They then experience a worsening of symptoms on January 30th. If healing is defined as return to participation, the 1034 1035 individual will be considered as having had two injuries (an index injury and a 1036 subsequent/recurrent injury), each with a separate length of time loss with the sum being the total 1037 time lost. If healing is defined as date of last treatment or by cessation of symptoms, the 1038 individual will be considered as having one injury (an index injury and an exacerbation), with a 1039 longer time loss (equal to the total time lost). While there is no consensus for the optimal 1040 definition of healing, researchers would benefit from clearly defining their outcomes. Further, 1041 when synthesizing and interpreting findings from multiple studies, researchers should ensure that 1042 aggregated results use similar definitions.

1043 Conclusion

Sport injuries are a concern for anyone participating in physical activity. Applying epidemiologic methods can greatly contribute to determining how to best prevent and treat sport injuries and their related morbidities. Understanding what constitutes a sport injury from a biological and epidemiologic perspective is important for researchers in these fields, who must determine how to best define and assess their research questions and measures.

References

- 1. Warburton DER, Nicol CW, Bredin SSD. Health benefits of physical activity: the evidence. CMAJ. 2006 Mar 14;174(6):801–9.
- Emery CA, Pasanen K. Current trends in sport injury prevention. Best Pract Res Clin Rheumatol. 2019 Feb 1;33(1):3–15.
- Patel DR, Baker RJ. Musculoskeletal Injuries in Sports. Prim Care Clin Off Pract. 2006 Jun 1;33(2):545–79.
- Toth C, McNeil S, Feasby T. Central Nervous System Injuries in Sport and Recreation. Sports Med. 2005 Aug 1;35(8):685–715.
- Maron BJ, Pelliccia A. The Heart of Trained Athletes. Circulation. 2006 Oct 10;114(15):1633–44.
- Bartlett R, Bussey M. Sports Biomechanics: Reducing Injury Risk and Improving Sports Performance [Internet]. Florence, UNITED STATES: Routledge; 2013 [cited 2019 Dec 21]. Available from:

http://ebookcentral.proquest.com/lib/mcgill/detail.action?docID=957394

- Gómez-González M, Latorre E, Arroyo M, Trepat X. Measuring mechanical stress in living tissues. Nat Rev Phys. 2020 Jun;2(6):300–17.
- Bahr R, Alfredson H, Järvinen M, Järvinen T, Khan K, Kjær M, et al. Types and Causes of Injuries. In: The IOC Manual of Sports Injuries [Internet]. John Wiley & Sons, Ltd; 2012 [cited 2023 Feb 15]. p. 1–24. Available from: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118467947.ch1
- 9. Baoge L, Van Den Steen E, Rimbaut S, Philips N, Witvrouw E, Almqvist KF, et al. Treatment of skeletal muscle injury: a review. ISRN Orthop. 2012;2012:689012.
- Radasch RM. Biomechanics of Bone and Fractures. Vet Clin North Am Small Anim Pract. 1999 Sep 1;29(5):1045–82.
- Stauber T, Blache U, Snedeker JG. Tendon tissue microdamage and the limits of intrinsic repair. Matrix Biol. 2020 Jan 1;85–86:68–79.
- Lieber RL. Biomechanical response of skeletal muscle to eccentric contractions. J Sport Health Sci. 2018 Jul;7(3):294–309.

- Wiesinger HP, Kösters A, Müller E, Seynnes OR. Effects of Increased Loading on In Vivo Tendon Properties: A Systematic Review. Med Sci Sports Exerc. 2015 Sep;47(9):1885– 95.
- Bahr R, Cook J, Langberg H, MacAuley D, Matheson G, Mæhlum S. Treating Sports Injuries. In: The IOC Manual of Sports Injuries [Internet]. John Wiley & Sons, Ltd; 2012 [cited 2023 Feb 27]. p. 25–39. Available from: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118467947.ch2
- 15. Butterfield TA. Eccentric Exercise In Vivo: Strain-Induced Muscle Damage and Adaptation in a Stable System. Exerc Sport Sci Rev. 2010 Apr;38(2):51.
- Gefen A, Weihs D. Cytoskeleton and plasma-membrane damage resulting from exposure to sustained deformations: A review of the mechanobiology of chronic wounds. Med Eng Phys. 2016 Sep 1;38(9):828–33.
- Signoretti S, Lazzarino G, Tavazzi B, Vagnozzi R. The Pathophysiology of Concussion. PM&R. 2011;3(10S2):S359–68.
- Park E, Baker AJ. The Pathophysiology of Concussion. In: Schweizer TA, Baker AJ, editors. Tackling the Concussion Epidemic: A Bench to Bedside Approach [Internet]. Cham: Springer International Publishing; 2022 [cited 2023 Apr 27]. p. 25–41. Available from: https://doi.org/10.1007/978-3-030-93813-0_2
- Enoka RM, Duchateau J. Muscle fatigue: what, why and how it influences muscle function. J Physiol. 2008 Jan 1;586(Pt 1):11–23.
- Westerblad H, Duty S, Allen DG. Intracellular calcium concentration during lowfrequency fatigue in isolated single fibers of mouse skeletal muscle. J Appl Physiol Bethesda Md 1985. 1993 Jul;75(1):382–8.
- Westerblad H, Lee JA, Lamb AG, Bolsover SR, Allen DG. Spatial gradients of intracellular calcium in skeletal muscle during fatigue. Pflugers Arch. 1990 Mar;415(6):734–40.
- 22. Bahr R, Clarsen B, Derman W, Dvorak J, Emery CA, Finch CF, et al. International Olympic Committee consensus statement: methods for recording and reporting of epidemiological data on injury and illness in sport 2020 (including STROBE Extension for Sport Injury and Illness Surveillance (STROBE-SIIS)). Br J Sports Med [Internet]. 2020

Feb 18 [cited 2020 Mar 16]; Available from:

https://bjsm.bmj.com/content/early/2020/02/28/bjsports-2019-101969

- Hainline B, Turner JA, Caneiro JP, Stewart M, Moseley GL. Pain in elite athletes neurophysiological, biomechanical and psychosocial considerations: a narrative review. Br J Sports Med. 2017 Sep 1;51(17):1259–64.
- Clarsen B, Myklebust G, Bahr R. Development and validation of a new method for the registration of overuse injuries in sports injury epidemiology: the Oslo Sports Trauma Research Centre (OSTRC) Overuse Injury Questionnaire. Br J Sports Med. 2013 May 1;47(8):495–502.
- 25. Bolling C, Barboza SD, Mechelen W van, Pasman HR. How elite athletes, coaches, and physiotherapists perceive a sports injury. Transl Sports Med. 2019;2(1):17–23.
- 26. Clarsen B, Bahr R, Myklebust G, Andersson SH, Docking SI, Drew M, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. Br J Sports Med. 2020 Apr 1;54(7):390–6.
- Tesarz J, Schuster AK, Hartmann M, Gerhardt A, Eich W. Pain perception in athletes compared to normally active controls: A systematic review with meta-analysis. PAIN. 2012 Jun;153(6):1253–62.
- Racine M, Tousignant-Laflamme Y, Kloda LA, Dion D, Dupuis G, Choinière M. A systematic literature review of 10 years of research on sex/gender and pain perception part 2: do biopsychosocial factors alter pain sensitivity differently in women and men? Pain. 2012 Mar;153(3):619–35.
- 29. Verhagen E, Warsen M, Bolling CS. 'I JUST WANT TO RUN': how recreational runners perceive and deal with injuries. BMJ Open Sport Exerc Med. 2021 Sep 1;7(3):e001117.
- Wiese-bjornstal DM, Smith AM, Shaffer SM, Morrey MA. An integrated model of response to sport injury: Psychological and sociological dynamics. J Appl Sport Psychol. 1998 Mar 1;10(1):46–69.
- Deyo RA, Mirza SK. Herniated Lumbar Intervertebral Disk. N Engl J Med. 2016 May 5;374(18):1763–72.
- McBain K, Shrier I, Shultz R, Meeuwisse WH, Klügl M, Garza D, et al. Prevention of sports injury I: a systematic review of applied biomechanics and physiology outcomes research. Br J Sports Med. 2012 Mar 1;46(3):169–73.

- 33. Stovitz SD, Johnson RJ. "Underuse" as a cause for musculoskeletal injuries: is it time that we started reframing our message? Br J Sports Med. 2006 Sep 1;40(9):738–9.
- Merkel DL, Molony JT. RECOGNITION AND MANAGEMENT OF TRAUMATIC SPORTS INJURIES IN THE SKELETALLY IMMATURE ATHLETE. Int J Sports Phys Ther. 2012 Dec;7(6):691–704.
- Mountjoy M, Junge A, Alonso J, Clarsen B, Pluim B, Shrier I, et al. Consensus statement on the methodology of injury and illness surveillance in FINA (aquatic sports): Table 1. Br J Sports Med. 2015 Nov 27;50:bjsports-2015.
- Flint JH, Wade AM, Giuliani J, Rue JP. Defining the Terms Acute and Chronic in Orthopaedic Sports Injuries: A Systematic Review. Am J Sports Med. 2014 Jan 1;42(1):235–41.
- Chou R, Cotton D, Cotton D, Rao JK, Taichman D, Williams S. Low Back Pain. Ann Intern Med. 2014 Jun 3;160(11):ITC6-1.
- Lavallee ME, Balam T. An Overview of Strength Training Injuries: Acute and Chronic. Curr Sports Med Rep. 2010 Oct;9(5):307.
- Stevenson M, Segui-Gomez M, Lescohier I, Scala CD, McDonald-Smith G. An overview of the injury severity score and the new injury severity score. Inj Prev. 2001 Mar 1;7(1):10–3.
- 40. Järvinen TAH, Kääriäinen M, Järvinen M, Kalimo H. Muscle strain injuries. Curr Opin Rheumatol. 2000 Mar;12(2):155.
- Noonan TJ, Garrett WEJ. Muscle Strain Injury: Diagnosis and Treatment. JAAOS J Am Acad Orthop Surg. 1999 Aug;7(4):262.
- 42. Habelt S, Hasler CC, Steinbrück K, Majewski M. Sport injuries in adolescents. Orthop Rev. 2011 Nov 7;3(2):e18.
- Wind H, Gouttebarge V, Kuijer PPFM, Frings-Dresen MHW. Assessment of Functional Capacity of the Musculoskeletal System in the Context of Work, Daily Living, and Sport: A Systematic Review. J Occup Rehabil. 2005 Jun 1;15(2):253–72.
- 44. Klenerman L. The skeleton and its attachments. In: Klenerman L, editor. Human Anatomy: A Very Short Introduction [Internet]. Oxford University Press; 2015 [cited 2023 Jan 25]. p. 0. Available from: https://doi.org/10.1093/actrade/9780198707370.003.0002

- Frank CB. Ligament structure, physiology and function. J Musculoskelet Neuronal Interact. 2004 Jun;4(2):199–201.
- Florencio-Silva R, Sasso GR da S, Sasso-Cerri E, Simões MJ, Cerri PS. Biology of Bone Tissue: Structure, Function, and Factors That Influence Bone Cells. BioMed Res Int. 2015 Jul 13;2015:e421746.
- Maia Ferreira Alencar CH, Sampaio Silveira CR, Cavalcante MM, Maia Vieira CG, Diógenes Teixeira MJ, Neto FA, et al. Periosteum: An imaging review. Eur J Radiol Open. 2020 Jan 1;7:100249.
- 48. Jorge AAL, Grimberg A, Dattani MT, Baron J. 11 Disorders of Childhood Growth. In: Sperling MA, editor. Sperling Pediatric Endocrinology (Fifth Edition) [Internet]. Philadelphia: Elsevier; 2021 [cited 2024 Feb 19]. p. 299–356. Available from: https://www.sciencedirect.com/science/article/pii/B9780323625203000117
- Caine D, DiFiori J, Maffulli N. Physeal injuries in children's and youth sports: reasons for concern? Br J Sports Med. 2006 Sep 1;40(9):749–60.
- Jones C, Wolf M, Herman M. Acute and Chronic Growth Plate Injuries. Pediatr Rev. 2017 Mar 1;38(3):129–38.
- 51. Court-Brown CM, Wood AM, Aitken S. The epidemiology of acute sports-related fractures in adults. Injury. 2008 Dec 1;39(12):1365–72.
- 52. Saunier J, Chapurlat R. Stress fracture in athletes. Joint Bone Spine. 2018 May 1;85(3):307–10.
- Hughes JM, Popp KL, Yanovich R, Bouxsein ML, Matheny RW. The role of adaptive bone formation in the etiology of stress fracture. Exp Biol Med. 2017 May;242(9):897– 906.
- Milner C. Functional Anatomy for Sport and Exercise: Quick Reference. London: Routledge; 2008. 160 p.
- 55. Shindle MK, Endo Y, Warren RF, Lane JM, Helfet DL, Schwartz EN, et al. Stress Fractures About the Tibia, Foot, and Ankle. JAAOS - J Am Acad Orthop Surg. 2012 Mar;20(3):167.
- Noonan KJ, Price CT. Forearm and Distal Radius Fractures in Children. JAAOS J Am Acad Orthop Surg. 1998 Jun;6(3):146.

- 57. Sharp JW, Edwards RM. Core curriculum illustration: pediatric buckle fracture of the distal radius. Emerg Radiol. 2019 Aug 1;26(4):483–4.
- Frontera WR, Ochala J. Skeletal Muscle: A Brief Review of Structure and Function. Calcif Tissue Int. 2015 Mar 1;96(3):183–95.
- Macefield VG, Knellwolf TP. Functional properties of human muscle spindles. J Neurophysiol. 2018 Aug;120(2):452–67.
- Kröger S. Proprioception 2.0: novel functions for muscle spindles. Curr Opin Neurol. 2018 Oct;31(5):592.
- 61. Beiner JM, Jokl P. Muscle Contusion Injuries: Current Treatment Options. JAAOS J Am Acad Orthop Surg. 2001 Aug;9(4):227.
- 62. Drew MK, Finch CF. The Relationship Between Training Load and Injury, Illness and Soreness: A Systematic and Literature Review. Sports Med. 2016 Jun 1;46(6):861–83.
- Cheung K, Hume PA, Maxwell L. Delayed Onset Muscle Soreness. Sports Med. 2003 Feb 1;33(2):145–64.
- 64. Gracey E, Burssens A, Cambré I, Schett G, Lories R, McInnes IB, et al. Tendon and ligament mechanical loading in the pathogenesis of inflammatory arthritis. Nat Rev Rheumatol. 2020 Apr;16(4):193–207.
- Maas H, Noort W, Smilde HA, Vincent JA, Nardelli P, Cope TC. Detection of epimuscular myofascial forces by Golgi tendon organs. Exp Brain Res. 2022;240(1):147– 58.
- Bass E. Tendinopathy: Why the Difference Between Tendinitis and Tendinosis Matters. Int J Ther Massage Bodyw. 2012 Mar 31;5(1):14–7.
- Gimigliano F, Resmini G, Moretti A, Aulicino M, Gargiulo F, Gimigliano A, et al. Epidemiology of Musculoskeletal Injuries in Adult Athletes: A Scoping Review. Medicina (Mex). 2021 Oct;57(10):1118.
- Sharma P, Maffulli N. Biology of tendon injury: healing, modeling and remodeling. J Musculoskelet Neuronal Interact. 2006;6(2):181–90.
- 69. Ahmed IM, Lagopoulos M, McConnell P, Soames RW, Sefton GK. Blood supply of the achilles tendon. J Orthop Res. 1998;16(5):591–6.
- Miroshnichenko K, Liu L, Tsukrov I, Li Y. Mechanical model of suture joints with fibrous connective layer. J Mech Phys Solids. 2018 Feb 1;111:490–502.

- 71. Juneja P, Munjal A, Hubbard JB. Anatomy, Joints. In: StatPearls [Internet]. Treasure Island (FL): StatPearls Publishing; 2022 [cited 2023 Feb 2]. Available from: http://www.ncbi.nlm.nih.gov/books/NBK507893/
- 72. Becker I, Woodley SJ, Stringer MD. The adult human pubic symphysis: a systematic review. J Anat. 2010;217(5):475–87.
- 73. Pacifici M, Koyama E, Iwamoto M. Mechanisms of synovial joint and articular cartilage formation: Recent advances, but many lingering mysteries. Birth Defects Res Part C Embryo Today Rev. 2005;75(3):237–48.
- Hui AY, McCarty WJ, Masuda K, Firestein GS, Sah RL. A systems biology approach to synovial joint lubrication in health, injury, and disease. WIREs Syst Biol Med. 2012;4(1):15–37.
- Ralphs JR, Benjamin M. The joint capsule: structure, composition, ageing and disease. J Anat. 1994 Jun;184(Pt 3):503–9.
- Fornalski S, Gupta R, Lee TQ. Anatomy and Biomechanics of the Elbow Joint. Sports Med Arthrosc Rev. 2003 Mar;11(1):1.
- 77. Hefzy MS, Grood ES. Review of Knee Models. Appl Mech Rev. 1988 Jan 1;41(1):1–13.
- 78. Batmanabane M, Malathi S. Movements at the carpometacarpal and metacarpophalangeal joints of the hand and their effect on the dimensions of the articular ends of the metacarpal bones. Anat Rec. 1985;213(1):102–10.
- Bowman KF, Fox J, Sekiya JK. A Clinically Relevant Review of Hip Biomechanics. Arthrosc J Arthrosc Relat Surg. 2010 Aug 1;26(8):1118–29.
- Peat M. Functional Anatomy of the Shoulder Complex. Phys Ther. 1986 Dec 1;66(12):1855–65.
- Skelley NW, McCormick JJ, Smith MV. In-game Management of Common Joint Dislocations. Sports Health. 2014 May;6(3):246–55.
- Hodge DK, Safran MR. Sideline Management of Common Dislocations. Curr Sports Med Rep. 2002 Jun;1(3):149.
- 83. Aronen JG. Anterior Shoulder Dislocations in Sports. Sports Med. 1986 May 1;3(3):224–34.
- Pollard TD, Earnshaw WC, Lippincott-Schwartz J, Johnson GT, editors. Chapter 32 -Connective Tissues. In: Cell Biology (Third Edition) [Internet]. Elsevier; 2017 [cited 2023

May 17]. p. 555–70. Available from:

https://www.sciencedirect.com/science/article/pii/B9780323341264000323

- 85. Krishnan Y, Grodzinsky AJ. Cartilage diseases. Matrix Biol. 2018 Oct 1;71–72:51–69.
- Sophia Fox AJ, Bedi A, Rodeo SA. The Basic Science of Articular Cartilage. Sports Health. 2009 Nov;1(6):461–8.
- Wang Y, Wei L, Zeng L, He D, Wei X. Nutrition and degeneration of articular cartilage. Knee Surg Sports Traumatol Arthrosc. 2013 Aug 1;21(8):1751–62.
- Palmoski MJ, Brandt KD. Running Inhibits the Reversal of Atrophic Changes in Canine Knee Cartilage After Removal of a Leg Cast. Arthritis Rheum. 1981;24(11):1329–37.
- Buckwalter JA. Sports, Joint Injury, and Posttraumatic Osteoarthritis. J Orthop Sports Phys Ther. 2003 Oct;33(10):578–88.
- 90. Chan WP, Huang GS, Hsu SM, Chang YC, Ho WP. Radiographic joint space narrowing in osteoarthritis of the knee: relationship to meniscal tears and duration of pain. Skeletal Radiol. 2008;37(10):917–22.
- Lohmander LS, Englund PM, Dahl LL, Roos EM. The Long-term Consequence of Anterior Cruciate Ligament and Meniscus Injuries: Osteoarthritis. Am J Sports Med. 2007 Oct 1;35(10):1756–69.
- Adams BG, Houston MN, Cameron KL. The Epidemiology of Meniscus Injury. Sports Med Arthrosc Rev. 2021 Sep;29(3):e24.
- Howell R, Kumar NS, Patel N, Tom J. Degenerative meniscus: Pathogenesis, diagnosis, and treatment options. World J Orthop. 2014 Nov 18;5(5):597–602.
- 94. Keener JD, Brophy RH. Superior Labral Tears of the Shoulder: Pathogenesis, Evaluation, and Treatment. JAAOS J Am Acad Orthop Surg. 2009 Oct;17(10):627.
- 95. O'Reilly OC, Behrens MB, Gulbrandsen TR, Wolf BR. Posterior Shoulder Instability and Labral Pathology. In: Werner BC, editor. MRI-Arthroscopy Correlations: A Case-Based Atlas of the Knee, Shoulder, Elbow, Hip and Ankle [Internet]. Cham: Springer International Publishing; 2022 [cited 2024 Mar 2]. p. 183–99. Available from: https://doi.org/10.1007/978-3-030-94789-7_13
- 96. Banerjee P, Mclean CR. Femoroacetabular impingement: a review of diagnosis and management. Curr Rev Musculoskelet Med. 2011 Mar 1;4(1):23–32.

- Solomonow M. Ligaments: a source of work-related musculoskeletal disorders. J Electromyogr Kinesiol. 2004 Feb 1;14(1):49–60.
- Wolfe MW, Uhl TL, Mattacola CG, Mccluskey LC. Management of Ankle Sprains. Am Fam Physician. 2001 Jan 1;63(1):93–105.
- Lacerda D, Pacheco D, Rocha AT, Diniz P, Pedro I, Pinto FG. Current Concept Review: State of Acute Lateral Ankle Injury Classification Systems. J Foot Ankle Surg. 2023 Jan 1;62(1):197–203.
- Beynnon BD, Johnson RJ, Abate JA, Fleming BC, Nichols CE. Treatment of Anterior Cruciate Ligament Injuries, Part I. Am J Sports Med. 2005 Oct 1;33(10):1579–602.
- Miyamoto RG, Bosco JA, Sherman OH. Treatment of Medial Collateral Ligament Injuries. JAAOS - J Am Acad Orthop Surg. 2009 Mar;17(3):152.
- 102. Stone AV, Marx S, Conley CW. Management of Partial Tears of the Anterior Cruciate Ligament: A Review of the Anatomy, Diagnosis, and Treatment. JAAOS - J Am Acad Orthop Surg. 2021 Jan 15;29(2):60.
- Pieters D, Wezenbeek E, Schuermans J, Witvrouw E. Return to Play After a Hamstring Strain Injury: It is Time to Consider Natural Healing. Sports Med. 2021 Oct 1;51(10):2067–77.
- 104. Bigham-Sadegh A, Oryan A. Basic concepts regarding fracture healing and the current options and future directions in managing bone fractures. Int Wound J. 2015;12(3):238–47.
- 105. Liu J, Saul D, Böker KO, Ernst J, Lehman W, Schilling AF. Current Methods for Skeletal Muscle Tissue Repair and Regeneration. BioMed Res Int. 2018 Apr 16;2018:e1984879.
- 106. Leong NL, Kator JL, Clemens TL, James A, Enamoto-Iwamoto M, Jiang J. Tendon and Ligament Healing and Current Approaches to Tendon and Ligament Regeneration. J Orthop Res. 2020;38(1):7–12.
- 107. Magnusson SP, Kjaer M. The impact of loading, unloading, ageing and injury on the human tendon. J Physiol. 2019;597(5):1283–98.
- 108. Kiviranta I, Tammi M, Jurvelin J, Säämänen AM, Helminen HJ. Moderate running exercise augments glycosaminoglycans and thickness of articular cartilage in the knee joint of young beagle dogs. J Orthop Res. 1988;6(2):188–95.

- Tipton CM, Matthes RD, Maynard JA, Carey RA. The influence of physical activity on ligaments and tendons. Med Sci Sports. 1975;7(3):165–75.
- 110. Liphardt AM, Godonou ET, Dreiner M, Mündermann A, Tascilar K, Djalal N, et al. Immobilization by 21 days of bed rest results in type II collagen degradation in healthy individuals. Osteoarthritis Cartilage. 2024 Feb;32(2):177–86.
- Sun HB. Mechanical loading, cartilage degradation, and arthritis. Ann N Y Acad Sci. 2010 Nov;1211:37–50.
- Killian ML, Cavinatto L, Galatz LM, Thomopoulos S. The role of mechanobiology in tendon healing. J Shoulder Elbow Surg. 2012 Feb 1;21(2):228–37.
- Fournier M. Principles of Rehabilitation and Return to Sports Following Injury. Clin Podiatr Med Surg. 2015 Apr 1;32(2):261–8.
- 114. Wade DT, Smeets RJEM, Verbunt JA. Research in rehabilitation medicine: Methodological challenges. J Clin Epidemiol. 2010 Jul 1;63(7):699–704.
- Hamilton G, Meeuwisse W, Emery C, Shrier I. Subsequent Injury Definition, Classification, and Consequence. Clin J Sport Med. 2011 Nov 1;21(6):508–14.
- 116. Fulton J, Wright K, Kelly M, Zebrosky B, Zanis M, Drvol C, et al. Injury risk is altered by previous injury: a systematic review of the literature and presentation of causative neuromuscular factors. Int J Sports Phys Ther. 2014 Oct;9(5):583–95.
- Shrier I, Steele RJ. Classification systems for reinjuries: a continuing challenge. Br J Sports Med. 2014 Sep 1;48(18):1338–9.

Declarations

Ethics approval and consent to participate

Not applicable

Consent for publication

Not applicable

Availability of data and materials

Not applicable

Competing interests

The authors declare that they have no competing interests.

Funding

This research did not receive any specific funding.

Authors' contributions

CW and IS conceived the manuscript. CW drafted the initial manuscript. IS, JK, RS, and SS critically reviewed and edited the manuscript. All authors read and approved the final manuscript.

Acknowledgements

Not applicable
Chapter 5: Manuscript 2

Preface

In Chapter 2, I discussed strategies for summarizing previous and current levels of physical activity using the acute:chronic workload ratio framework. The vast majority of studies investigating the relationship between changes in physical activity and injury have summarized activity occurring over particular time windows using either unweighted or exponentially weighted moving averages. Both these approaches make assumptions about the associations between physical activity performed at different timepoints and current injury risk.

In this manuscript, we explore the use of flexible weighted cumulative exposure methods to summarize previous levels of physical activity and determine their cumulative association with current injury risk.

This manuscript has been submitted to *Epidemiology*. Conference abstracts based on contents of this manuscript were presented at the Society for Epidemiology Annual Meeting (Chicago, 2022), and the Canadian Academy of Sports and Exercise Medicine Annual Symposium (Banff, 2023) as poster presentations.

Time-varying associations between physical activity and injury risk among children

Chinchin Wang, MSc, PhD(c)^{a,b}; Michal Abrahamowicz, PhD^{a,c}; Marie-Eve Beauchamp, PhD^c; Jay S. Kaufman, PhD^a; Russell Steele, PhD^d; Eva Jespersen, PhD^{e,f}; Niels Wedderkopp, MD, PhD^f; Ian Shrier, MD, PhD^b

Affiliations: ^aDepartment of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada; ^bCentre for Clinical Epidemiology, Lady Davis Institute for Medical Research, Jewish General Hospital, Montreal, Quebec, Canada; ^cCentre for Outcomes Research and Evaluation, Research Institute of the McGill University Health Centre, Montreal, Quebec, Canada; ^dDepartment of Mathematics and Statistics, McGill University, Montreal, Quebec, Canada; ^eDepartment of Oncology, Odense University Hospital, Odense, Denmark; ^fDepartment of Clinical Research, University of Southern Denmark, Odense, Denmark

Corresponding Author: Ian Shrier; Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, McGill University, 3755 Côte Ste-Catherine Road, Montreal, Quebec, Canada; [ian.shrier@mcgill.ca]; 1-514-229-0114

Running head: Physical activity and injury risk in children

Conflict of Interests: The authors declare no conflicts of interest.

Sources of Funding: No specific funding support was secured for this study. Salary support for Chinchin Wang was funded through a Canadian Institute of Health Research (CIHR) Doctoral Research Award.

Data and Code Availability: Data and code are available upon request.

Acknowledgements: N/A

Abstract

Background: Physical activity has time-varying associations with injury risk among children. While previous activity may predispose to injury through tissue damage, fatigue, and insufficient recovery, it may protect against injury by strengthening tissues and improving fitness and skills. It is unclear what the relevant time window and relative importance of past activity are with regard to current injury risk in children. The objectives of this study were to assess how previous activity patterns are associated with injury risk among children.

Methods: Our data source was a prospective cohort study of Danish school children conducted between 2008 and 2014. We applied flexible weighted cumulative exposure methods within a Cox proportional hazards model to estimate the time-varying association between number of weekly activity sessions and time-to-first injury in each school year. We estimated several models with varying time windows and compared goodness-of-fit.

Results: The best-fitting model included 20 weeks of past physical activity. Higher levels of activity performed 10 to 20 weeks ago were associated with decreased injury risk, while higher levels of activity performed 2 to 9 weeks ago were associated with increased injury risk. The estimated injury hazard ratio was 1.63 (95% CI: 1.18, 2.23) for children who were highly active in the past 10 weeks after being minimally active 11 to 20 weeks ago, relative to those who were minimally active for the past 20 weeks.

Conclusions: Flexible weighted cumulative exposure methods provide insight into the timespecific associations between physical activity history and injury in children.

Keywords: child, adolescent, exercise, wounds and injuries, fatigue

1 Introduction

2 Physical activity plays an important role in the health and development of children and

3 adolescents.^{1,2} However, physical activity also increases likelihood of injury and related

4 morbidities.² Understanding how different activity patterns are associated with injury risk is an

5 important step in injury prevention.

The association between physical activity and injury is time-varying,^{3,4} but not necessarily 6 7 linear.¹⁴ When an individual performs activity, their tissues are exposed to physical stresses 8 known as load.⁵ Loads below a tissue's load capacity cause microscopic damage, while loads 9 above the load capacity cause significant damage defined as injury.^{5,6} Although tissues 10 strengthen and increase their load capacity given sufficient recovery time, repeated microscopic damage decreases load capacity, increasing susceptibility to overuse injury.^{5,6} Large amounts of 11 physical activity without sufficient recovery also cause fatigue.⁷ Fatigue results in decreased load 12 capacity,⁵ and affects physical capacities including balance and proprioception.^{8–11} As such, 13 14 fatigue increases susceptibility to overuse and acute injuries. Previous physical activity can protect against injury by strengthening tissues and improving fitness and skill,^{12,13} or predispose 15

16 to injury through fatigue.

17 It is unclear what the relevant time window and relative importance of past activity at different 18 time points are regarding injury risk. Research on the association between loads and injury risk 19 has largely been based on the acute:chronic workload ratio (ACWR), defined as the ratio of current (acute) load to previous (chronic) load.^{15,16} Chronic load is most commonly measured 20 over 4 weeks, although windows of up to 8 weeks have been explored.^{15,17} ACWR-based studies 21 have summarized chronic loads as an unweighted average, or an exponentially weighted moving 22 average (EWMA).¹⁵ The unweighted average assumes that activity performed in each previous 23 24 week is equally associated with current injury risk, while EWMA weights assume that recent 25 activity has a greater association with current injury risk.

26 Flexible weighted cumulative exposure (WCE) methods are an alternative for estimating the

27 cumulative effect of a time-varying exposure without imposing a pre-specified (e.g. exponential)

28 function.¹⁸ The WCE metric is a weighted sum of past exposures, combining information about

29 duration, intensity, and timing into a summary metric.¹⁸ The objective of this study is to

30 determine the relative importance of activity in past weeks on current injury risk using flexible

- 31 WCE methods. Specifically, we aim to determine how previous activity patterns are associated
- 32 with injury risk in a cohort of Danish schoolchildren.

33 Methods

34 Data Source

- 35 Our data source was the Childhood Health, Activity, and Motor Performance School Study
- 36 Denmark (CHAMPS), a prospective cohort study conducted between 2008 and 2014.
- 37 Schoolchildren aged 6 to 15 were followed during each school year.²⁵ Each week, parents
- 38 reported the number of recreational activity sessions their child participated in over the past week
- 39 using a short message service text. Parents also reported whether their child experienced any
- 40 musculoskeletal pain, and whether pain was new or continuing from a previous injury (see
- 41 Supplementary Material for specific questions). Children with new pains were offered to be
- 42 examined by a clinician, who would establish a diagnosis if warranted.²⁵
- 43 All children in the 10 primary schools that agreed to participate were eligible for the parent
- 44 study. Participants could enter or leave the study at any time. Missing activity data were imputed
- 45 by random hot deck imputation with 5 datasets.²⁶ In total, 1,667 children were eligible for the
- 46 current study, contributing 7,296 schoolyears.
- 47 Ethics committee approval was obtained for the CHAMPS study (ID S20080047). The study was
- 48 registered with the Danish Data Protection Agency, as stipulated by the law J.nr. 2008-41-2240.
- 49 Parents provided written informed consent for study participation.

50 Exposure definition

51 The time-varying exposure was the number of activity sessions a child participated in within a 52 given week, including parent-reported recreational activity and physical education classes.

53 **Outcome definition**

54 The outcome was the first clinician-diagnosed injury (acute or overuse) within a schoolyear.

55 Weighted cumulative exposure

56 The weighted cumulative exposure at week *u*, for participant *i*, is defined as:

57
$$WCE_i(u) = \sum_{t=1}^{u} w(u-t)X_i(t),$$

58 where $X_i(t)$ is the number of activity sessions for participant *i* at week *t*, *u*-*t* is the time elapsed 59 since the exposure $X_i(t)$, and w(u-t) is an estimated weight assigned to exposure at week t, based 60 on time elapsed since exposure *u*-*t*. The estimated weights quantify the relative importance of 61 exposures that occurred *u*-*t* weeks prior for the hazard of injury at week *u*, with positive weights indicating increased hazard and negative weights decreased hazard.^{23,24} Weighted past exposures 62 63 are summed for each past week t, from the start of a user-selected time window of relevant past 64 exposure up to week u. The resulting $WCE_i(u)$ is a time-varying exposure metric, calculated at each week *u* during follow-up, until the end of the participant's follow-up, at the time of event or 65 censoring.¹⁸ 66

67 The weight function w(u-t) is estimated using flexible cubic regression B-splines.¹⁸ A higher 68 number of interior knots increases the number of spline functions, and therefore flexibility of the 69 weight function.¹⁸

70 Statistical Analysis

We fit multivariable Cox proportional hazards models for time to first injury, defined as the number of weeks elapsed until the first reporting of musculoskeletal pain leading to cliniciandiagnosed injury. Time zero was the start of each schoolyear. The time-varying exposure was the WCE metric for previous activity (up to 1 week ago), with activity in the current week (last 7 days), sex, and school-grade included as covariates. Children who did not get injured were censored at the end of each school year.

We estimated alternative WCE models using various combinations of (a) the number of interior knots *m* to be between 1 and 3, and (b) time windows of 5, 10, 15, 20, 25, and 30 past weeks of activity. Weight functions were constrained to smoothly decrease to zero at the week furthest in the past.¹⁸ We used the Bayesian Information Criterion (BIC)³⁰ to identify the best-fitting WCE

model.¹⁸ We employed likelihood ratio tests (LRT) and the Akaike Information Criterion (AIC) 81 82 to compare the fit of the best-fitting WCE model with (a) the conventional cumulative exposure 83 model that relied on the unweighted sum of exposures over the same time window, and (b) the Cox model that included covariates but excluded the time-varying exposure.¹⁸ The former 84 85 comparison assessed the usefulness of differential weighting of past exposures, while the latter 86 provided evidence that the estimated association was unlikely to only reflect sampling error. 87 Using the best-fitting model, we estimated hazard ratios (HRs) for injury comparing different 88 physical activity patterns. The 95% confidence bands for the weight function and the 95% 89 confidence intervals (CI) for the hazard ratios were estimated by cluster bootstrap (accounting for clustering within children who contributed several schoolyears) with 300 replicates.²⁹ 90 91 We conducted a sensitivity analysis where we included musculoskeletal pain reported in the

92 previous 4 weeks (not resulting in a clinician-diagnosed injury) as a binary time-varying

93 covariate. We also conducted subgroup analyses stratified by sex. We pre-specified a time

94 window of 20 weeks and 1 interior knot for sex-stratified analyses, based on the best-fitting

95 WCE model for the main analyses.

96 We tested for effect modification by sex by conducting separate analyses where time windows 97 and numbers of knots were allowed to vary by sex. We identified the best-fitting models for each 98 sex, and assessed to what extent these sex-specific weight functions improved fit compared to 99 the overall weight function by comparing deviances using LRTs and AIC (further details 90 provided in Supplementary Material).³¹

101 All results were averaged over the 5 imputed datasets. Analyses were conducted in R,³²

102 specifically the *WCE* package.³³

103 **Results**

104 Table 1 describes characteristics of the study population. Injuries occurred in 986 participants

105 (59%) at least once. A total of 1,752 first injuries in a given schoolyear were included in

106 analyses, with an incidence rate of 16.6 (95% CI: 15.8-17.4) per 1,000 person-weeks (16.8, 95%

107 CI: 15.8-17.9 among girls; 15.8, 95% CI: 16.8-17.9 among males). The average time-to-first

108 injury from the beginning of the schoolyear among injured participants was 18 weeks, with a

median of 15 weeks. The median number of activity sessions per week was 4 (interquartilerange: 3-5).

111 In WCE analyses, the best-fitting weight function for the association between physical activity 112 and time-to-first injury had a time window of 20 weeks and 1 interior knot (Table S1 displays 113 BICs for all models). The final WCE model improved the fit over the Cox model that excluded 114 the time-varying exposure (LRT p < 0.005 and smaller AICs across imputed datasets) and the 115 unweighted cumulative exposure model (LRT p<0.01 and smaller AICs across imputed 116 datasets), confirming the importance of differential weighting of past exposures. The best-fitting 117 weight function (Figure 1) suggested that higher levels of activity performed more than 10 weeks 118 ago were associated with decreased injury risk, while higher levels of activity 2 to 9 weeks ago 119 were associated with increased injury risk (peaking at 5 weeks ago). Activity performed one week ago was associated with a slight reduction in injury risk, with large uncertainty. The weight 120 121 function was qualitatively similar when recent musculoskeletal pain was included as a covariate 122 (Figure S1).

Table 2 displays HRs comparing different activity patterns. Consistently higher activity levels were associated with higher hazard of injury than consistently lower activity levels. Injury hazard was 60% greater for those who were highly active recently after only being minimally active 11 to 20 weeks ago relative to those who were consistently minimally active, and 30% greater for those who were moderately active recently. Injury hazard was 20% lower for those who were minimally active recently after being moderately active relative to those who were consistently moderately active.

Figure 2 displays sex-stratified weight functions. For both sexes, increased activity 3 to 9 weeks ago was associated with increased injury risk, peaking at 5 weeks. However, positive weights were somewhat greater but less spread over previous weeks for females. While activity performed over 9 weeks ago was associated with a lower injury risk among females, no decrease was observed for males.

The optimal sex-stratified weight functions both had 1 interior knot, and time windows of 15 weeks for females and 10 weeks for males (data not shown). While the optimal sex-stratified WCE models yielded slightly lower deviance, differences were not large enough to indicate a systematic improvement in goodness-of-fit over the common model for both sexes (LRT p-values between 0.10 and 0.17, with higher AICs).

140 **Discussion**

In this study, we applied a data-driven method to examine how activity patterns in previous weeks are associated with current injury risk in children. We found that higher levels of activity performed more than 10 weeks ago had a protective association with the risk of first injury in a schoolyear, whereas higher levels of activity performed between 2 and 10 weeks ago had an adverse association.

Unlike the ACWR framework,¹⁶ these results imply that previous activity does not necessarily 146 147 protect against injury. Indeed, recent loads may increase injury risk by causing microscopic damage and fatigue without adequate recovery.^{5,6} Biologically, activity performed more than 10 148 weeks ago might result in a decreased risk of injury by increasing tissue strength and load 149 capacity,⁵ as well as improving fitness, skill, and coordination.^{12,13} Given that an individual has 150 151 been active and uninjured for over 10 weeks, they have likely had sufficient recovery time and 152 are at lower risk of injury due to the long-term beneficial effects of activity. In contrast, an 153 individual who has only begun being active in the previous 10 weeks may not have developed 154 the load capacity and fitness to handle the same amount of activity, putting them at higher risk of 155 fatigue and injury. The protective association of activity performed more than 10 weeks ago may also be attributed to a "survivor effect",³⁴ where only those who were uninjured for a 20-week 156 157 interval are observed (i.e. uncensored) for the entire 20-week time window. Those who are more 158 susceptible to injury may have been censored prior to achieving 20 weeks of activity, and would 159 not contribute to the protective associations observed in the 10-to-20 week time window.

We also found limited evidence for a protective association of activity performed one week previous, with high uncertainty. This may be because those who were able to do more activity one week ago are unlikely to have experienced early symptoms of injury (e.g., soreness, pain) or fatigue. These individuals may be less likely to be injured in the current week than those who were unable to do as much activity due to early symptoms of injury experienced prior to clinician diagnosis. Although we did not observe qualitative differences with our weight functions when the presence of recent musculoskeletal pain was included as a covariate in sensitivity analyses,
 other factors such as fatigue may also result in decreased activity and increased risk of injury.³⁵

168 Our analyses stratified by sex did not meaningfully differ from our overall analyses; however, 169 we may not have had sufficient power to detect meaningful differences. These analyses 170 suggested that patterns of physical activity may be more strongly associated with injury risk 171 among females than males. We similarly found in a previous study with the same data source 172 that large increases in activity in the current week were more strongly associated with injury risk for females than for males.³⁶ This is consistent with other studies that found higher rates of 173 musculoskeletal pains and overuse injuries among females than males,^{37–39} which some have 174 hypothesized might be partially attributed to biological differences in hormones or anatomy.⁴⁰ It 175 176 may also be that males are self-regulating their activity more than females to avoid injury,

177 resulting in smaller associations between previous activity and injury risk.

Based on the best-fitting WCE model, we found that physical activity done up to 20 weeks ago is
associated with injury risk in the current week. This contrasts with the most common
formulations of the ACWR used to assess injury risk based on increases in load, which assume
equal weight for each previous week of activity. The conventional ACWR is defined as the ratio
of load in the current week (acute load) to the average load over the current and previous 3
weeks (chronic load).⁴

184 Instead, some researchers have employed an exponentially weighted moving average (EWMA) function to summarize loads before calculating the ACWR.^{28,41} The EWMA gives exponentially 185 decreasing weights for activity done further in the past.^{14,28} Typically, the EWMA function is 186 187 employed for both the acute load and the chronic load, and assumes that all activity performed in 188 the current week is adverse and that all activity performed in previous weeks is protective against 189 injury. As such, when considering previous activity, activity done one week ago would have the 190 greatest protective association with current injury risk, whereas activity done 20 weeks ago 191 would have only a slight protective association (Supplementary Material, Figure S1). The 192 assumption that all previous activity has a protective association is inconsistent with the 193 biological principle that recent activity can result in reduced load capacity and fatigue if there is

insufficient recovery.⁵ In contrast, our results suggested that activity performed in the previous 3
to 4 weeks is associated with increased risk of injury in our study population.

196 The cumulative effect of load on injury risk among elite youth handball players has previously been explored using distributed lag non-linear models (DLNMs).⁴² Like WCE models, DLNMs 197 use flexible weight functions used to represent the cumulative effect of a time-varying exposure 198 on an outcome.^{43,44} They were originally developed for use in time series analysis, but have been 199 extended to other contexts.⁴⁴ Unlike the current study, the handball study used daily session 200 ratings of perceived exertion as a proxy for load, with a previous time window of 28 days. Their 201 202 outcome of interest was health problems, which includes injuries, illnesses, and pains or soreness that may not result in clinician-diagnosed injuries.⁴⁵ The authors did not condition on activity 203 204 done on the current day; rather, it was included in their lag function. They found, with high 205 uncertainty, an increased risk of health problems on the current day with increasing activity, and protective associations for activity performed further than 6 days ago.⁴² Their shorter period of 206 207 adverse associations compared to our period of 10 weeks may be attributed to differences in 208 exposure and outcome definitions, time window of analysis (limited to 28 days), as well as the 209 study population. The handball study focused on elite athletes, who may be accustomed to 210 increases in activity and have shorter tissue recovery periods relative to our population of 211 schoolchildren.

212 Strengths and Limitations

This study employed a large dataset to assess how previous activity patterns are associated with injury risk in children and adolescents. It is one of few studies to explore the use of non-linear methods to express the cumulative effect of physical activity on injury, and the first to use flexible WCE methods specifically in this area.

217 This study had several limitations. We only had data available on activity frequency, not

218 intensity or duration. Our exposure definition of the number of activity sessions a child

219 participated in within a given week was parent-reported and very broad, and there was likely

220 large heterogeneity in the intensity, duration, and type of activity. As such, our findings are an

average across many different contexts. Limiting the target population to specific sports and

employing an exposure definition that accounts for intensity and duration would provide moreprecise inferences, at the cost of decreased generalizability.

We assumed that all children have the same weight function regardless of their sex, age, and other characteristics. Although we explored effect modification by sex, future studies may also wish to explore other effect modifiers in the relationship between activity and injury. We also grouped together all types of injuries for our outcome, and different patterns may exist for acute versus overuse injuries. Our study sample included children from a single county within Denmark, and findings may not be generalizable to all populations.

230 We only assessed time to first injury within each schoolyear in our analyses, as subsequent 231 injuries within a short timeframe may be influenced by the initial injury, and the resulting 232 changes in activity patterns. Finally, we employed an approach that did not account for 233 confounding in the relationship between physical activity patterns and injury risk. As such, our 234 results should not be interpreted as causal. Fatigue and soreness are two unmeasured factors that 235 may confound or mediate the relationship between physical activity patterns and injury. WCE 236 methods have been adapted to marginal structural model Cox analyses to handle time-varying 237 confounders or mediators, which could be applied in future research given sufficient data.⁴⁶

238 Conclusion

WCE methods provide insight into the time-specific associations between past physical activity history and injury in children. High levels of activity performed in the recent past is associated with increased injury risk, potentially due to acute tissue damage or fatigue. Activity performed further in the past is associated with decreased injury risk, which may be attributed to improved fitness and strengthened tissues.

244

References

- 1. Warburton DER, Nicol CW, Bredin SSD. Health benefits of physical activity: the evidence. *CMAJ*. 2006;174(6):801-809. doi:10.1503/cmaj.051351
- Janssen I, LeBlanc AG. Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. *Int J Behav Nutr Phys Act.* 2010;7(1):40. doi:10.1186/1479-5868-7-40
- 3. Gabbett TJ. The training—injury prevention paradox: should athletes be training smarter and harder? *Br J Sports Med*. 2016;50(5):273-280. doi:10.1136/bjsports-2015-095788
- Soligard T, Schwellnus M, Alonso JM, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. *Br J Sports Med.* 2016;50(17):1030-1041. doi:10.1136/bjsports-2016-096581
- Bartlett R, Bussey M. Sports Biomechanics: Reducing Injury Risk and Improving Sports Performance. Routledge; 2013. Accessed December 21, 2019. http://ebookcentral.proquest.com/lib/mcgill/detail.action?docID=957394
- 6. Bahr R, Alfredson H, Järvinen M, et al. Types and Causes of Injuries. In: *The IOC Manual of Sports Injuries*. John Wiley & Sons, Ltd; 2012:1-24. doi:10.1002/9781118467947.ch1
- Bestwick-Stevenson T, Toone R, Neupert E, Edwards K, Kluzek S. Assessment of Fatigue and Recovery in Sport: Narrative Review. *Int J Sports Med.* 2022;43(14):1151-1162. doi:10.1055/a-1834-7177
- Wilkins JC, Valovich McLeod TC, Perrin DH, Gansneder BM. Performance on the Balance Error Scoring System Decreases After Fatigue. *J Athl Train*. 2004;39(2):156-161.
- Stevens ST, Lassonde M, de Beaumont L, Paul Keenan J. In-Game Fatigue Influences Concussions in National Hockey League Players. *Res Sports Med.* 2008;16(1):68-74. doi:10.1080/15438620701879020
- Hiemstra LA, Lo IKY, Fowler PJ. Effect of Fatigue on Knee Proprioception: Implications for Dynamic Stabilization. *J Orthop Sports Phys Ther*. 2001;31(10):598-605. doi:10.2519/jospt.2001.31.10.598
- Proske U. Exercise, fatigue and proprioception: a retrospective. *Exp Brain Res*. 2019;237(10):2447-2459. doi:10.1007/s00221-019-05634-8

- Collard DCM, Verhagen EALM, Chin A Paw MJM, van Mechelen W. Acute physical activity and sports injuries in children. *Appl Physiol Nutr Metab*. 2008;33(2):393-401. doi:10.1139/H07-182
- Carter CW, Micheli LJ. Training the child athlete: physical fitness, health and injury. *Br J Sports Med.* 2011;45(11):880-885. doi:10.1136/bjsports-2011-090201
- Wang C, Vargas JT, Stokes T, Steele R, Shrier I. Analyzing Activity and Injury: Lessons Learned from the Acute: Chronic Workload Ratio. *Sports Med.* 2020;50(7):1243-1254. doi:10.1007/s40279-020-01280-1
- Andrade R, Wik EH, Rebelo-Marques A, et al. Is the Acute: Chronic Workload Ratio (ACWR) Associated with Risk of Time-Loss Injury in Professional Team Sports? A Systematic Review of Methodology, Variables and Injury Risk in Practical Situations. *Sports Med.* 2020;50(9):1613-1635. doi:10.1007/s40279-020-01308-6
- Blanch P, Gabbett TJ. Has the athlete trained enough to return to play safely? The acute:chronic workload ratio permits clinicians to quantify a player's risk of subsequent injury. *Br J Sports Med.* 2016;50(8):471-475. doi:10.1136/bjsports-2015-095445
- Stares J, Dawson B, Peeling P, et al. Identifying high risk loading conditions for in-season injury in elite Australian football players. *J Sci Med Sport*. 2018;21(1):46-51. doi:10.1016/j.jsams.2017.05.012
- Sylvestre MP, Abrahamowicz M. Flexible modeling of the cumulative effects of timedependent exposures on the hazard. *Stat Med.* 2009;28(27):3437-3453. doi:10.1002/sim.3701
- 19. Abrahamowicz M. Assessing cumulative effects of medication use: New insights and new challenges. *Pharmacoepidemiol Drug Saf.* 2024;33(1):e5746. doi:10.1002/pds.5746
- Biel R, Danieli C, Shekarrizfard M, et al. Acute cardiovascular health effects in a panel study of personal exposure to traffic-related air pollutants and noise in Toronto, Canada. *Sci Rep.* 2020;10(1):16703. doi:10.1038/s41598-020-73412-6
- Danieli C, Cohen S, Liu A, et al. Flexible Modeling of the Association Between Cumulative Exposure to Low-Dose Ionizing Radiation From Cardiac Procedures and Risk of Cancer in Adults With Congenital Heart Disease. *Am J Epidemiol*. 2019;188(8):1552-1562. doi:10.1093/aje/kwz114

- Niedhammer I, Milner A, Geoffroy-Perez B, Coutrot T, LaMontagne AD, Chastang JF. Psychosocial work exposures of the job strain model and cardiovascular mortality in France: results from the STRESSJEM prospective study. *Scand J Work Environ Health*. 2020;46(5):542-551. doi:10.5271/sjweh.3902
- 23. Abrahamowicz M, Bartlett G, Tamblyn R, du Berger R. Modeling cumulative dose and exposure duration provided insights regarding the associations between benzodiazepines and injuries. *J Clin Epidemiol*. 2006;59(4):393-403. doi:10.1016/j.jclinepi.2005.01.021
- Sylvestre MP, Abrahamowicz M, Čapek R, Tamblyn R. Assessing the cumulative effects of exposure to selected benzodiazepines on the risk of fall-related injuries in the elderly. *Int Psychogeriatr.* 2012;24(4):577-586. doi:10.1017/S1041610211002031
- Wedderkopp N, Jespersen E, Franz C, et al. Study protocol. The Childhood Health, Activity, and Motor Performance School Study Denmark (The CHAMPS-study DK). *BMC Pediatr*. 2012;12(1):128. doi:10.1186/1471-2431-12-128
- 26. Wang C, Stokes T, Steele RJ, Wedderkopp N, Shrier I. Implementing Multiple Imputation for Missing Data in Longitudinal Studies When Models are Not Feasible: An Example Using the Random Hot Deck Approach. *Clin Epidemiol*. 2022;14:1387-1403. doi:10.2147/CLEP.S368303
- Borresen J, Lambert MI. The Quantification of Training Load, the Training Response and the Effect on Performance. *Sports Med.* 2009;39(9):779-795. doi:10.2165/11317780-000000000-00000
- Williams S, West S, Cross MJ, Stokes KA. Better way to determine the acute:chronic workload ratio? *Br J Sports Med.* 2017;51(3):209-210. doi:10.1136/bjsports-2016-096589
- Xiao Y, Abrahamowicz M. Bootstrap-based methods for estimating standard errors in Cox's regression analyses of clustered event times. *Stat Med.* 2010;29(7-8):915-923. doi:10.1002/sim.3807
- 30. Neath AA, Cavanaugh JE. The Bayesian information criterion: background, derivation, and applications. *WIREs Comput Stat.* 2012;4(2):199-203. doi:10.1002/wics.199
- Akaike H. Information Theory and an Extension of the Maximum Likelihood Principle. In: Parzen E, Tanabe K, Kitagawa G, eds. *Selected Papers of Hirotugu Akaike*. Springer Series in Statistics. Springer; 1998:199-213. doi:10.1007/978-1-4612-1694-0_15

- R: The R Project for Statistical Computing. Accessed August 24, 2023. https://www.rproject.org/
- 33. Sylvestre MP, Beauchamp ME, Kyle RP, Abrahamowicz M. WCE: Weighted Cumulative Exposure Models. Published online August 8, 2018. Accessed August 24, 2023. https://cran.r-project.org/web/packages/WCE/index.html
- Brown DM, Picciotto S, Costello S, et al. The Healthy Worker Survivor Effect: Target Parameters and Target Populations. *Curr Environ Health Rep.* 2017;4(3):364-372. doi:10.1007/s40572-017-0156-x
- Jones CM, Griffiths PC, Mellalieu SD. Training Load and Fatigue Marker Associations with Injury and Illness: A Systematic Review of Longitudinal Studies. *Sports Med*. 2017;47(5):943-974. doi:10.1007/s40279-016-0619-5
- 36. Wang C, Stokes T, Vargas JT, Steele R, Wedderkopp N, Shrier I. Predicting Injury Risk Over Changes in Physical Activity in Children Using the Acute:Chronic Workload Ratio. *Am J Epidemiol.* 2022;191(4):665-673. doi:10.1093/aje/kwab280
- 37. Keeratisiroj O, Siritaratiwat W. Prevalence of self-reported musculoskeletal pain symptoms among school-age adolescents: age and sex differences. *Scand J Pain*. 2018;18(2):273-280. doi:10.1515/sjpain-2017-0150
- Picavet HSJ, Gehring U, van Haselen A, et al. A widening gap between boys and girls in musculoskeletal complaints, while growing up from age 11 to age 20 - the PIAMA birth Cohort study. *Eur J Pain*. 2021;25(4):902-912. doi:10.1002/ejp.1719
- Wu M, Fallon R, Heyworth BE. Overuse Injuries in the Pediatric Population. Sports Med Arthrosc Rev. 2016;24(4):150. doi:10.1097/JSA.00000000000129
- Lin CY, Casey E, Herman DC, Katz N, Tenforde AS. Sex Differences in Common Sports Injuries. *PM&R*. 2018;10(10):1073-1082. doi:10.1016/j.pmrj.2018.03.008
- Murray NB, Gabbett TJ, Townshend AD, Blanch P. Calculating acute:chronic workload ratios using exponentially weighted moving averages provides a more sensitive indicator of injury likelihood than rolling averages. *Br J Sports Med.* 2017;51(9):749-754. doi:10.1136/bjsports-2016-097152
- 42. Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Clarsen B, Fagerland MW. Assessing the cumulative effect of long-term training load on the risk of injury in team

sports. *BMJ Open Sport Exerc Med*. 2022;8(2):e001342. doi:10.1136/bmjsem-2022-001342

- Gasparrini A, Armstrong B, Kenward MG. Distributed lag non-linear models. *Stat Med*.
 2010;29(21):2224-2234. doi:10.1002/sim.3940
- 44. Gasparrini A. Modeling exposure–lag–response associations with distributed lag nonlinear models. *Stat Med.* 2014;33(5):881-899. doi:10.1002/sim.5963
- 45. Clarsen B, Bahr R, Myklebust G, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. *Br J Sports Med.* 2020;54(7):390-396. doi:10.1136/bjsports-2019-101337
- 46. Xiao Y, Abrahamowicz M, Moodie EEM, Weber R, Young J. Flexible Marginal Structural Models for Estimating the Cumulative Effect of a Time-Dependent Treatment on the Hazard: Reassessing the Cardiovascular Risks of Didanosine Treatment in the Swiss HIV Cohort Study. J Am Stat Assoc. 2014;109(506):455-464.

Tables

Table 1. Characteristics of study participants from a Danish cohort of schoolchildren,
2008-2014 (N=1,667).

Number of participants (%)	
789 (47%)	
878 (53%)	
648 (39%)	
711 (43%)	
308 (19%)	

Table 2. Injury hazard ratios for different patterns of activity over the previous 20 weeks. Activity levels are defined by number of activity sessions per week: minimally active = 1 session/week, moderately active = 4 sessions/week, highly active = 7 sessions/week.

Activity pattern		Reference	HR (95% CI)
Moderately active in previous 20 weeks		Minimally active in previous 20 weeks	1.15 (0.99, 1.36)
Highly active in previous 20 weeks			1.32 (0.98, 1.86)
Minimally active 11- 20 weeks ago	Moderately active 1-10 weeks ago	Minimally active in	1.27 (1.49, 1.09)
	Highly active 1-10 weeks ago	previous 20 weeks	1.63 (1.18, 2.23)
Moderately active 11- 20 weeks ago	Minimally active 1-10	Minimally active in	0.90 (0.78, 1.00)
Highly active 11-20 weeks ago	– weeks ago	previous 20 weeks	0.81 (0.62, 1.00)
Moderately active 11- 20 weeks ago	Minimally active 1-10 weeks ago	Moderately active in previous 20 weeks	0.78 (0.67, 0.92)

Abbreviations: HR, hazard ratio; CI, confidence interval.

Figures



Figure 1. Weight function for the association between past physical activity and the current hazard of first injury in children. The curve represents the best fitting weight function, averaged over 5 imputed datasets. The shaded area represents the 95% bootstrapped confidence bands. Time elapsed = 0 represents the current week (last 7 days). The values of the points represent relative importance weights estimated for activity performed in specific weeks. Positive weights (red) imply a harmful association while negative weights (blue) imply a protective association.



Figure 2. Weight functions for the separate associations between past physical activity and the current hazard of first injury in children, stratified by sex. The curves represent the best fitting weight functions averaged over 5 imputed datasets. The shaded areas represent the 95% bootstrapped confidence bands. Time elapsed = 0 represents the current week (last 7 days). The values of the points represent relative importance weights estimated for activity performed in specific weeks.

Supplementary Material

Collection of physical activity and pain data in the Childhood Health, Activity, and Motor Performance School Study Denmark

Musculoskeletal pains and physical activity participation were measured weekly by the "Short Messaging Service-Track-Questionnaire" (SMS-Track) version 2.1 (New Agenda Solutions, SMSTrack ApS, Esbjerg). SMS-Track is a web based IT-system developed as a tool for frequent surveillance, complying with Shiffman's principle of Ecological Momentary Assessment.

The questionnaire was automatically sent to the parent's mobile phone once a week asking:

A. "Has [NAME OF CHILD] during the last week had any pain in

- 1. Neck, back or low back
- 2. Shoulder, arm or hand
- 3. Hip, leg or foot
- 4. No my child has not had any pain."

Parents were instructed to type the number in front of the correct answer.

B. "How many times did [NAME OF CHILD] engage in sports during the last week"? Parents were instructed to answer with a relevant number between 0 and 8. The answers 0 to 7 represent the unique number of times engaging in sports, whereas 8 stood for 'more than 7 times'.

The returned answers were automatically recorded and inserted into a database. To improve compliance rate, the responders were contacted by telephone if the answer did not meet the instructions. Furthermore, a reminder was automatically sent, if participants had not responded 72 hours later and, if necessary, 120 hours after receiving the message.

Assessing effect modification by sex

We tested for possible effect modification by sex by conducting separate analyses where the time windows and numbers of knots were allowed to vary by sex. We identified the models with the lowest BIC for each sex, and assessed to what extent these separate sex-specific weight functions improved fit in order to test for possible effect modification by sex. In particular, using a test proposed in an earlier WCE study,¹ we compared the total deviances for the entire dataset, obtained using two different sex-stratified WCE analysis approaches, in which the previously estimated weight functions were used to define pre-specified time-varying WCE exposures. (i) The first 'common model' approach used the same weight function, based on the best fitting main model for all children, for each sex. (ii) In contrast, the 'sex-specific models' approach employed separate weight functions, corresponding to the BIC-optimal WCE model for each sex. For each approach, the total deviance was calculated as the sum of the deviances yielded by the models fit to females and males. Finally, the difference between the total deviances of the (i) common versus (ii) sex-specific models was used as a likelihood ratio test (LRT) statistic for a chi-square test with 3 degrees of freedom (df).¹ Three df's correspond to the three additional spline coefficients used to fit two constrained cubic spline weight functions with one interior knot for the sex-stratified models, compared to the common model with only a single weight function.^{1,2} We also used the aforementioned total deviances to calculate and compare the Akaike Information Criterion (AIC) for each of the two models.

References:

- Danieli C, Cohen S, Liu A, Pilote L, Guo L, Beauchamp ME, et al. Flexible Modeling of the Association Between Cumulative Exposure to Low-Dose Ionizing Radiation From Cardiac Procedures and Risk of Cancer in Adults With Congenital Heart Disease. Am J Epidemiol. 2019 Aug 1;188(8):1552–62.
- Sylvestre MP, Abrahamowicz M. Flexible modeling of the cumulative effects of timedependent exposures on the hazard. Stat Med. 2009;28(27):3437–53.

Table S1. Bayesian Information Criterion for weighted cumulative exposure models with different user-specified time windows and interior knots. WCE models modelled the association between cumulative past activity sessions and current injury risk. BICs were averaged across 5 imputed datasets.

Time window	Number of interior knots	BIC
5	1	41,453.21
	2	41,458.86
	3	41,466.62
10	1	41,371.56
	2	41,372.76
	3	41,374.54
15	1	41,448.67
	2	41,454.06
	3	41,461.61
20	1	41,447.86
	2	41,456.17
	3	41,462.21
25	1	41,449.67
	2	41,455.62
	3	41,463.88

Abbreviations: BIC, Bayesian Information Criterion; WCE, weighted cumulative exposure

Comparison between WCE and EWMA

We compared injury risks using flexible weight function versus using an exponentially weighted moving average (EWMA) function over 20 weeks.²⁸ The EWMA function for a given week *t* is estimated as:

$$EWMA(t) = X_i(t) * \left(\frac{2}{N+1}\right) + \left(\left(1 - \left(\frac{2}{N+1}\right)\right) * EWMA_{t-1}\right),$$

where $X_i(t)$ is the number of activity sessions for participant *i* at week *t*, and N is the time window of interest (20 weeks).²⁸



Figure S1. Exponentially weighted moving average (EWMA) function for the association between previous physical activity and injury. The line represents the EWMA function for activity performed in the previous 20 weeks, conditional on activity in the current week. Time elapsed = 0 represents the current week. The points represent weights for activity performed in specific weeks. Negative weights (blue) imply a protective association.

Chapter 6: Manuscript 3

Preface

In Chapter 2, I discussed limitations with the current state of evidence regarding the relationship between changes in physical activity and injury risk. Specifically, I noted that the IOC consensus statement on load in sport and injury risk indicates an increased injury risk at low ACWRs, which is expected due to bias. Our research group had noted this problem in a previous critical review published in *Sports Medicine* in 2020 (written as part of the requirements for my MSc);¹² however, we did not have concrete strategies to resolve it at the time. In this manuscript, I identify the apparent increase in injury risk at low ACWRs as occurring due to "immortal time bias", a commonly acknowledged issue in other fields of epidemiology. I also discuss strategies for mitigating immortal time bias in the context of load and injury.

This manuscript has submitted to Journal of Clinical Epidemiology.

Immortal time bias: a common problem in observational studies of training load and injury

Chinchin Wang, MSc, PhD(c)^{a,b}; Jay S. Kaufman, PhD^a; Ian Shrier, MD, PhD^b

Affiliations:

^aDepartment of Epidemiology and Biostatistics, McGill University, Montreal, Quebec, Canada

^bCentre for Clinical Epidemiology, Lady Davis Institute of Medical Research, Jewish General Hospital, Montreal, Quebec, Canada

Corresponding Author:

Ian Shrier, MD, PhD

Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, 3755 Côte Ste-Catherine Road, Montreal, Quebec, Canada

Email: ian.shrier@mcgill.ca

Phone: 1-514-229-0114

Abstract

Several observational studies of the relationship between training load and injury have found increased risks of injury at low loads. These associations are expected because load is often assessed at the end of the injury follow-up period. As such, athletes who get injured earlier in the follow-up period will have systematically lower loads than athletes who get injured later in the follow-up period. In this commentary, we identify this problem as a type of "immortal time bias", a methodological issue that has been recognized in other areas of epidemiology. We also discuss how immortal time bias can be prevented by aligning the measurement of load with the start of follow-up for injury.

Keywords: training load, injury, acute:chronic workload ratio, immortal time bias

What is new

- Immortal time bias is common in observational studies of training load and injury risk
- Bias occurs because the measurement of load occurs after the start of follow-up for injuries
- Immortal time bias can be prevented by using planned rather than observed loads as the exposure of interest, following up for injuries after load measurement, or advanced methods such as inverse probability weighting

1 Introduction

The relationship between training load (referred to as "load" for simplicity) and injury has become a popular topic in sports injury epidemiology in recent years. Much research in this area have been observational in nature, taking advantage of existing training surveillance systems or cohort studies. While observational studies have the advantages of larger sample sizes, feasibility, and lower cost compared to randomized trials, they are also more easily prone to biases in their analyses. In this commentary, we introduce the concept of immortal time bias as applied to the relationship between load and injury, and discuss strategies for prevention.

9 Source of immortal time bias

10 The "acute:chronic workload ratio" (ACWR) is a common metric used to summarize relative changes in load, including by the International Olympic Committee in a consensus statement on 11 12 load and injury in sport.¹ We previously identified flaws in the relationship presented in this 13 statement, including that increased injury risks are expected at low ACWRs due to bias.² Briefly, 14 consider Athlete A who planned 2 hours of activity each day for the next week, and Athlete B 15 who planned 1 hour of activity each day. Athlete A gets injured on Day 2, with an observed 16 weekly load of 4 hours. Athlete B gets injured on Day 6, with an observed weekly load of 6 17 hours. Athlete A will have a lower load and ACWR than Athlete B, despite having planned more 18 activity and having performed more activity in Days 1 and 2 (Figure 1).

19 This is a type of "immortal time bias", a methodological issue that has been recognized in other areas of epidemiology.³ "Immortal time" refers to a period of time during which the outcome 20 21 cannot occur conditional on the exposure. Generally, immortal time bias will occur when 22 treatment assignment (load measurement) is not aligned with the start of follow-up [time zero 23 (*t0*)].^{4,5} Studies of load and injury commonly follow-up for injury over a particular period (e.g. 24 one week), and measure load in that same week. Since load is not determined until the end of 25 that week, treatment assignment occurs after the start of follow-up. This problem occurs in both 26 studies looking at changes in load, and absolute load.

27 In our example, load measurement only occurs at the end of follow-up (*t1* in Figure 1). Given

28 Athlete B's planned load of 1 hour per day and their observed load of 6 hours over the week,

29 Athlete B must not have been injured between Days 1 and 6. As such, they were "immortal" for

30 this time. Similarly, Athlete A is "immortal" for Days 1 and 2 given their planned load of 2 hours

31 per day and their observed load of 4 hours over the week. In other words, the timing of their

32 injuries are fixed given the available information on their planned and observed loads.

The extent of the immortal time bias depends on the degree to which load and injury data are summarized. Load is often defined using daily moving averages, with injury incidence measured over a day rather than a week.⁶ While this reduces bias, as the use of smaller time intervals means less possible immortal time, load measurement still occurs after the start of follow-up. Athletes who get injured earlier in the day will have necessarily smaller loads than athletes who complete the day uninjured.² If acute load is defined as the load on the current day, the relative bias may remain considerable.

40 **Preventing immortal time bias in studies of load and injury**

To prevent immortal time bias, we must align treatment assignment with time zero. In other words, we must determine load using only information we have at time zero.^{4,5} One way to do this is to use planned load rather than observed load as the exposure and estimate the intentionto-treat effect of load on injury. In our example, we know that Athlete A planned 14 hours of activity while Athlete B planned 7 hours. By comparing their injury risk using planned loads, we would infer that Athlete A's larger planned load resulted in a greater risk of injury.

47 Alternatively, we can align treatment assignment with time zero by measuring injury risk over a follow-up period that begins at the time that load is measured.² In our example, we might 48 49 compare injury risks for Athletes A and B in the subsequent week. While this avoids immortal 50 time bias, it does not account for variations in load in the subsequent week that affect injury, and may still result in biased inferences.² For instance, an athlete with a low ACWR in the current 51 52 week may be exposed to a large increase in load in the subsequent week that they were not 53 sufficiently prepared for, resulting in injury. This would result in an apparent increased injury 54 risk at low ACWRs, as observed in the International Olympic Committee consensus statement.¹ 55 Alternatively, load can be measured using daily moving averages, with injury incidence 56 measured in the subsequent day to minimize (but not avoid) unaccounted variations in load 57 leading up to the time of injury.

58 Another option is to assign each individual to multiple treatments at time zero, and censor them when their treatment is no longer consistent with their assignment.^{4,5} Censoring must be 59 accounted for with inverse probability weighting to avoid selection bias.^{4,5} For instance, suppose 60 we wanted to investigate whether a weekly load of 4 hours was associated with the same injury 61 62 risk as a weekly load greater than 4 hours, as was seen in our naïve analysis of Athletes A and B. We could clone each individual and assign each clone to a different treatment at time zero (≤ 4 63 64 hours versus > 4 hours of activity). We would then follow-up each clone for injury, and censor 65 the clone at the point that their observed load is no longer consistent with their treatment assignment. The Athlete B clone assigned to ≤ 4 hours of activity would therefore be censored 66 after doing 4 hours of activity, and any injuries occurring after that point would not be included 67 in the analysis (Figure 2). As such, we would conclude that the risk of injury at a load > 4 hours 68 is greater than \leq 4 hours. We note that such an analysis is only feasible when treatments are 69 70 categorical, as assessing injury risk for specific load or ACWR values would result in an infinite 71 number of clones.

72 **Conclusion**

Immortal time bias is a common problem in observational studies of load and injury risk. To
prevent this bias and estimate causal effects, we must design our analyses so that treatment
assignment or load measurement occurs at time zero, or the start of follow-up. This can be done
by using planned loads rather than observed loads as the exposure, or through cloning and
censoring individual observations in the case of categorical exposures.

78

References

- Soligard T, Schwellnus M, Alonso JM, Bahr R, Clarsen B, Dijkstra HP, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. Br J Sports Med. 2016 Sep 1;50(17):1030–41.
- 2. Wang C, Vargas JT, Stokes T, Steele R, Shrier I. Analyzing Activity and Injury: Lessons Learned from the Acute:Chronic Workload Ratio. Sports Med. 2020 Jul 1;50(7):1243–54.
- 3. Suissa S. Immortal Time Bias in Pharmacoepidemiology. Am J Epidemiol. 2008 Feb 15;167(4):492–9.
- Hernán MA, Sauer BC, Hernández-Díaz S, Platt R, Shrier I. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. J Clin Epidemiol. 2016 Nov 1;79:70–5.
- 5. Hernán MA, Sterne JA C, Higgins JPT, Shrier I, Hernández-Díaz S. A structural description of biases that generate immortal time. Epidemiology. 2024;(Submitted).
- Carey DL, Blanch P, Ong KL, Crossley KM, Crow J, Morris ME. Training loads and injury risk in Australian football—differing acute: chronic workload ratios influence match injury risk. Br J Sports Med. 2017 Aug 1;51(16):1215–20.

Author Contributions:

CW conceptualized the manuscript, drafted the initial manuscript, and critically reviewed and revised the manuscript.

JSK critically reviewed and revised the manuscript.

IS critically reviewed and revised the manuscript.

Funding Sources:

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors. CW is funded by a Doctoral Research Award from the Canadian Institutes for Health Research.

Figures



Figure 1. Immortal time bias in the measurement of load. Loads (measured using activity duration) are indicated for each Day 1-7 during the calendar week. Observed loads are indicated in orange for Athlete A, and blue for Athlete B. Planned loads that were not observed due to injury are indicated in grey. Injuries are represented by red X's. Follow-up for injury starts at time zero (t0, beginning of the calendar week) and ends at t1 (end of the calendar week). Load is assessed at t1. Despite having planned a larger load and having been exposed to a larger load up to the point of injury, a smaller load is observed for Athlete A than Athlete B.



Figure 2. Cloning and censoring method to prevent immortal time bias in studies of load and injury. Loads (measured using activity duration) are indicated for each Day 1-7 during the calendar week. Observed loads are indicated for Athlete A, and blue for Athlete B. Injuries are represented by red X's. At time zero (*t0*), each Athlete is cloned and assigned to a different treatment (\leq 4 hours or >4 hours). Clones are censored (indicated in grey) when either (a) they become injured, or (b) their observed exposure is no longer consistent with their assigned treatment. Calculated injury risks are 50% for athletes assigned \leq 4 hours of load, and 100% for athletes assigned >4 hours of load, correctly indicating an increased rather than equal risk for the >4 hour treatment group.

Chapter 7: Manuscript 4

Preface

In Chapter 2, I discussed how the vast majority of studies regarding the relationship between changes in physical activity or load and injury risk are associational; yet their findings have been used to generate recommendations for athletes and the general public. Various authors have emphasized the need to employ causal frameworks to determine the relationship between changes in load and injury risk.^{12,35,93,145,146} However, to our knowledge, no studies have done so up to this point in time. Further, limited guidance exists for researchers looking to conduct causal inference in the area of sport injury epidemiology.

Studies in other fields of epidemiology have employed a "target trial framework" to estimate causal effects in observational data. This framework has been shown to provide results consistent with those from randomized controlled trials. Its application helps to avoid biases in observational analyses, including the immortal time bias described in Manuscript 3. In this manuscript, I discuss the potential application of the target trial framework to studies on changes in physical activity or load on injury risk, and provide guidance to researchers looking to employ this framework for various causal questions.

This manuscript has been published in *BMJ Open Sport & Exercise Medicine* (https://doi.org/10.1136/bmjsem-2024-002037).
The target trial framework for determining the effect of changes in training load on injury risk using observational data: a methodological commentary

Chinchin Wang, MSc, PhD(c)^{a,b}; Jay S. Kaufman, PhD^a; Russell J. Steele, PhD^c; Ian Shrier, MD, PhD^b

Affiliations:

^aDepartment of Epidemiology, Biostatistics and Occupational Health, McGill University, Montreal, Quebec, Canada

^bCentre for Clinical Epidemiology, Lady Davis Institute for Medical Research, Jewish General Hospital, Montreal, Quebec, Canada

^cDepartment of Mathematics and Statistics, McGill University, Montreal, Quebec, Canada

Email Addresses:

Chinchin Wang, chinchin.wang@mail.mcgill.ca

Jay S. Kaufman, jay.kaufman@mcgill.ca

Russell J. Steele, russell.steele@mcgill.ca

Ian Shrier, ian.shrier@mcgill.ca

Corresponding Author:

Ian Shrier, MD, PhD

Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, 3755 Côte Ste-Catherine Road, Montreal, Quebec, Canada

Email: ian.shrier@mcgill.ca; Phone: 1-514-229-0114

Abstract

In recent years, a large focus has been placed on managing training load for injury prevention. To minimize injuries, training recommendations should be based on research that examines causal relationships between load and injury risk. While observational studies can be used to estimate causal effects, conventional methods to study the relationship between load and injury are prone to bias. The target trial framework is a valuable tool that requires researchers to emulate a hypothetical randomized trial using observational data. This framework helps to explicitly define research questions and design studies in a way that estimates causal effects. This article provides an overview of the components of the target trial framework as applied to studies on load and injury, and describes various considerations that should be made in study design and analyses to minimize bias.

Keywords: load monitoring, training load, sport injury, injury prevention, target trial

Summary Box

What is already known on this topic:

- There is large interest among athletes, coaches, and clinicians in managing changes in training load for injury prevention
- Longitudinal data from observational studies or load monitoring programs can provide valuable insights into the causal effect of changing load on injury risk, but data must be analyzed appropriately
- The target trial framework is a tool for designing and analyzing observational studies in a way that emulates a randomized controlled trial and minimizes bias

What this study adds:

- This review discusses considerations for applying the target trial framework to studies examining the causal effects of changes in load on injury risk
- We provide guidance for defining the research question, eligibility criteria, treatment strategies, and outcomes, and for conducting appropriate analyses

How this study might affect research, practice or policy

- The application of the target trial framework in research can be used to generate valid recommendations to minimize injuries
- The insights outlined in this review can aid researchers in designing rigorous observational studies that estimate the causal effects of changing load on injury risk

1 Introduction

2 Avoiding injury is an important goal for athletes of all sports and levels. Training load (also 3 referred to as 'load' or 'workload') is considered an important risk factor for injury.^{1,2} Training 4 load refers generally to a broad range of exposure variables related to sport or physical activity that can be manipulated to elicit a physiological response.^{3–5} For simplicity, the term 'load' will 5 be used to refer to this concept henceforth. It is generally accepted that larger absolute loads are 6 associated with higher injury risks.^{1,2} Mechanistically, this may occur through increased 7 8 mechanical stress on tissues, increased fatigue affecting decision-making, coordination, and/or 9 neuromuscular control,⁶ and increased exposure time at risk.⁷

10 In recent years, a large focus has been placed on the relationship between changes in load and injury. Gabbett et al. proposed an "acute-chronic workload ratio" (ACWR) model to relate 11 changes in load to injury based on Banister et al.'s fitness and fatigue performance model.^{6,8,9} In 12 13 this model, athletes with similar acute loads (causing fatigue) and chronic loads (proxy for 14 fitness) are thought to be performing activity at a level that they are well-prepared for, 15 minimizing injury risk, whereas athletes with high acute loads and low chronic loads are generally exceeding what they are prepared for, increasing injury risk.^{6,9} Athletes with low acute 16 loads and high chronic loads are also thought to be at increased injury risk.^{6,10} Although no 17 18 biological explanations were initially provided, it was later suggested that one's past (chronic) load may promote physical adaptations (e.g. tissue strengthening) that protect against injury.^{6,10} 19 However, one's recent (acute) load may cause fatigue and decrease tissue strength and 20 mechanical stress capacity, increasing risk of injury.⁶ No biological explanations have been 21 22 provided for the increased injury risk associated with low acute loads and high chronic loads 23 (excluding a decrease in technical skill following rest periods for sports requiring high precision such as gymnastics)¹¹, and this finding is likely due to methodological flaws.^{12,13} 24

The monitoring of load to inform training decisions with the goal of reducing injury is now done across a variety of sport types and levels.¹⁴ Training recommendations largely depend on existing models resulting from observational studies.^{6,14} While randomized controlled trials (RCTs) are considered the gold standard for identifying causal relationships and evidence-based decision making, they are not often feasible. RCTs generally require a large sample size and long follow30 up, which is often impractical, especially in elite settings.¹⁵ As such, researchers often rely on

- 31 observational data. However, none of the observational studies reported in existing systematic
- 32 reviews have explicitly estimated a causal effect of changes in load on injury risk.^{1,2,16–18} Further,
- 33 conventional methods used to study this relationship are prone to bias, and are unlikely to
- 34 correspond to true causal effects.¹²
- 35 Observational data can be used to estimate causal effects only if certain assumptions hold.
- 36 Meaningful differences have been observed between results from observational studies with
- 37 traditional study designs and those from RCTs, leading to concerns about their validity.
- 38 However, some authors have shown that if the observational study design and analysis emulates
- 39 a hypothetical randomized trial (called a "target trial"),¹⁹ the results are generally consistent with
- 40 those from RCTs, $^{20-24}$ although this is not always the case. 25,26 We propose that this framework
- 41 be applied to studies of load and injury risk to generate higher quality evidence regarding their
- 42 relationship.

The objective of this review is to describe the components of the target trial framework as
applied to studies of load and injury risk, including potential biases and other challenges as well
as strategies to address them.

46 Target trial framework components

The target trial framework requires researchers to define their research question and study
protocol in a way that mimics a hypothetical RCT, and conduct their study analyses using
observational data in a way that emulates that protocol.¹⁹ This process minimizes errors and
resulting biases that are common in observational analyses.

- 51 The major components of a target trial protocol are: 1) eligibility criteria (population); 2)
- 52 treatment strategies (intervention and comparison); 3) assignment procedures; 4) outcome; 5)
- 53 follow-up period; 6) causal contrasts of interest; and 7) analysis plan. These components have
- 54 been described in further detail elsewhere.^{20,27} In this section, we outline these components and
- 55 discuss specific considerations for studies of changes in load and injury risk.

56 Eligibility criteria

57 In an RCT, we would start by identifying our population of interest, and specifying inclusion and

58 exclusion criteria to determine eligible individuals. The same criteria should be used for an

59 observational study. Eligibility should be determined at "time zero", or the start of follow-up,

60 and only using baseline information prior to the follow-up period.²⁸ If there are missing data on

61 important baseline variables, results may not be meaningful given the potential for bias.

62 **Defining the population of interest**

In both RCTs and target trial emulations, the population should be defined by who we are 63 64 interested in intervening on. This may be a specific athletic population (e.g. elite soccer players) 65 or a general population (e.g. youth). When studying general populations, we note that an 66 intervention of a "change in load" is likely to have different effects in different participants 67 (effect heterogeneity). For instance, the same increase in load is expected to affect inactive 68 individuals and regularly active individuals differently. This can promote generalizability; 69 however, if we are interested in a specific subset of the population (e.g. regularly active 70 individuals) it may be appropriate to restrict our study population to those with certain baseline 71 levels of activity measured over a run-in period, with participants only eligible for analyses 72 following this period. Otherwise, we may explore heterogeneity using stratification or an 73 interaction term between baseline activity and the intervention. Any subgroup analyses of 74 primary interest (i.e. not exploratory or hypothesis generating) should be considered in the 75 sample size calculation.

We must also consider how our outcome of injury informs our population of interest. Previous injury is considered an important risk factor for new injuries.²⁹ In an RCT, we might restrict to healthy individuals (e.g. those who are not currently injured or recovering from injury). We would include the same restrictions in an observational study. Data from a participant who is eligible at baseline is included until injury. Once recovered, data from the same participant would only be included once they are again eligible for the study, after several (e.g. four or five) weeks without injury.

83 Selection bias affecting internal validity

84 Individuals should not be included or excluded from analyses based on information gathered during follow-up. The selection of individuals based on factors that result from their intervention 85 86 and outcome may cause bias through several mechanisms, affecting the internal validity of findings.³⁰ One example might be analyses that are restricted to those who attended a certain 87 88 number of training sessions over the follow-up period. Participants who experience health 89 problems (e.g. illness, pain, mental health conditions) are less likely to participate in training,³¹ 90 and health problems may be a consequence of changes in load and injury. Excluding participants 91 based on training participation during follow-up may therefore create bias in both RCTs and 92 observational studies. Rather, alternative methods exist that address adherence/non-adherence to planned activity within RCTs (see discussion of per-protocol effects below).³² The same 93 94 principles should be applied to observational studies.

95 **Dropouts and censoring**

96 Dropouts affect both RCTs and observational studies. Individuals who drop out or are lost to follow up are considered censored, as their outcome (and potentially their intervention) is not 97 observed.³³ Excluding censored individuals from analyses will result in selection bias when the 98 99 reason for drop out/loss to follow up are associated with the intervention and outcome.³³ For 100 instance, individuals who are less accustomed to activity and experience higher levels of 101 discomfort or soreness from small increases in activity may be less motivated to remain in a study. Instead, censoring can be accounted for by imputing missing data,³⁴ or using inverse 102 probability weighting, assuming that data are available on the covariates associated with drop 103 out.33 104

105 **Treatment strategies**

106 Most analyses of RCTs and observational studies compare two treatment strategies: an

107 intervention, and a comparison or control. In our context, the intervention is a change in load.

108 Load has been operationalized in numerous ways,³ and over various time frames.¹⁶ The optimal

109 measure of load depends on the research context and available data. However, the same

110 principles apply for defining treatment strategies regardless of the load metric.

111 **Defining changes in load**

The target trial framework prompts researchers to define treatment strategies that are relevant to stakeholders (e.g., athletes, coaches, policymakers) within the specific sporting context. When defining a "change in load", we must consider the baseline load, whether change is expressed as an absolute vs. relative amount, and whether change is measured at a single time point or as a continuous intervention.

117 Measurements of change require a baseline or reference value. A simple measure of change in 118 load might be a weekly change, or the change in load during the follow-up week (beginning at 119 time zero) relative to the previous week. In this case, the baseline load would be the absolute 120 load in the previous week. Other options might be an unweighted average load over multiple 121 weeks (akin to chronic load within the ACWR framework), a weighted average, or a cumulative 122 measure. When deciding on a baseline load, researchers should consider any theories underlying 123 the relevance of previous loads in affecting current injury risk, as well as utility for athletes, 124 clinicians, and other stakeholders. For instance, whereas large increases in load may increase 125 susceptibility to injury, these increases are common after recovery or taper weeks which are thought to reduce injury risk.³⁵ 126

127 We must also consider whether to express change as an absolute amount (e.g. an hour more of 128 training this week) or a relative amount (e.g. 10% increase in running distance this week). We 129 will distinguish between individual and policy-level interventions to illustrate this decision. 130 Individuals are typically interested in how their injury risk may differ under different behaviours 131 or patterns to inform their training decisions. For instance, a runner might ask questions like 132 "What is my injury risk if I increase my total distance covered by 5km this week?" (absolute 133 change) or "What is my injury risk if I increase my total distance covered by 10% this week?" 134 (relative change). The impact of changes in load on injury risk on an individual is expected to differ by their baseline fitness.³⁶ For instance, a 5km increase in running distance is likely to 135 136 result in a much greater injury risk for someone who regularly runs 5km per week versus 50km 137 per week. Similarly, a 10% increase in distance may also result in differing injury risks for these 138 two individuals, but perhaps not to the same extent as the absolute change. Policymakers are 139 interested in improving the health of an entire population. Policies are generally on an absolute

scale, such as one where children within a school are mandated to take at least one physical
education class,³⁷ or where youth community rugby players are allowed a maximum of 90
minutes of playing time per day.³⁸

143 Furthermore, we must decide whether we are interested in change at a single time point or as a 144 continuous intervention. While a soccer team might be interested in increasing their practices by 145 one hour in a single week (single time point), individuals training for a marathon might be 146 interested in gradually increasing their running distance relative to their previous distance over 147 several weeks (continuous intervention). Within continuous interventions, changes in load are 148 not limited to an increase or a decrease. An intervention to decrease injury risk might incorporate 149 maintenance weeks and recovery weeks (e.g. taper) where load is unchanged or decreased. These 150 weeks are not easily studied under single time point interventions, particularly when baseline 151 load is measured as an average over several weeks such as in the ACWR framework.¹² Under a 152 continuous framework, we might compare: (1) a tapering program with a 10% increase in 153 activity for 3 weeks followed by a 20km decrease in activity for 1 week prior to competition, 154 versus (2) a 10% increase in activity for 4 weeks prior to competition. Note that a continuous 155 intervention can incorporate both absolute and relative changes in load.

156 **Defining the comparison strategy**

157 The comparison of two treatment strategies should reflect meaningful real-world decisions, such 158 as a reasonable alternative behaviour/pattern/policy, or one that is currently in place. For 159 instance, a suitable comparison for a runner interested in increasing their total distance by 20% 160 each week might be an increase in total distance by 10% each week, until a maximal distance is 161 reached, while a suitable comparison for a soccer team wanting to include an extra hour of 162 training moving forwards might be maintaining their current training schedule. A comparison for 163 a policy mandating at least one physical education class per week might be to not have this 164 mandate in place, allowing the population to participate in physical education as they choose.

To determine causal effects, ideally all aspects of training would be maintained between the
treatment strategy and comparator except for the aspect that is being intervened on. For instance,
if we were interested in increasing training volume (e.g. distance run), we would want to keep

intensity (e.g. pace) constant. This may not be feasible using observational data, and we may
instead be limited to assessing the impact of increasing training volume on injury risk regardless
of intensity. This is a limitation of using observational data compared to RCTs. At the same time,
it is a strength of the target trial emulation approach because it makes these challenges more
transparent compared to traditional observational approaches.

173 Thus far, we have only considered comparisons between 2 treatment strategies, with specific yet 174 arbitrary values for changes in load. In practice, researchers may choose to dichotomize or 175 categorize changes in load when defining their treatment strategies (e.g. increase distance by 5-176 10km versus increase distance by 0-4 km). These categorizations should be done in a way that reflects realistic training practices, rather than arbitrarily. Determining the effect of a continuous 177 178 range of changes in load on injury risk is analogous to determining a dose-response curve. The 179 development of a dose-response curve requires a single RCT with many arms, or multiple RCTs. 180 This remains true with the target trial emulation approach, and therefore requires defining 181 multiple comparison strategies and a more complex analytical strategy (covered in more detail in 182 "Analysis plan").

183 **Consistency and positivity**

Positivity and consistency are two conditions necessary for causal inference (along with exchangeability, covered in the following section).³⁹ Under positivity, each individual should theoretically have a positive probability of receiving each level of exposure for every combination of covariates.^{39,40} As such, each individual should be theoretically capable of changing their load by a specified amount, which may not be the case for large relative increases in load (e.g. tripling training time in a day when someone is currently training for eight hours per day). The treatment strategies should be realistic given the eligibility criteria for a study.

Briefly, consistency requires that treatments be defined unambiguously so that there cannot be two versions of a single treatment that would result in the same individual having different outcomes.^{39–41} In our context, this involves specifically defining what a change in load represents, including the type of activity, frequency, intensity, and/or duration. Further, there cannot be interference, where an individual's outcome depends on another individual's 196 treatment. In our context, one individual's load should not affect another individual's injury risk.

197 Consistency is likely to be violated when there is a broad intervention, such as an increase in

198 activity duration that does not account for intensity over a variety of sports. While this can be

199 avoided by having more specific research questions, in reality, stakeholders may be interested in

200 general recommendations. Researchers should aim to strike a balance between defining clear

201 treatment strategies and generalizability.

202 Assignment procedures

203 **Controlling for baseline confounders**

204 In an RCT, treatments are assigned at random at baseline. This achieves exchangeability, one of 205 the necessary conditions for causal inference, given a large enough sample size and perfect adherence to the assigned treatment strategy.³⁹ Simply, exchangeability means that there is no 206 207 inherent difference in the risk of injury between treatment and control groups, and that any 208 observed differences are due to the treatment itself. Under full exchangeability, the outcomes for 209 the intervention group are the same as the outcomes for the control group had the control group 210 received the intervention, and the outcomes for the control group are the same as the outcomes 211 for the intervention group had the intervention group not received intervention, all else being 212 equal.42

213 Training decisions are rarely random in observational data. An individual's magnitude of change

in load may be influenced by factors such as sex, age, experience, baseline activity levels,

215 planned strength and conditioning training, recent recovery or taper weeks, or previous injuries.

216 These factors may also influence injury risk, and therefore act as confounders. As full

217 exchangeability requires that there be no unmeasured confounding,⁴² confounders must be

adjusted for in observational analyses through methods such as inverse probability of treatment

219 weighting,⁴³ multivariable regression, or both (doubly robust estimation).⁴⁴

220 For a treatment strategy that occurs at a single time point (e.g. increase in load in a single week),

adjustment must only be done for factors measured at baseline. Adjustment for factors measured

during follow-up affected by the treatment or outcome (e.g. illness) may result in bias^{30,45} and

223 decrease precision.⁴⁵ For treatment strategies that occur over a period of time (e.g. consistently

increasing load by 10% each week), there may be time-varying confounders that affect injury

- risk and subsequent changes in loads. One example is fatigue or soreness causing one to decrease
- their load. Time-varying confounding must be handled using specialized methods developed by Rehins and collectures 3^{39}
- 227 Robins and colleagues.³⁹

228 Timing of treatment assignment and immortal time bias

229 Treatment assignment, or the observational analogue of defining an individual's exposure, must 230 be done at baseline to properly emulate a target trial. However, observational studies of changes 231 in load and injury often only measure acute load at the end of the follow-up period. As such, any 232 injury occurring during follow-up can affect one's measured load and cause a bias akin to immortal time bias in other fields of epidemiology.^{28,46} For instance, load may be measured as 233 234 one's activity performed over a week. Athletes who get injured earlier in the calendar week will 235 not be able to perform their planned activity for the rest of the week, and will have systematically lower loads than athletes who complete the week without injury.¹² The same principles apply 236 when daily averages are used to calculate loads, but with reduced bias.¹² 237

238 Researchers sometimes impose an injury lag period, in which only injuries occurring in a 239 specified time window (e.g. one week) subsequent to the load window will be attributed to that load.¹⁶ In this setting, treatment assignment would occur at the beginning of the follow-up 240 241 period, defined as the week following the load window. This eliminates the bias explained 242 above, but ignores the principle that current load is the inciting factor for injury and assumes that the load between the end of the load window and the time of injury is not relevant.¹² 243 244 Alternatively, researchers may use planned loads rather than observed loads to calculate changes 245 in load, and estimate an intention-to-treat (ITT) effect of changes in load on activity. ITT effects 246 are discussed further under "Causal contrasts of interest".

247 **Outcome**

A well-designed study requires a clear definition of the outcome. Injury can be defined in many

249 ways. Common categorizations include any athlete-reported complaint, medical attention

250 injuries, and/or time-loss injuries.⁴⁷ The onset of injury might be defined at the time of first

251 complaint, initiation of time lost from sport, or at the time of medical diagnosis.

252 Multiple injuries

Injuries can and often do occur more than once in the same individual, and one's risk of subsequent injury may be affected by previous injuries.²⁹ Furthermore, injuries often influence one's subsequent activity patterns. As such, previous or current injuries are a confounder for the relationship between changes in load and injury, and must be accounted for in study design or analyses.

258 Previous or current injuries at the start of follow-up can be adjusted for as baseline confounders 259 in observational studies. These might be included as dichotomous variables (e.g. yes/no injury in 260 the previous X months), or continuous variables (e.g. number of injuries in the previous X 261 months). However, most RCTs would only include healthy individuals as part of their eligibility 262 criteria, excluding those who have returned to training but are not fully healed. We might 263 emulate this criterion by only including individuals in our study up to their initial injury, after 264 which they are no longer eligible. However, this would greatly reduce our effective sample size. 265 Alternatively, we may believe that one's injury risk is unaffected by previous injuries after a 266 certain time period (e.g. one month). Similar to an RCT that might restrict to individuals who 267 have not been injured in the past month, we can restrict our observational analyses to those who 268 have been uninjured for one month prior to the start of follow-up. This is equivalent to a 269 "washout" period commonly employed in pharmacoepidemiology studies, where participants are observed for a period of time prior to follow-up to ensure that outcomes are not due to exposures 270 that occurred prior to the study.^{48,49} However, if we are interested in a sustained intervention 271 272 such as an increase in load over several time points, we must treat injuries occurring during 273 follow-up as time-varying confounders, and account for them using the appropriate methods.¹²

Finally, we may explore effect heterogeneity between initial and subsequent injuries through stratified analyses or by assessing interactions if relevant to our research question.

276 Causal contrasts of interest

277 Data from RCTs can be used to obtain an intention-to-treat (ITT) or per-protocol (PP) effect

278 estimate.⁵⁰ Analogs of these effects can be estimated using observational data.²⁰

279 ITT versus PP effects

280 The ITT estimate addresses the question "What is the effect of assigning a policy or intervention 281 on injury?". Participants are analyzed in the group that they were assigned during randomization, 282 irrespective of the treatment they actually received, non-adherence, or drop-out. This maintains 283 exchangeability between groups assuming no drop-outs, but will generally result in conservative 284 effect estimates for the treatment actually received due to noncompliance.^{50,51} The ITT estimate 285 may be of interest on a policy level because not everyone is expected to comply to policies or recommendations in real life.³² For instance, coaches or clinicians may be interested in ITT 286 287 estimates because they prescribe training plans rather than follow them.

288 The PP estimate addresses the question "What is the effect of a policy or intervention on injury if 289 everyone adhered to the policy or intervention?".³⁹ Traditional methods to estimate the PP effect 290 include "as-treated" analyses which compare participants based on the treatment they actually 291 took, or "naïve per protocol"/"on-treatment" analyses that are restricted to participants who followed their assigned treatment.⁵² These analyses are essentially observational, as individuals 292 293 are able to choose their intervention. To properly estimate the PP effect, more sophisticated 294 analyses with additional assumptions are required to adjust for confounding and non-adherence to avoid bias, even in an RCT setting.^{39,52} For example, although the objective of a recent RCT 295 296 was to estimate the ITT effect of providing a load management software program on injury risk, the conclusion referred to "managing training loads" (a PP effect).⁵³ Such a conclusion would 297 require more assumptions, different analyses, and higher quality data. The PP estimate is 298 299 generally of greater interest to individuals for informing decisions (e.g. athletes trying to 300 minimize injury risk).³⁹

The ITT and PP estimates will differ when there is non-adherence to treatment assignment. Nonadherence to a training plan or pattern may occur due to reasons such as injuries at baseline, fatigue, soreness, illness, or motivation. Importantly, individuals who get injured during followup and stop training should be considered as having adhered to their treatment assignment so long as they were following their strategy up to the point of injury, as we would not expect injured participants to continue their regular training.

307 Estimating ITT effects using observational data

308 To estimate ITT effects using observational data, we must determine an individual's treatment 309 assignment using their planned loads at baseline, and adjust for baseline confounders related to 310 their planned training. This is only feasible if planned loads such as a weekly training program 311 are available. We recommend that planned training schedules be collected in observational 312 studies to allow ITT analyses to be conducted, and to avoid immortal time bias as discussed 313 above. Within team sports, participants generally have the same training schedule and planned 314 loads. However, their baseline loads may differ due to absences, non-adherence, etc. The planned 315 "changes in load" should be based on each individual's observed baseline load. Further, because 316 participants on the same team may have similar training schedules and propensities for injury, 317 clustering by team should be accounted for in analyses.

318 Estimating PP effects in observational data

To estimate PP effects using observational data, we must compare individuals based on their actual activity patterns, as opposed to their planned training.

321 Above, we discussed how immortal time bias can occur if acute load is measured at the end of 322 the follow-up period. This creates difficulties in estimating PP effects, as we are unable to obtain 323 an unbiased measure of an individual's observed exposure or training. To estimate PP effects for 324 a specific change in load at a single timepoint, we must impose an injury lag period and follow-325 up for injuries *after* the acute load is measured. For instance, we could define the outcome as 326 injuries occurring within a day after the current week, and assign treatments based on an 327 individual's change in load for that week compared to the previous week. In a nested target trial 328 approach, the follow-up period for injury would be on the following Monday for the trial where 329 load was measured from Monday to Sunday, the following Tuesday for the trial where load was 330 measured from Tuesday to Monday, and so forth. However, this approach would ignore 331 variations in load in the current day that might affect injury risk. Despite the advantages of target 332 trial emulation, it does not solve the challenge in estimating PP effects for specific single 333 timepoint interventions which may be of interest for athletes, coaches, and clinicians.

334 PP effects can be estimated using a cloning and censoring approach when the treatment is categorical.²⁰ For instance, we might be interested in whether an increase in distance by 5-10km 335 336 increases injury risk compared to an increase in distance by 0-4 km. Under this approach, we 337 would clone each individual in our analyses, assigning each clone to a different treatment at time 338 zero (5-10km vs. 0-4km). We would follow-up each clone for injury, and censor the clone at the 339 point that their observed load is no longer consistent with their treatment assignment. Such an 340 analysis is only feasible for dichotomous treatments or treatments with few categories, as 341 assessing injury risk for a continuous range of changes in load would result in an infinite number 342 of clones.

343 For sustained interventions such as a consistent increase in load over several weeks, we must 344 adjust for time-varying confounders related to non-adherence and injury using methods such as inverse probability weighting or g-estimation.⁵² Important confounders include fatigue and 345 soreness, and this information should be collected in load and injury surveillance or studies to 346 347 determine causal effects. Alternatively, if training schedules are available, planned training can 348 be used as an instrumental variable to estimate the effect of changing load on injury in the 349 presence of unmeasured confounding, providing the underlying assumptions are likely to hold true.12,52 350

351 Analysis plan

352 Generally, the study analysis requires creating a statistical model that reflects the relationship 353 between the exposure and outcome, and estimating the effect of interest.⁴⁵

354 **Pooling multiple trials**

In RCTs, eligible participants are typically identified and randomized into one of two treatment groups at baseline or "time zero". In observational data, an individual may meet eligibility criteria at multiple time points. To increase the number of observations and effective sample size, we might allow individuals to contribute multiple trials or follow-up periods, given they meet eligibility requirements.²⁵ This is analogous to a repeated measures design in an RCT, where individuals participate in a trial multiple times.^{39,55} In both a repeated measures RCT and observational study, we would have to account for repeated measures in the analyses (e.g.
 through cluster bootstrapping,⁵⁶ mixed models,^{57,58} or generalized estimating equations⁵⁹).

363 Estimating effects using observed versus predicted data

The majority of studies employing the target trial framework assign individuals to a treatment group consistent with their observed data. For instance, if we were interested in the per-protocol effect on injury risk for an increase in load by 2-fold or more versus less than 2-fold, we would categorize each individual into a group based on their observed increase in load assessed at time zero. If an individual's observed exposure was compatible with multiple treatments at time zero, we could employ a cloning and censoring approach to minimize bias.²⁰

370 Treatment assignment using observed data becomes inefficient for treatments that are continuous 371 variables. For instance, we may be interested in comparing injury risk for an increase in load by 372 2-fold compared to 1-fold. Any individual with an increase in load by a value other than 2-fold 373 or 1-fold would be excluded from analyses, drastically reducing the sample size. Instead, we can 374 employ marginal standardization.^{60,61} Briefly, we create a model reflecting the relationship 375 between continuous increases in load and injury risk (appropriately accounting for confounding, 376 loss to follow-up, etc.), and predict each individual's outcomes under different hypothetical 377 treatments. In this scenario, we could include all eligible individuals in our predictive model, and 378 predict whether or not they would become injured under either treatment (2-fold increase vs. 1-379 fold increase). We can then use these results to estimate the average treatment effect across the different treatments, with bootstrapping to calculate standard errors and confidence intervals.^{60,61} 380

381 Conclusion

To inform training recommendations and prevent injuries among athletes, we require evidence on the relationships between changes in load and injury. While observational data is often used in studying the relationship between changes in load and injury risk, conventional analytic approaches are prone to bias. The target trial framework is a valuable and simple tool to explicitly define causal questions and design studies to estimate causal effects using observational data. By applying this framework, we can strengthen the validity of future research in the sport medicine field. Although the target trial framework solves some of the challenges

- 389 compared to current approaches, other challenges remain including isolating the effects of a
- 390 single aspect of load, implementing intention-to-treat or instrumental variable analyses when
- 391 planned loads are not available, and limitations in estimating per-protocol effects.

Competing Interests: The authors declare no competing interests.

Contributorship: C.W. conceptualized and drafted the initial manuscript. J.S.K., R.J.S., and I.S. critically reviewed and revised the manuscript. All authors provided final approval for submission.

Acknowledgements: N/A

Funding: N/A

Ethics Approval: N/A

Data Sharing Statement: Not applicable.

References

- Drew MK, Finch CF. The Relationship Between Training Load and Injury, Illness and Soreness: A Systematic and Literature Review. *Sports Med.* 2016;46(6):861-883. doi:10.1007/s40279-015-0459-8
- Eckard TG, Padua DA, Hearn DW, Pexa BS, Frank BS. The Relationship Between Training Load and Injury in Athletes: A Systematic Review. *Sports Med*. 2018;48(8):1929-1961. doi:10.1007/s40279-018-0951-z
- Impellizzeri FM, Marcora SM, Coutts AJ. Internal and External Training Load: 15 Years On. Int J Sports Physiol Perform. 2019;14(2):270-273. doi:10.1123/ijspp.2018-0935
- Lambert MI, Borresen J. Measuring Training Load in Sports. Int J Sports Physiol Perform. 2010;5(3):406-411. doi:10.1123/ijspp.5.3.406
- 5. Coutts AJ, Crowcroft S, Kempton T. Developing athlete monitoring systems: Theoretical basis and practical applications. In: *Sport, Recovery, and Performance*. Routledge; 2017.
- Soligard T, Schwellnus M, Alonso JM, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. *Br J Sports Med.* 2016;50(17):1030-1041. doi:10.1136/bjsports-2016-096581
- Kalkhoven JT, Watsford ML, Coutts AJ, Edwards WB, Impellizzeri FM. Training Load and Injury: Causal Pathways and Future Directions. *Sports Med.* Published online January 5, 2021. doi:10.1007/s40279-020-01413-6
- Calvert TW, Banister EW, Savage MV, Bach T. A Systems Model of the Effects of Training on Physical Performance. *IEEE Trans Syst Man Cybern*. 1976;SMC-6(2):94-102. doi:10.1109/TSMC.1976.5409179
- Hulin BT, Gabbett TJ, Blanch P, Chapman P, Bailey D, Orchard JW. Spikes in acute workload are associated with increased injury risk in elite cricket fast bowlers. *Br J Sports Med.* 2014;48(8):708-712. doi:10.1136/bjsports-2013-092524

- 10. Gabbett TJ. The training—injury prevention paradox: should athletes be training smarter and harder? *Br J Sports Med.* 2016;50(5):273-280. doi:10.1136/bjsports-2015-095788
- Orlando C, Levitan EB, Mittleman MA, Steele RJ, Shrier I. The effect of rest days on injury rates. *Scand J Med Sci Sports*. 2011;21(6):e64-e71. doi:10.1111/j.1600-0838.2010.01152.x
- Wang C, Vargas JT, Stokes T, Steele R, Shrier I. Analyzing Activity and Injury: Lessons Learned from the Acute: Chronic Workload Ratio. *Sports Med.* 2020;50(7):1243-1254. doi:10.1007/s40279-020-01280-1
- Impellizzeri FM, Tenan MS, Kempton T, Novak A, Coutts AJ. Acute:Chronic Workload Ratio: Conceptual Issues and Fundamental Pitfalls. *Int J Sports Physiol Perform*. 2020;15(6):907-913. doi:10.1123/ijspp.2019-0864
- Bourdon PC, Cardinale M, Murray A, et al. Monitoring Athlete Training Loads: Consensus Statement. *Int J Sports Physiol Perform*. 2017;12(Suppl 2):S2-161-S2-170. doi:10.1123/IJSPP.2017-0208
- Wang C, Stokes T, Vargas JT, Steele R, Wedderkopp N, Shrier I. Predicting Injury Risk Over Changes in Physical Activity in Children Using the Acute:Chronic Workload Ratio. *Am J Epidemiol*. 2022;191(4):665-673. doi:10.1093/aje/kwab280
- Andrade R, Wik EH, Rebelo-Marques A, et al. Is the Acute: Chronic Workload Ratio (ACWR) Associated with Risk of Time-Loss Injury in Professional Team Sports? A Systematic Review of Methodology, Variables and Injury Risk in Practical Situations. Sports Med. 2020;50(9):1613-1635. doi:10.1007/s40279-020-01308-6
- Jones CM, Griffiths PC, Mellalieu SD. Training Load and Fatigue Marker Associations with Injury and Illness: A Systematic Review of Longitudinal Studies. *Sports Med*. 2017;47(5):943-974. doi:10.1007/s40279-016-0619-5
- Griffin A, Kenny IC, Comyns TM, Lyons M. The Association Between the Acute:Chronic Workload Ratio and Injury and its Application in Team Sports: A Systematic Review. Sports Med Auckl NZ. 2020;50(3):561-580. doi:10.1007/s40279-019-01218-2

- Hernán MA, Wang W, Leaf DE. Target Trial Emulation: A Framework for Causal Inference From Observational Data. *JAMA*. 2022;328(24):2446-2447. doi:10.1001/jama.2022.21383
- Hernán MA, Robins JM. Using Big Data to Emulate a Target Trial When a Randomized Trial Is Not Available. *Am J Epidemiol*. 2016;183(8):758-764. doi:10.1093/aje/kwv254
- Kutcher SA, Brophy JM, Banack HR, Kaufman JS, Samuel M. Emulating a Randomised Controlled Trial With Observational Data: An Introduction to the Target Trial Framework. *Can J Cardiol.* 2021;37(9):1365-1377. doi:10.1016/j.cjca.2021.05.012
- Boyne DJ, Cheung WY, Hilsden RJ, et al. Association of a Shortened Duration of Adjuvant Chemotherapy With Overall Survival Among Individuals With Stage III Colon Cancer. JAMA Netw Open. 2021;4(3):e213587. doi:10.1001/jamanetworkopen.2021.3587
- Cain LE, Saag MS, Petersen M, et al. Using observational data to emulate a randomized trial of dynamic treatment-switching strategies: an application to antiretroviral therapy. *Int J Epidemiol*. 2016;45(6):2038-2049. doi:10.1093/ije/dyv295
- Wang SV, Schneeweiss S, RCT-DUPLICATE Initiative. Emulation of Randomized Clinical Trials With Nonrandomized Database Analyses: Results of 32 Clinical Trials. JAMA. 2023;329(16):1376-1385. doi:10.1001/jama.2023.4221
- 25. Danaei G, García Rodríguez LA, Cantero OF, Logan RW, Hernán MA. Electronic medical records can be used to emulate target trials of sustained treatment strategies. J *Clin Epidemiol.* 2018;96:12-22. doi:10.1016/j.jclinepi.2017.11.021
- Pearce N, Vandenbroucke JP. Are Target Trial Emulations the Gold Standard for Observational Studies? *Epidemiology*. 2023;34(5):614. doi:10.1097/EDE.00000000001636
- Matthews AA, Danaei G, Islam N, Kurth T. Target trial emulation: applying principles of randomised trials to observational studies. *BMJ*. 2022;378:e071108. doi:10.1136/bmj-2022-071108

- Hernán MA, Sauer BC, Hernández-Díaz S, Platt R, Shrier I. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. J *Clin Epidemiol.* 2016;79:70-75. doi:10.1016/j.jclinepi.2016.04.014
- Fulton J, Wright K, Kelly M, et al. Injury risk is altered by previous injury: a systematic review of the literature and presentation of causative neuromuscular factors. *Int J Sports Phys Ther.* 2014;9(5):583-595.
- 30. Cole SR, Platt RW, Schisterman EF, et al. Illustrating bias due to conditioning on a collider. *Int J Epidemiol*. 2010;39(2):417-420. doi:10.1093/ije/dyp334
- Clarsen B, Bahr R, Myklebust G, et al. Improved reporting of overuse injuries and health problems in sport: an update of the Oslo Sport Trauma Research Center questionnaires. *Br J Sports Med.* 2020;54(7):390-396. doi:10.1136/bjsports-2019-101337
- Shrier I, Steele RJ, Verhagen E, Herbert R, Riddell CA, Kaufman JS. Beyond intention to treat: What is the right question? *Clin Trials*. 2014;11(1):28-37. doi:10.1177/1740774513504151
- Boston 677 Huntington Avenue, Ma 02115 +1495-1000. Causal Inference: What If (the book). Miguel Hernan's Faculty Website. Published October 19, 2012. Accessed March 6, 2023. https://www.hsph.harvard.edu/miguel-hernan/causal-inference-book/
- Altman DG. Missing outcomes in randomized trials: addressing the dilemma. *Open Med*. 2009;3(2):e51-e53.
- 35. Mujika I, Padilla S. Scientific Bases for Precompetition Tapering Strategies. *Med Sci* Sports Exerc. 2003;35(7):1182. doi:10.1249/01.MSS.0000074448.73931.11
- Mohseni MM, Filmalter SE, Taylor WC, Vadeboncoeur TF, Thomas CS. Factors Associated With Half- and Full-Marathon Race-Related Injuries: A 3-Year Review. *Clin J* Sport Med. 2021;31(5):e277. doi:10.1097/JSM.00000000000775
- Ministry of Education. The Ontario Curriculum, Grades 9-12: Health and Physical Education, 2015. Publications Ontario. Accessed May 17, 2024.

https://www.publications.gov.on.ca/browse-catalogues/health-physical-educationcurriculum/secondary/the-ontario-curriculum-grades-9-12-health-and-physical-education-2015

- 2023 Age Grade Law Variations Introduced Across Canada. Rugby Canada. Published April 6, 2023. Accessed May 17, 2024. https://rugby.ca/en/news/2023/04/2023-age-gradelaw-variations-introduced-across-canada
- 39. Hernán M, Robins J. *Causal Inference: What If.* Chapman & Hall/CRC; 2020. Accessed October 6, 2021. https://www.hsph.harvard.edu/miguel-hernan/causal-inference-book/
- 40. Cole SR, Frangakis CE. The Consistency Statement in Causal Inference: A Definition or an Assumption? *Epidemiology*. 2009;20(1):3. doi:10.1097/EDE.0b013e31818ef366
- Rehkopf DH, Glymour MM, Osypuk TL. The Consistency Assumption for Causal Inference in Social Epidemiology: When a Rose is Not a Rose. *Curr Epidemiol Rep.* 2016;3(1):63-71. doi:10.1007/s40471-016-0069-5
- Sarvet AL, Wanis KN, Stensrud MJ, Hernán MA. A graphical description of partial exchangeability. *Epidemiol Camb Mass*. 2020;31(3):365-368. doi:10.1097/EDE.00000000001165
- Chesnaye NC, Stel VS, Tripepi G, et al. An introduction to inverse probability of treatment weighting in observational research. *Clin Kidney J.* 2022;15(1):14-20. doi:10.1093/ckj/sfab158
- Funk MJ, Westreich D, Wiesen C, Stürmer T, Brookhart MA, Davidian M. Doubly Robust Estimation of Causal Effects. *Am J Epidemiol*. 2011;173(7):761-767. doi:10.1093/aje/kwq439
- Schisterman EF, Cole SR, Platt RW. Overadjustment Bias and Unnecessary Adjustment in Epidemiologic Studies. *Epidemiol Camb Mass*. 2009;20(4):488-495. doi:10.1097/EDE.0b013e3181a819a1

- 46. Suissa S. Immortal Time Bias in Pharmacoepidemiology. *Am J Epidemiol*.
 2008;167(4):492-499. doi:10.1093/aje/kwm324
- 47. Bahr R, Clarsen B, Derman W, et al. International Olympic Committee Consensus Statement: Methods for Recording and Reporting of Epidemiological Data on Injury and Illness in Sports 2020 (Including the STROBE Extension for Sports Injury and Illness Surveillance (STROBE-SIIS)). *Orthop J Sports Med.* 2020;8(2):2325967120902908. doi:10.1177/2325967120902908
- Rothman KJ, Greenland S, Lash TL. *Modern Epidemiology*. Lippincott Williams & Wilkins; 2008.
- Harvey RD, Mileham KF, Bhatnagar V, et al. Modernizing Clinical Trial Eligibility Criteria: Recommendations of the ASCO-Friends of Cancer Research Washout Period and Concomitant Medication Work Group. *Clin Cancer Res Off J Am Assoc Cancer Res*. 2021;27(9):2400-2407. doi:10.1158/1078-0432.CCR-20-3855
- Ranganathan P, Pramesh CS, Aggarwal R. Common pitfalls in statistical analysis: Intention-to-treat versus per-protocol analysis. *Perspect Clin Res*. 2016;7(3):144. doi:10.4103/2229-3485.184823
- Gupta SK. Intention-to-treat concept: A review. *Perspect Clin Res*. 2011;2(3):109-112. doi:10.4103/2229-3485.83221
- 52. Hernán MA, Hernández-Díaz S. Beyond the intention to treat in comparative effectiveness research. *Clin Trials Lond Engl.* 2012;9(1):48. doi:10.1177/1740774511420743
- 53. Dalen-Lorentsen T, Bjørneboe J, Clarsen B, Vagle M, Fagerland MW, Andersen TE. Does load management using the acute:chronic workload ratio prevent health problems? A cluster randomised trial of 482 elite youth footballers of both sexes. *Br J Sports Med.* Published online October 7, 2020. doi:10.1136/bjsports-2020-103003
- Petersen ML, van der Laan MJ. Causal Models and Learning from Data: Integrating Causal Modeling and Statistical Estimation. *Epidemiology*. 2014;25(3):418. doi:10.1097/EDE.000000000000078

- Singh V, Rana RK, Singhal R. Analysis of repeated measurement data in the clinical trials. J Ayurveda Integr Med. 2013;4(2):77-81. doi:10.4103/0975-9476.113872
- 56. Huang FL. Using Cluster Bootstrapping to Analyze Nested Data With a Few Clusters. *Educ Psychol Meas.* 2018;78(2):297-318. doi:10.1177/0013164416678980
- Lindstrom MJ, Bates DM. Nonlinear Mixed Effects Models for Repeated Measures Data. Biometrics. 1990;46(3):673-687. doi:10.2307/2532087
- 58. Wallace D, Green SB. Analysis of Repeated Measures Designs with Linear Mixed Models. In: *Modeling Intraindividual Variability With Repeated Measures Data*. Psychology Press; 2001.
- Ballinger GA. Using Generalized Estimating Equations for Longitudinal Data Analysis. Organ Res Methods. 2004;7(2):127-150. doi:10.1177/1094428104263672
- Muller CJ, MacLehose RF. Estimating predicted probabilities from logistic regression: different methods correspond to different target populations. *Int J Epidemiol*. 2014;43(3):962-970. doi:10.1093/ije/dyu029
- Naimi AI, Whitcomb BW. Estimating Risk Ratios and Risk Differences Using Regression. *Am J Epidemiol*. 2020;189(6):508-510. doi:10.1093/aje/kwaa044

Figures



Figure 1. Immortal time bias in the measurement of load. Loads (measured as duration) are indicated for each Day 1-7 during the calendar week. Observed loads are indicated in orange for Athlete A, and blue for Athlete B. Planned loads that were not observed due to injury are indicated in grey. Injury is represented by a red X. Follow-up for injury starts at time zero (t0, beginning of the calendar week) and ends at t1 (end of the calendar week). Load is assessed at t1. Despite having planned a larger load and having been exposed to a larger load up to the point of injury, a smaller load is observed for Athlete A than Athlete B who completed the week without injury. This creates a bias known as "immortal time bias" in epidemiology.^{28,46}

Chapter 8: Manuscript 5

Preface

In Manuscript 3, I noted that immortal time bias can be avoided in studies of changes in load and injury risk by measuring the "intention-to-treat effect", or the effect of a planned (rather than observed) change in load on injury risk. In Manuscript 4, I discussed the application of the target trial framework to observational studies of changes in physical activity or load and injury risk. In this manuscript, I employ this framework to determine the intention-to-treat effect of changes in participation in practice and games on injury risk among adolescent ice hockey players. It is the first study to our knowledge to employ the target trial framework in sport injury epidemiology.

This manuscript has undergone one round of revisions in *Journal of Science and Medicine in Sports*. Conference abstracts based on contents of this manuscript were accepted as an oral presentation at the Canadian Academy of Sports and Exercise Medicine Annual Symposium (Niagara Falls, May 2024), and as a poster presentation at the Society for Epidemiology Annual Meeting (Austin, June 2024).

The effect of changes in planned participation on injury risk in adolescent ice hockey players: a target trial emulation

Authors: Chinchin Wang, MSc, PhD(c)^{1,2}; Paul Eliason, PhD³; Jean-Michel Galarneau, PhD³; Carolyn A. Emery, PT, PhD³; Sabrina Yusuf, MSc(c)^{1,4}; Russell J. Steele, PhD⁵; PhD; Jay S. Kaufman, PhD²; Ian Shrier, MD, PhD^{1,4}

Affiliations: ¹Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, McGill University, Montreal, Quebec, Canada; ²Department of Epidemiology and Biostatistics, McGill University, Quebec, Canada; ³Sport Injury Prevention Research Centre, Faculty of Kinesiology, University of Calgary, Calgary, Canada; ⁴Department of Family Medicine, McGill University, Montreal, Quebec, Canada; ⁵Department of Mathematics and Statistics, McGill University, Quebec, Canada;

Corresponding Author:

Ian Shrier, MD, PhD

Centre for Clinical Epidemiology, Lady Davis Institute, Jewish General Hospital, 3755 Côte Ste-Catherine Road, Montreal, Quebec, Canada

Email: ian.shrier@mcgill.ca

Phone: 1-514-229-0114

Abstract

Objective: Few studies have estimated causal relationships between training load and injury risk. Target trial emulation is a framework for conducting causal inference using observational data. We employ this framework to estimate the effect of changing planned participation duration, measured using the acute:chronic workload ratio (ACWR), on injury risk among adolescent ice hockey players without recent injuries.

Design: Prospective cohort study designed to emulate a hypothetical randomized trial.

Methods: We used data from a 5-year cohort study (2013-2018) of ice hockey players aged 13 to 17 years in Alberta and British Columbia. We estimated injury risks associated with different planned changes in hockey participation duration (e.g. half [ACWR=0.5], no change [ACWR=1], two-fold [ACWR=2], three-fold [ACWR=3], and five-fold [ACWR=5]) relative to participation in the previous 4 weeks. Outcomes were modelled using generalized additive models. We conducted secondary analyses restricted to concussions, and stratified by league bodychecking status.

Results: There were 2,633 eligible participants, contributing 115,821 player-trials. Injury risk was 1.9% (95%CI: 1.7%-2.3%) for no change in participation (ACWR=1). Injury risk ratios (RRs) were 0.43 at ACWR=0.5 (95%CI: 0.31-0.54), 1.62 (95% CI: 1.33-1.98) at ACWR=2, 1.91 at ACWR=3 (95%CI: 1.52-2.48) and 2.35 at ACWR=5 (95%CI: 1.68-3.26). Patterns were similar by league bodychecking status. Concussion RRs were stable between ACWR=1 to 1.5, but RRs were greater than for any injury past ACWR=2.

Conclusion: Within the assumptions of this target trial emulation, injury risk increases consistently (no sweet spots) for increases in planned changes in participation duration relative to the previous 4 weeks among adolescent ice hockey players without recent injuries. Increases in injury risk are less than expected for the increased exposure time at risk, suggesting beneficial effects of increasing participation that partially counteract the increased exposure time.

Keywords: injury, concussion, adolescent, male, female, humans, hockey, workload, cohort studies, intention-to-treat analysis

Introduction

Ice hockey is a popular sport among adolescents in Canada.¹ However, it is also a high-risk sport for injuries, particularly concussions.^{1,2} It is important to minimize injury risk while promoting the benefits of ice hockey participation.

Participation in sports such as ice hockey exposes individuals to forces, known collectively as "load".³ Injuries occur when tissues are exposed to greater loads than their load capacity.^{3,4} Injuries may occur due to sudden large loads, such as falls or collisions.^{2,5} These are acute injuries,⁶ and include concussions, sprains/strains, dislocations, bruises, cuts, and fractures.² Injuries may also occur due to repeated exposure to loads without sufficient recovery, resulting in decreased load capacity and eventual significant damage.^{7,8} These are gradual onset injuries,⁶ and include tendinopathies, apophyseal injuries, and stress fractures, among others.^{9–11}

Increases in sport participation may increase injury risk through multiple pathways: 1) increased exposure time at risk,¹² 2) increased physical and/or mental fatigue from increased loads affecting balance, proprioception, spatial awareness, and reaction time in the case of acute injuries,^{13–18} and 3) increased loads without sufficient recovery time affecting load capacity in the case of overuse injuries.³

The acute:chronic workload ratio (ACWR) is a popular metric for quantifying relative changes in load.^{19–21} The ACWR is calculated as the acute (i.e. current) load divided by the chronic (i.e. previous) load. As tissue loads are infeasible to measure, particularly on a population-level, various proxy measures are used to quantify load that may differ depending on the sport and research context.^{20–22} Some previous studies have found higher injury risks at high ACWR values,^{20,21,23} consistent with the principle that loads beyond what an individual is prepared for cause injury, whether through increased time at risk, fatigue, or insufficient recovery. However, other studies have found lower injury risks at high ACWRs.^{24,25} Despite being used to make recommendations about sport participation,²⁶ most ACWR-based studies have been associational, with methodological limitations that are likely to create biases which could explain these contradictory findings.^{27,28} Further, some studies have suggested that the associations between week-to-week changes (an absolute measure) and injury risk are similar to those

between the ACWR and injury risk.²⁹ Despite these limitations, we use the ACWR in this study as a measure of relative changes in load based on its popularity in current literature.

Target trial emulation is a framework for conducting causal inference using observational data.^{30,31} It involves defining the research question and study protocol to mimic a hypothetical randomized controlled trial (RCT) (the "target trial"), and emulating the protocol using observational data.^{30,32} This process helps avoid common biases arising in observational analyses.³³ Target trial emulation has yielded results comparable to RCTs,^{34,35} and is recommended by researchers as a "best practice" for conducting observational studies.^{32,36–38} Despite being increasingly used in epidemiologic research, it has not been commonly applied to sport injury epidemiology. We refer readers to a methodological article for more information on this framework and its potential application to this research area.³¹

RCTs can be analyzed to obtain an "intention-to-treat" or "per-protocol" effect.³⁹ The intentionto-treat effect compares participants based on the intervention they were assigned, whereas the per-protocol effect compares participants based on the intervention they actually received.³⁹ Whereas the intention-to-treat effect is unbiased in an RCT assuming successful randomization, the per-protocol effect can be biased if reasons for non-adherence are not appropriately controlled for. Intention-to-treat effects can be estimated under the target trial framework by using planned loads and adjusting for baseline confounders to emulate randomization.³¹

The objective of this study is to conduct target trial emulation to estimate the intention-to-treat effect of relative changes in planned participation in practices and games on injury and concussion risk among adolescent ice hockey players without recent injuries. Additionally, we examine differences in effects between bodychecking and non-bodychecking leagues.

Methods

Data Source

The data source for this study was Safe2Play, a 5-year (2013 to 2018) longitudinal prospective cohort study of adolescent ice hockey players in Alberta and British Columbia, Canada. Safe2Play included male and female players on co-ed teams across all levels of play in Under-13 (ages 11 to 12), Under-15 (ages 13 to 14), and Under-18 (ages 15 to 17) age-groups, in leagues

that allowed or disallowed bodychecking. All hockey associations within their respective regions were invited to participate, and individual teams were recruited if they could identify a team designate (e.g. manager, coach, parent) to report participation in games and practices on a weekly exposure sheet (WES). Injury report forms (IRF) were initiated by study personnel based on self-report, team designate report on WES, or team therapist report. Individual players (≥14 years for mature minor consent) providing written consent and parents (children <14 years) providing written consent (child providing written assent) were included in the study. Participants could enter the study at any time in the hockey season (October to April), but most were recruited at the beginning of the season. Participants could be followed for multiple seasons.

During each season, weekly exposure data were recorded by the team designate, including practice and game durations in minutes, participation for each player (full, partial, or none), and reasons for missed participation (hockey-related injury, non-hockey-related injury, sickness, or other). Hockey-related injuries resulting in medical attention, inability to complete the session, or missed participation from subsequent sessions (i.e. time-loss) were recorded on an IRF by the team designate. The Safe2Play study has resulted in numerous secondary analyses and publications; more details about its procedures can be found elsewhere.^{40–44}

Target Trial Specification and Emulation

We specified the protocol of hypothetical target trials to estimate the intention-to-treat effect of changing participation by different amounts on injury risk and emulated this protocol using data from the Safe2Play study. Components of the target trials are summarized in Table 1.

Eligibility Criteria

The target trials would include adolescent ice hockey players who were participating in hockey practices and games and had not been injured or recovering from injury in the 4 weeks leading up to the study.

Our observational study included participants from the Safe2Play parent cohort study followed in the 2013-2014 through 2016-2017 playing seasons. The 2017-2018 season was excluded due to

significant missing exposure data. Participants were excluded if they had missing data on whether they played in a league that allowed or disallowed bodychecking. Those in the Under-13 age group were excluded as bodychecking was banned for this group in Canada in 2013, and including this group would violate the positivity assumption required for causal inference.⁴⁵

Individuals must have been participating in hockey practices and games for the previous 4 weeks and must not have been injured or recovering from injury in the previous 4 weeks. As such, individuals must have had data for at least 5 weeks within a season to be eligible (4 weeks for eligibility criteria and 1 week for follow-up). Within the Safe2Play data, individuals may have been eligible at multiple time points. To maximize sample size and power, individuals at each eligible time point were considered as separate units of analysis (player-trials),³³ with adjustment for clustering in the analyses. An individual could have been eligible the day after the previous player-trial, and therefore have contributed multiple overlapping player-trials (Supplementary Material, Figure S1).³¹

Intervention

Because we were interested in intention-to-treat effects, interventions were expressed as X-fold changes (ranging from 0.1- to 5-fold) in planned participation for the upcoming week, measured using the ACWR (acute load divided by chronic load). The acute load was calculated as the daily average planned participation duration over 7 days, starting at time zero. Planned participation duration was defined as the number of minutes of hockey practices and games in a team's schedule, as reported by the team designate, and did not account for individuals' absences from practices or games. As would occur in an RCT accounting for previous activity, the chronic load was calculated as the daily average participation duration over the previous 28 days, ending at time zero (Supplementary Material, Figure S1). The chronic load was measured using observed rather than planned participation, accounting for absences from practices or games. Individuals who were completely absent from a practice or game were assigned a session duration of 0, whereas individuals who partially participated in a practice or game were assigned 50% of the session duration.

Outcome

The outcome was any ice hockey-related injury occurring during participation in practices or games resulting in medical attention, inability to complete the session, or missed participation from subsequent sessions (i.e. time-loss). In a secondary analysis, we limited the outcome to concussions.

Intervention Assignment

In our target trials, participants would be randomly assigned to an X-fold change in participation duration (ACWR value) ranging from 0.1 (10-fold decrease in activity) to 5.0 (5-fold increase in activity). Their assigned participation duration (planned acute load) would be equal to their chronic load multiplied by their assigned X-fold change or ACWR.

In our observational study, the observed X-fold changes in participation duration were calculated for each individual at each eligible time point. To emulate exchangeability obtained by randomization in an RCT, we identified baseline confounders based on background knowledge of factors that may be common causes of changes in planned participation and injury risk. We identified chronic load, age group (Under-15/Under-17), and league bodychecking status (allows bodychecking/disallows bodychecking) as potential confounders. A causal directed acyclic graph displaying our assumptions about the causal relationships between our variables of interest can be found in Supplementary Material, Figure S2. We assumed that chronic load affects the change in planned participation, and might differentially affect current injury risk (e.g. by impacting fitness). We also assumed that teams with older and more experienced players as well as those in bodychecking leagues might have more variation in session durations, and that these factors may also impact injury risk. We did not identify sex as a confounder because the large majority of participants were male, and sex was unlikely to affect planned participation as teams were co-ed. We acknowledge that there may be other important factors that we did not identify or have data on, and that other researchers may have different assumptions regarding the causal relationships between these factors.

We then assigned all individuals at each eligible time zero to the same fold change in participation duration. This was done for a range of ACWRs from 0.1 to 5.0 in 0.1-unit increments.
Follow-Up Period

In the target trials, participants would be followed up from time zero for one week to determine whether they became injured. In our observational study, we determined whether or not a participant became injured over the next 7-day window starting at time zero for each player-trial (5-week unit of analysis including 4-week eligibility period [chronic load] and 1-week follow-up [acute load]).

Causal Contrasts of Interest

Our causal contrasts of interest were the intention-to-treat effects of changing participation duration on injury risk. In the target trial, this would be the effect of being assigned to a particular X-fold change in participation duration compared to no change (ACWR=1). In our observational study, this was the effect of planned changes in participation duration compared to no change, based on a team's practice and game schedule, adjusted for baseline confounders.

Subgroup Analyses

We pre-specified subgroup analyses by league bodychecking status as ice hockey players in leagues that allow bodychecking have higher injury risks.^{41,46}

Statistical Analysis

We modelled the relationship between X-fold changes in participation duration and injury using pooled data across eligible person-trials. We fit a generalized additive model (GAM) with injury as the outcome and ACWR as the exposure, with a logit link function for the binomial outcome. We applied a smoothing term to the ACWR using a cubic regression spline with 8 degrees of freedom. We included chronic load, age group, and bodychecking status as covariates.

We then predicted injury risks for each player-trial using our fitted model under interventions where the ACWR was set to values ranging from 0.1 to 5.0, in increments of 0.1. We calculated marginal injury risks and risk ratios (RRs) relative to no change in participation duration (ACWR=1). We conducted separate analyses stratified by league bodychecking status, and with concussion as the outcome. We applied cluster bootstrapping with 300 replicates to account for

repeated measures within players when estimating 95% confidence intervals. Analyses were conducted in RStudio.⁴⁷ GAMs were fit using the *mgcv* package.⁴⁸

Results

A total of 2,633 players were eligible for target trial emulation, contributing 115,821 player-trials over 3,034 player-seasons (Figure 1). Males accounted for 2,406 (91.4%) participants. Most eligible player-trials were contributed by the Under-15 age group, and in leagues that allowed bodychecking (Table 2). Among eligible player-trials, the median planned participation duration was 5 hours/week or 41 minutes/day (IQR: 30 to 54 minutes/day), while the median observed participation duration was 4.5 hours/week or 39 minutes/day (IQR: 26 to 51 minutes/day). The mean difference between planned and observed participation duration was 4 minutes/day (95% CI: -12 to 19 minutes/day). Injuries occurred in 2,588 (2.2%) of eligible player-trials, of which 507 (19.6%) were concussions.

The estimated injury risk for a 1.0-fold (no) change in participation duration was 1.9% (95% CI: 1.7%-2.3%). The largest increases in injury risk occurred with increases in participation duration up to 2.0-fold (RR = 1.62, 95% CI: 1.33-1.98). Injury risk increased further with increases in participation duration up to 5.0-fold (RR = 2.35, 95% CI: 1.68-3.26). Injury risk decreased largely for decreases in participation up to 0.5-fold (RR = 0.43, 95% CI: 0.31-0.54), but was stable for further decreases in participation, with large uncertainty and wide CI's (Figure 2; Supplementary Material, Figure S3 shows 95% CIs for RRs).

Injury risks were higher for leagues allowing bodychecking compared to leagues disallowing bodychecking for increases in participation up to 2-fold and decreases down to 0.8-fold (Figure 3). RRs relative to no change in participation did not differ meaningfully by bodychecking status, except between 0.6-fold and 1-fold, where decreases in participation resulted in lower injury RRs among leagues allowing bodychecking (Supplementary Material, Figure S4).

Concussion risk did not change for increases in participation from 1.0- to 1.5-fold; however, it increased for further increases in participation (Figure 4). Concussion RRs associated with increases in participation beyond 2-fold were greater than observed for all injuries.

Discussion

We found that injury risk increased consistently for planned increases in participation duration relative to the previous 4 weeks, while injury risk decreased for planned decreases in participation duration relative to the previous 4 weeks. Increases in injury risk were less than would be expected for the increase in exposure time at risk, suggesting beneficial effects of increasing participation that partially counteract the increased exposure time at risk. For example, 2-fold, 3-fold, and 5-fold increases in planned participation resulted in 1.5-fold, 2-fold, and 2.5-fold increases in injury risk, respectively. Increases in planned participation relative to the previous 4 weeks resulted in larger increases in concussion risk than any injury, but still less than would be expected for the increased exposure time (1.6-fold, 2.4-fold, and 3.2-fold increase in concussion risk for 2-, 3-, and 5-fold increases in participation, respectively). While injury risks were higher among leagues allowing bodychecking, intention-to-treat effects of changing participation duration on injury risk did not differ substantially by league bodychecking status.

The observed relationship between planned changes in participation and injury risk is likely due to a mixing of mechanisms. Increasing participation results in increased exposure time at risk.¹² It can also result in increased load on tissues without sufficient recovery, resulting in overuse injuries,³ and increased physical and mental fatigue affecting balance, proprioception, spatial awareness, and reaction time, resulting in acute injuries.^{13–18} Increased participation can also have beneficial effects for injury prevention, including increased tissue load capacity given sufficient recovery,³ improved skills and confidence,^{49,50} and improved fitness.^{51,52} Separating these different mechanisms is complex, and not necessarily of interest to coaches, players, and parents whose goal is to increase participation while minimizing risk of all injuries.

Some previous studies looking at associations between changes in load and injury risk identified "sweet spots" where load can be increased without increasing injury risk.^{21,53} For instance, studies have identified ACWRs between 0.8-1.3 to be associated with decreased injury risk compared to ACWRs below or above this range.⁵³ Our results using target trial emulation suggest no such "sweet spot" for changing participation while minimizing injury risk in adolescent ice hockey, although concussion risk was stable for increases in participation between

1.0- to 1.5-fold. Participation decisions in this setting should balance the benefits of increased participation with the consequences of increased injury risk.

Strengths

Our study is the first to our knowledge to apply target trial emulation to assess the relationship between changes in load and injury risk. This framework requires defining a study protocol in a way that mimics a hypothetical RCT, helping avoid common errors in biases in conventional observational analyses.³⁰

For instance, researchers have often categorized load into weekly blocks by calendar time and calculated the ACWR exposure within these blocks.^{54–56} This creates an issue where the exposure assignment is not aligned with the start of follow-up and measurement of baseline confounders (time zero), resulting in bias where increased injury risks are observed for decreases in load.^{27,33} In our study, we aligned exposure assignment with the start of follow-up by using planned participation durations rather than observed participation to calculate the acute load.

Few studies have considered whether individuals were recently injured in analyses of changes in load and injury risk.²⁵ Instead, analyses often pool initial and subsequent injuries,⁵⁷ which assumes that they have similar relationships with changes in load. However, initial injuries are an important confounder, affecting both subsequent activity and injury risk.⁵⁸ In RCTs, this confounding is avoided by 1) restricting eligibility to uninjured participants at baseline, or 2) randomization of intervention assignment so that groups have similar distributions of injured versus uninjured participants. We avoided confounding by recent injuries in our target trial emulation by restricting participation to players who were uninjured in the previous 4 weeks. Although adjusting for recent injuries at baseline (emulating randomization) would also avoid confounding, we chose not to because there may be important differences in the effects of increasing planned participation on initial versus subsequent injury risk that would be masked by regression adjustment.

We employed a large data source, allowing the use of flexible modelling strategies. The current study and previous studies have suggested that the relationship between changes in load and

injury risk does not follow a simple exponential or polynomial curve.^{59–61} Future research should also employ non-linear modelling approaches.

Limitations

Our study had several limitations. We did not have data on potentially important confounders of fatigue or soreness,⁶² which may have biased our findings. However, we assume that fatigue or soreness would only affect changes in planned participation by affecting the chronic load (as planned participation is based on the team schedule); therefore, by adjusting for chronic load, any bias from ignoring fatigue or soreness for our intention-to-treat estimate is likely to be minimal. We also did not adjust for division skill level. Participants in higher level divisions may have larger increases in planned participation during the season (e.g. due to more tournament play) and may be at higher risk of injury due to increased intensity and speed of play.^{2,63} Although bodychecking generally occurred in higher level divisions, and was included as a confounder, we would still expect some residual confounding that may have resulted in an apparently larger association between changes in planned participation and injury risk.

We operationalized changes in participation duration using the ACWR, a measure that has been criticized for assessing causal effects.^{27,28,62} Although we use an "uncoupled" measure (excluding acute load from calculations of chronic load) that avoids more serious limitations of the ACWR,^{27,64} some authors have suggested that even the uncoupled measure is prone to erroneous inferences.²⁸ We used a ratio as our exposure variable, rather than assessing the acute and chronic loads as separate variables.⁶⁵ This was to be consistent with our research question which concerned the effect of changes in planned participation on injury risk rather than the separate effects of current planned participation and previous participation. We combined acute and overuse injuries in our outcome of interest, assuming that changes in planned participation would impact both acute and overuse injury risk. Our findings are likely due to a mixing of mechanisms as relationships may differ between injury subtypes, and our results only apply under our assumptions. Finally, we did not differentiate between practices and games. Studies have shown that injury incidence is higher during games than practices in team sports,⁶⁶ and the relationship between changes in load and injury risk may differ between practices and games.⁶⁷

Future Directions

Our research question focused on the intention-to-treat effects of changing participation duration at one point in time on injury risk. This aligns with the current application of the ACWR to quantify and reduce injury risk.²⁶ Future directions include determining the effects of participation duration on injury risk at different times throughout the season,⁶⁸ and estimating per-protocol effects based on observed rather than planned participation.⁶⁹ More advanced causal inference methods are required (1) to assess injury risk at multiple timepoints, which requires adjusting for time-varying confounding,⁶⁹ and (2) to estimate per-protocol effects, which requires adjusting for factors related to non-adherence to participation schedules.⁶⁹

Conclusion

Using target trial emulation, we found that injury risk increased consistently with planned changes in duration of participation in practices and games compared to the previous 4 weeks among adolescent ice hockey players without recent injuries, but by less than would be expected for the increase in exposure time at risk. Target trial emulation is easily extended to other study populations and should be considered by sport injury researchers as a valuable tool to explicitly define their study protocols and avoid common errors in observational analyses.

Practical Implications

- Injury risk increased consistently with planned increases in participation among adolescent ice hockey players without recent injuries, but by less than would be expected for the increase in exposure time at risk
- This study did not find a "sweet spot" for increasing ice hockey participation without increasing injury risk
- Participation decisions should balance the benefits of increased participation with the consequences of increased injury risk

References

- Black AM, Meeuwisse DW, Eliason PH, Hagel BE, Emery CA. Sport participation and injury rates in high school students: A Canadian survey of 2029 adolescents. J Safety Res. 2021 Sep 1;78:314–21.
- Emery CA, Meeuwisse WH. Injury Rates, Risk Factors, and Mechanisms of Injury in Minor Hockey. Am J Sports Med. 2006 Dec 1;34(12):1960–9.
- Bartlett R, Bussey M. Sports Biomechanics: Reducing Injury Risk and Improving Sports Performance [Internet]. Florence, UNITED STATES: Routledge; 2013 [cited 2019 Dec 21]. Available from: http://ebookcentral.proquest.com/lib/mcgill/detail.action?docID=957394
- McBain K, Shrier I, Shultz R, Meeuwisse WH, Klügl M, Garza D, et al. Prevention of sports injury I: a systematic review of applied biomechanics and physiology outcomes research. Br J Sports Med. 2012 Mar 1;46(3):169–73.
- Mosenthal W, Kim M, Holzshu R, Hanypsiak B, Athiviraham A. Common Ice Hockey Injuries and Treatment: A Current Concepts Review. Curr Sports Med Rep. 2017 Oct;16(5):357.
- Bahr R, Clarsen B, Derman W, Dvorak J, Emery CA, Finch CF, et al. International Olympic Committee consensus statement: methods for recording and reporting of epidemiological data on injury and illness in sport 2020 (including STROBE Extension for Sport Injury and Illness Surveillance (STROBE-SIIS)). Br J Sports Med [Internet]. 2020 Feb 18 [cited 2020 Mar 16]; Available from: https://bjsm.bmj.com/content/early/2020/02/28/bjsports-2019-101969
- Wiesinger HP, Kösters A, Müller E, Seynnes OR. Effects of Increased Loading on In Vivo Tendon Properties: A Systematic Review. Med Sci Sports Exerc. 2015 Sep;47(9):1885– 95.
- 8. Bahr R, Alfredson H, Järvinen M, Järvinen T, Khan K, Kjær M, et al. Types and Causes of Injuries. In: The IOC Manual of Sports Injuries [Internet]. John Wiley & Sons, Ltd; 2012

[cited 2023 Feb 15]. p. 1–24. Available from: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118467947.ch1

- Daly PJ, Sim FH, Simonet WT. Ice Hockey Injuries. Sports Med. 1990 Aug 1;10(2):122– 31.
- Lorentzon R, Wedrèn H, Pietilä T. Incidence, nature, and causes of ice hockey injuries: A three-year prospective study of a Swedish elite ice hockey team. Am J Sports Med. 1988 Jul 1;16(4):392–6.
- Husen M, Burgsmüller L, Burggraf M, Jäger M, Dudda M, Kauther M. Injuries and Overuse Syndromes in Rink Hockey Players. Int J Sports Med. 2021 Feb;42(02):132–7.
- 12. Stovitz SD, Shrier I. Injury rates in team sport events: tackling challenges in assessing exposure time. Br J Sports Med. 2012 Nov 1;46(14):960–3.
- Wilkins JC, Valovich McLeod TC, Perrin DH, Gansneder BM. Performance on the Balance Error Scoring System Decreases After Fatigue. J Athl Train. 2004;39(2):156–61.
- Stevens ST, Lassonde M, de Beaumont L, Paul Keenan J. In-Game Fatigue Influences Concussions in National Hockey League Players. Res Sports Med. 2008 Feb 7;16(1):68– 74.
- Hiemstra LA, Lo IKY, Fowler PJ. Effect of Fatigue on Knee Proprioception: Implications for Dynamic Stabilization. J Orthop Sports Phys Ther. 2001 Oct;31(10):598–605.
- Proske U. Exercise, fatigue and proprioception: a retrospective. Exp Brain Res. 2019 Oct 1;237(10):2447–59.
- 17. Meeusen R, Van Cutsem J, Roelands B. Endurance exercise-induced and mental fatigue and the brain. Exp Physiol. 2021;106(12):2294–8.
- 18. Bestwick-Stevenson T, Toone R, Neupert E, Edwards K, Kluzek S. Assessment of Fatigue and Recovery in Sport: Narrative Review. Int J Sports Med. 2022 Dec;43(14):1151–62.

- Blanch P, Gabbett TJ. Has the athlete trained enough to return to play safely? The acute:chronic workload ratio permits clinicians to quantify a player's risk of subsequent injury. Br J Sports Med. 2016 Apr 1;50(8):471–5.
- 20. Andrade R, Wik EH, Rebelo-Marques A, Blanch P, Whiteley R, Espregueira-Mendes J, et al. Is the Acute: Chronic Workload Ratio (ACWR) Associated with Risk of Time-Loss Injury in Professional Team Sports? A Systematic Review of Methodology, Variables and Injury Risk in Practical Situations. Sports Med. 2020 Sep 1;50(9):1613–35.
- Griffin A, Kenny IC, Comyns TM, Lyons M. The Association Between the Acute:Chronic Workload Ratio and Injury and its Application in Team Sports: A Systematic Review. Sports Med Auckl NZ. 2020 Mar;50(3):561–80.
- Impellizzeri FM, Marcora SM, Coutts AJ. Internal and External Training Load: 15 Years On. Int J Sports Physiol Perform. 2019 Oct 24;14(2):270–3.
- Maupin D, Schram B, Canetti E, Orr R. The Relationship Between Acute: Chronic Workload Ratios and Injury Risk in Sports: A Systematic Review. Open Access J Sports Med. 2020 Feb 24;11:51–75.
- Ahmun R, McCaig S, Tallent J, Williams S, Gabbett T. Association of Daily Workload, Wellness, and Injury and Illness During Tours in International Cricketers. Int J Sports Physiol Perform. 2019 Mar 1;14(3):369–77.
- Nakaoka G, Barboza SD, Verhagen E, van Mechelen W, Hespanhol L. The Association Between the Acute:Chronic Workload Ratio and Running-Related Injuries in Dutch Runners: A Prospective Cohort Study. Sports Med. 2021 Nov 1;51(11):2437–47.
- 26. Soligard T, Schwellnus M, Alonso JM, Bahr R, Clarsen B, Dijkstra HP, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. Br J Sports Med. 2016 Sep 1;50(17):1030–41.
- 27. Wang C, Vargas JT, Stokes T, Steele R, Shrier I. Analyzing Activity and Injury: Lessons Learned from the Acute:Chronic Workload Ratio. Sports Med. 2020 Jul 1;50(7):1243–54.

- Impellizzeri FM, McCall A, Ward P, Bornn L, Coutts AJ. Training Load and Its Role in Injury Prevention, Part 2: Conceptual and Methodologic Pitfalls. J Athl Train. 2020 Sep;55(9):893–901.
- Tiernan C, Comyns T, Lyons M, Nevill AM, Warrington G. The Association Between Training Load Indices and Injuries in Elite Soccer Players. J Strength Cond Res. 2022 Nov;36(11):3143.
- Hernán MA, Wang W, Leaf DE. Target Trial Emulation: A Framework for Causal Inference From Observational Data. JAMA. 2022 Dec 27;328(24):2446–7.
- 31. Wang C, Kaufman JS, Steele RJ, Shrier I. Target trial framework for determining the effect of changes in training load on injury risk using observational data: a methodological commentary. BMJ Open Sport Exerc Med. 2024 Jul 1;10(3):e002037.
- Hernán MA, Robins JM. Using Big Data to Emulate a Target Trial When a Randomized Trial Is Not Available. Am J Epidemiol. 2016 Apr 15;183(8):758–64.
- Hernán MA, Sauer BC, Hernández-Díaz S, Platt R, Shrier I. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. J Clin Epidemiol. 2016 Nov 1;79:70–5.
- Kutcher SA, Brophy JM, Banack HR, Kaufman JS, Samuel M. Emulating a Randomised Controlled Trial With Observational Data: An Introduction to the Target Trial Framework. Can J Cardiol. 2021 Sep 1;37(9):1365–77.
- 35. Admon AJ, Donnelly JP, Casey JD, Janz DR, Russell DW, Joffe AM, et al. Emulating a Novel Clinical Trial Using Existing Observational Data. Predicting Results of the PreVent Study. Ann Am Thorac Soc. 2019 Aug;16(8):998–1007.
- Didelez V, Haug U, Garcia-Albeniz X. Re: Are Target Trial Emulations the Gold Standard for Observational Studies? Epidemiology. 2024 Jan;35(1):e3.
- 37. Robins J. Interview with Jamie Robins. Obs Stud. 2022;8(2):37–76.

- 38. Matthews AA, Danaei G, Islam N, Kurth T. Target trial emulation: applying principles of randomised trials to observational studies. BMJ. 2022 Aug 30;378:e071108.
- Ranganathan P, Pramesh CS, Aggarwal R. Common pitfalls in statistical analysis: Intention-to-treat versus per-protocol analysis. Perspect Clin Res. 2016 Sep;7(3):144.
- 40. Black AM, Hagel BE, Palacios-Derflingher L, Schneider KJ, Emery CA. The risk of injury associated with body checking among Pee Wee ice hockey players: an evaluation of Hockey Canada's national body checking policy change. Br J Sports Med. 2017 Dec 1;51(24):1767–72.
- 41. Emery CA, Eliason P, Warriyar V, Palacios-Derflingher L, Black AM, Krolikowski M, et al. Body checking in non-elite adolescent ice hockey leagues: it is never too late for policy change aiming to protect the health of adolescents. Br J Sports Med. 2022 Jan 1;56(1):12–7.
- 42. Eliason P, Galarneau JM, Shill I, Kolstad A, Babul S, Mrazik M, et al. Factors Associated With Concussion Rates in Youth Ice Hockey Players: Data From the Largest Longitudinal Cohort Study in Canadian Youth Ice Hockey. Clin J Sport Med. 2023 Sep;33(5):497.
- 43. Eliason P, Galarneau JM, Schneider K, Babul S, Mrazik M, Lebrun C, et al. Higher hazard of concussion combined with recurrent concussion in female youth ice hockey compared with males: An opportunity for concussion prevention. Sports Orthop Traumatol. 2023 Jun 1;39(2):113–23.
- 44. Kolstad AT, Eliason PH, Galarneau JM, Black AM, Hagel BE, Emery CA. Protective equipment in youth ice hockey: are mouthguards and helmet age relevant to concussion risk? Br J Sports Med. 2023 May 1;57(10):571–7.
- Westreich D, Cole SR. Invited Commentary: Positivity in Practice. Am J Epidemiol. 2010 Mar 15;171(6):674–7.
- 46. Emery C, Palacios-Derflingher L, Black AM, Eliason P, Krolikowski M, Spencer N, et al. Does disallowing body checking in non-elite 13- to 14-year-old ice hockey leagues reduce

rates of injury and concussion? A cohort study in two Canadian provinces. Br J Sports Med. 2020 Apr 1;54(7):414–20.

- 47. Posit Team. RStudio: Integrated Development for R [Internet]. Boston, MA: Posit Software, PBC; 2024 [cited 2024 Feb 16]. Available from: https://www.posit.co/
- Wood S. mgcv: Mixed GAM Computation Vehicle with Automatic Smoothness Estimation [Internet]. 2023 [cited 2024 Feb 16]. Available from: https://cran.rproject.org/web/packages/mgcv/index.html
- Academy USS. Relationships Between Perfectionism, Training Load and Elite Junior Football Players' Self-Assessed and Coach-Assessed Skills [Internet]. The Sport Journal. 2017 [cited 2024 Mar 20]. Available from: https://thesportjournal.org/article/relationshipsbetween-perfectionism-training-load-and-elite-junior-football-players-self-assessed-andcoach-assessed-skills/
- Wagnsson S, Lindwall M, Gustafsson H. Participation in Organized Sport and Self-Esteem Across Adolescence: The Mediating Role of Perceived Sport Competence. J Sport Exerc Psychol. 2014 Dec 1;36(6):584–94.
- 51. Borresen J, Lambert MI. The Quantification of Training Load, the Training Response and the Effect on Performance. Sports Med. 2009 Sep 1;39(9):779–95.
- Fox JL, Stanton R, Sargent C, Wintour SA, Scanlan AT. The Association Between Training Load and Performance in Team Sports: A Systematic Review. Sports Med. 2018 Dec 1;48(12):2743–74.
- 53. Gabbett TJ. The training—injury prevention paradox: should athletes be training smarter and harder? Br J Sports Med. 2016 Mar 1;50(5):273–80.
- Hulin BT, Gabbett TJ, Lawson DW, Caputi P, Sampson JA. The acute:chronic workload ratio predicts injury: high chronic workload may decrease injury risk in elite rugby league players. Br J Sports Med. 2016 Feb 1;50(4):231–6.

- 55. Bowen L, Gross AS, Gimpel M, Li FX. Accumulated workloads and the acute:chronic workload ratio relate to injury risk in elite youth football players. Br J Sports Med. 2017 Mar 1;51(5):452–9.
- Colby MJ, Dawson B, Heasman J, Rogalski B, Gabbett TJ. Accelerometer and GPS-Derived Running Loads and Injury Risk in Elite Australian Footballers. J Strength Cond Res. 2014 Aug;28(8):2244.
- Malone S, Owen A, Newton M, Mendes B, Collins KD, Gabbett TJ. The acute:chonic workload ratio in relation to injury risk in professional soccer. J Sci Med Sport. 2017 Jun 1;20(6):561–5.
- Stares JJ, Dawson B, Peeling P, Heasman J, Rogalski B, Fahey-Gilmour J, et al. Subsequent Injury Risk Is Elevated Above Baseline After Return to Play: A 5-Year Prospective Study in Elite Australian Football. Am J Sports Med. 2019 Jul 1;47(9):2225– 31.
- 59. Wang C, Stokes T, Vargas JT, Steele R, Wedderkopp N, Shrier I. Predicting Injury Risk Over Changes in Physical Activity in Children Using the Acute:Chronic Workload Ratio. Am J Epidemiol. 2022 Mar 24;191(4):665–73.
- Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Clarsen B, Fagerland MW. Assessing the cumulative effect of long-term training load on the risk of injury in team sports. BMJ Open Sport Exerc Med. 2022 May 1;8(2):e001342.
- Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Clarsen B, Fagerland MW. Not straightforward: modelling non-linearity in training load and injury research. BMJ Open Sport — Exerc Med. 2021 Aug 6;7(3):e001119.
- Kalkhoven JT, Watsford ML, Coutts AJ, Edwards WB, Impellizzeri FM. Training Load and Injury: Causal Pathways and Future Directions. Sports Med [Internet]. 2021 Jan 5 [cited 2021 Jan 8]; Available from: https://doi.org/10.1007/s40279-020-01413-6

- Willer B, Kroetsch B, Darling S, Hutson A, Leddy J. Injury Rates in House League, Select, and Representative Youth Ice Hockey. Med Sci Sports Exerc. 2005 Oct;37(10):1658.
- 64. Lolli L, Batterham AM, Hawkins R, Kelly DM, Strudwick AJ, Thorpe R, et al. Mathematical coupling causes spurious correlation within the conventional acute-tochronic workload ratio calculations. Br J Sports Med. 2017 Nov 3;bjsports-2017-098110.
- Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Tabben M, Chamari K, Clarsen B, et al. A new statistical approach to training load and injury risk: separating the acute from the chronic load. Biol Sport. 2023;41(1):119–34.
- 66. Hartwig TB, Gabbett TJ, Naughton G, Duncan C, Harries S, Perry N. Training and match volume and injury in adolescents playing multiple contact team sports: A prospective cohort study. Scand J Med Sci Sports. 2019;29(3):469–75.
- Zouhal H, Boullosa D, Ramirez-Campillo R, Ali A, Granacher U. Editorial: Acute: Chronic Workload Ratio: Is There Scientific Evidence? Front Physiol. 2021 May 7;12:669687.
- West SW, Clubb J, Torres-Ronda L, Howells D, Leng E, Vescovi JD, et al. More than a Metric: How Training Load is Used in Elite Sport for Athlete Management. Int J Sports Med. 2021 Apr;42(4):300–6.
- 69. Hernán MA, Hernández-Díaz S. Beyond the intention to treat in comparative effectiveness research. Clin Trials Lond Engl. 2012 Feb;9(1):48.

Table 1. Target trial specification and emulation to estimate the intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries.

Protocol	Hypothetical Target Trial	Observational Study
Component		
Eligibility Criteria	 Adolescent ice hockey players aged 13 to 17 years in British Columbia and Alberta Have been participating in ice hockey practices and games for at least 4 weeks Not injured or recovering from injury in the previous 4 weeks Individuals are eligible at one time point 	 Same as for target trial, but participants must have been part of the Safe2Play parent study and have 5 weeks of participation data (previous 4 weeks and current week) Individuals can be eligible at multiple time points
Intervention	 Relative changes in participation duration (number of minutes of ice hockey practices and games) defined by the ACWR Chronic load: daily average participation duration over the previous 4 weeks Acute load: daily average planned participation duration over the next 7 days Assigned participation duration for the next 7 days is calculated as the participant's chronic load multiplied by their assigned ACWR Comparison is an ACWR of 1, where the acute load (daily average participation duration over the next 7 days) is equal to the chronic load (daily average participation duration over the next 7 days) is equal to the chronic load (daily average participation duration over the next 7 days) is equal to the chronic load (daily average participation duration over the next 7 days) is equal to the chronic load (daily average participation duration over the next 7 days) is equal to the chronic load (daily average participation duration over the next 9 days) 	- Same as target trial
Intervention Assignment	 Participants randomly assigned at baseline to a particular ACWR value ranging from 0.1 to 5.0, in 0.1-unit increments 	 Observed ACWRs calculated for each individual at each eligible time point Modelled ACWR-injury relationship adjusted for baseline confounders (chronic load, age group) to emulate randomization Assigned all individuals at each eligible time point to the same

		ACWR, and estimated injury risks using model • Done for ACWR values ranging from 0.1 to 5.0, in 0.1-unit increments
Outcome	- Any ice hockey-related injury occurring during participation in practices or games resulting in medical attention, inability to complete the session, or missed participation from subsequent sessions	- Same as target trial
Follow-Up Period	- 7 days starting from eligibility assessment and randomization	- 7 days starting from eligibility assessment
Causal Contrasts of Interest	- Intention-to-treat effect (effect of assigning a X-fold change in planned participation duration)	- Intention-to-treat effect (effect of a X-fold change in planned participation duration as determined by the team's planned participation schedule)
Subgroup Analyses	- Stratification by league bodychecking status	- Same as target trial
Statistical Analyses	- Compare injury risks for each intervention group to the injury risk for the group where ACWR was assigned as 1.0	- Compare predicted injury risks among all eligible individuals under each hypothetical intervention to the injury risk for the hypothetical intervention where ACWR was assigned as 1.0

Abbreviations: ACWR, acute:chronic workload ratio

Table 2. Characteristics of participants and player-trials included in a target trial emulation to estimate the intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries. Participants were from the Safe2Play parent study (2013-2017). There was a total of 2,633 eligible participants.

	No. of player-trials (%) (n=133,332)		
Characteristic	Eligible player-trials (n=115,821; 86.9%)	Ineligible player-trials ^a (n=17,511; 11.3%)	
Age group			
Under-15 (13 to 14 years)	72,137 (62.3%)	9,385 (53.6%)	
Under-18 (15 to 17 years)	43,684 (37.7%)	8,225 (46.4%)	
Sex			
Male	101,040 (87.2%)	15,316 (87.5%)	
Female	14,781 (12.8%)	2,195 (12.5%)	
League bodychecking status			
Allows bodychecking	78,102 (67.4%)	13,291 (75.3%)	
Disallows bodychecking	37,719 (32.6%)	4,319 (24.7%)	

^a Ineligible player-trials were those where the player was injured or recovering from injury in the previous 4 weeks, or had been absent from all hockey practices and games in the previous 4 weeks.



Figure 1. Participant flow chart for the emulation of a target trial assessing the effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries.



Figure 2. Intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries. Participation duration was measured as minutes of hockey practices and games. X-fold changes were measured as the ratio of the planned average daily participation duration in the current week to the average daily participation duration over the previous 4 weeks (uncoupled ACWR). Risk ratios were calculated relative to no change in participation duration (1.0-fold). The line represents estimated effects for the study population; the shaded area represents 95% confidence intervals for injury risk. All axes are on the log-scale.



Figure 3. Intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries, by league bodychecking status. Participation duration was measured as minutes of hockey practices and games. X-fold changes were measured as the ratio of the planned average daily participation duration in the current week to the average daily participation duration over the previous 4 weeks. Risk ratios were calculated relative to no change in participation duration (1.0-fold). The lines represent estimated effects for the study population stratified by league bodychecking status; the shaded areas represent 95% confidence intervals for injury risk. All axes are on the log-scale.



Figure 4. Intention-to-treat effect of changing participation duration on concussion risk among adolescent ice hockey players without recent injuries. Participation duration was measured as minutes of hockey practices and games. X-fold changes were measured as the ratio of the planned average daily participation duration in the current week to the average daily participation duration over the previous 4 weeks (uncoupled ACWR). Risk ratios were calculated relative to no change in participation duration (1.0-fold). The line represents estimated effects for the study population; the shaded area represents 95% confidence intervals for injury risk. All axes are on the log-scale.

Supplementary Material



Figure S1. Schematic displaying the specification of the intervention in a target trial estimating the intention-to-treat effect of changing participation duration on injury risk. (a) The intervention is a change in participation duration defined by the uncoupled acute:chronic workload ratio (ACWR). The uncoupled ACWR is calculated as the acute load divided by the chronic load. (b) Participants can contribute multiple overlapping player-trials (trials "n" and "n+1") as long as they meet eligibility criteria at time zero (t0).



Figure S2. Directed acyclic graph for the relationship between planned changes in participation and injury among adolescent ice hockey players. Previous participation, age group, and league bodychecking status are identified as baseline confounders because they are upstream of both the exposure and outcome. Sex is not a baseline confounder because it is only upstream of the outcome.



Figure S3. Risk ratios for the intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries. Participation duration was measured as minutes of hockey practices and games. X-fold changes were measured as the ratio of the planned average daily participation duration in the current week to the average daily participation duration over the previous 4 weeks. Risk ratios were calculated relative to no change in participation duration (1.0-fold). The line represents estimated effects for the study population; the shaded area represents 95% confidence intervals for injury risk ratios. All axes are on the log-scale.



(current week/previous 4 weeks)

Figure S4. Risk ratios for the intention-to-treat effect of changing participation duration on injury risk among adolescent ice hockey players without recent injuries, by league bodychecking status. Participation duration was measured as minutes of hockey practices and games. X-fold changes were measured as the ratio of the planned average daily participation duration in the current week to the average daily participation duration over the previous 4 weeks. Risk ratios were calculated relative to no change in participation duration (1.0-fold) by league bodychecking status. The lines represent estimated effects for the study population stratified by league bodychecking status; the shaded areas represent 95% confidence intervals for injury risk ratios. All axes are on the log-scale.

Chapter 9: Discussion

Preface

In this chapter, I provide a summary of the findings in this thesis and their significance, discuss the limitations of the included manuscripts, and discuss future directions regarding research on the relationship between changes in physical activity and injury.

Summary of Findings

Although physical activity plays a crucial role in the healthy development and socialization of children and adolescents,² participation in physical activity and sport has an inherent risk of injury.⁴ Maximizing the benefits of physical activity while minimizing injury risk is an important goal for athletes, parents, coaches, and clinicians. The objectives of this thesis were to assess the relationship between changes in physical activity levels and injury risk among children and adolescents, and to inform the methodology for future research in this area.

In Chapter 4 (Manuscript 1), I provided an overview of musculoskeletal sport injuries targeted towards epidemiologists who might not have substantive or clinical expertise in this area. I reviewed the definition of a sport injury from both biological and epidemiological aspects, and explained how these relate to common categorizations of sport injuries in epidemiologic research. I summarized the tissues and organs of the musculoskeletal system and common injuries that occur to them, and covered the principles of healing and rehabilitation of sport injuries.

In Chapter 5 (Manuscript 2), I employed flexible weighted cumulative exposure methods to assess the relative importance of activity done in past weeks on current injury risk in a cohort of Danish schoolchildren. I showed that high levels of activity done in the recent past (2 to 9 weeks ago) were associated with increased injury risk, conditional on the activity level in the current week, whereas activity done further in the past (11 to 20 weeks ago) were associated with decreased injury risk. For instance, being minimally active 11 to 20 weeks ago but highly active 1 to 10 weeks ago was associated with a 40% greater hazard of first injury than being minimally active for the previous 20 weeks (HR = 1.63, 95% CI: 1.18-2.23), whereas being highly active 11

to 20 weeks ago but minimally active 1 to 10 weeks ago was associated with a 20% lower hazard of first injury than being minimally active for the previous 20 weeks (HR = 0.81, 95% CI: 0.62-1.00). Although patterns were similar between girls and boys, the association between activity and injury appeared to be stronger among girls.

In Chapter 6 (Manuscript 3), I explored the problem of immortal time bias in observational studies assessing the relationship between activity and injury risk. I illustrated how immortal time bias may occur when the assessment of activity is not aligned with the start of follow-up for injury. I provide several options to mitigate this bias, including assessing injury risk subsequent to the time window over which activity is assessed, using planned activity rather than observed activity as the exposure of interest, or through cloning and censoring when activity is defined as a categorical exposure.

In Chapter 7 (Manuscript 4), I illustrated how the target trial framework might be applied to studies of activity and injury. I discussed the major components of a target trial protocol and specific considerations for designing a protocol to assess the relationship between changes in activity and injury risk.

In Chapter 8 (Manuscript 5), I applied the target trial framework to estimate the intention-to-treat effect of changes in activity on injury risk among adolescent ice hockey players. I found that while injury risk increased consistently with planned increases in hockey participation duration, increases in risk were less than would be expected for the increase in exposure time at risk. For instance, a 2-fold increase in participation resulted in a 1.6-fold increase in risk (RR = 1.62, 95% CI: 1.33-1.98), whereas a 5-fold increase in participation resulted in a 2.4-fold increase in risk (RR = 2.35, 95% CI: 1.68-3.26). I found that decreases in participation resulted in decreases in injury risk, with a 0.4-fold injury risk for a 0.5-fold decrease in activity (RR = 0.43, 95% CI: 0.31-0.54)

Significance of Findings

This thesis contributes to the literature by providing insights into the relationship between changes in physical activity and injury risk in children and adolescents, and informing how these relationships might be analyzed in future studies to estimate causal effects.

Although there is a breadth of information available on the biological and clinical aspects of sport injuries, few texts link these aspects to the design of epidemiologic research in a concise and accessible manner. Manuscript 1 may act as an important resource for epidemiologists without substantive background in anatomy, injury definitions, mechanisms, and clinical care who are interested in conducting research relating to sport injuries.

Manuscript 2 is the first study to our knowledge that applied flexible WCE methods to sport injury epidemiology. The vast majority of studies in this area have summarized past activity patterns using unweighted or exponentially weighted moving averages.^{63,64} Further, the majority of studies have arbitrarily assumed that any activity done in the previous 3 to 4 weeks is protectively associated with injury.^{63,64} We are not aware of any other studies that have applied a data-driven approach to identify the relevant time window over which previous activity might influence injury risk. Flexible WCE methods are a promising approach to assign weights to previous activity using a data-driven method, and to identify etiologically relevant time windows for specific sporting contexts in future studies.

Manuscript 3 describes the bias resulting from measuring injury incidence during the same time window as exposure assessment. Although other authors have noted issues with temporality and the potential for artificially low loads for athletes injured early in the week,^{52,81} Manuscript 3 is the first to explicitly describe this issue as a form of immortal time bias. Immortal time bias is a commonly acknowledged problem in other areas of epidemiology,¹⁴⁹ and several design and analytic solutions have been outlined to avoid this bias.^{149,150} By identifying the source of immortal time bias in studies of activity and injury and outlining how it can be avoided in future studies, we hope to strengthen the methodology of future research in this area.

Manuscript 4 is the first commentary to our knowledge that describes how the target trial framework can be applied to observational studies of activity and injury. Despite being used to generate training recommendations, many existing studies are of a descriptive or predictive nature and suffer from methodological flaws that impact the validity of these recommendations.^{12,95} To provide valid recommendations, researchers should aim for causal inferences.^{95,151} As a conceptual framework for conducting causal inference using observational data,¹⁵² the target trial framework has potential to be a valuable tool for researchers aiming to

estimate the causal effect of changing physical activity levels on injury risk. Our hope is that this manuscript can provide researchers in sport injury epidemiology the necessary background and guidance to employ target trial emulation, thereby strengthening the methodology of future research which may be used to generate valid training recommendations.

Manuscript 5 applies the target trial framework to a study of adolescent ice hockey players to determine the relationship between changes in duration of hockey participation and injury risk. It is the first study to our knowledge in the sport injury field to explicitly utilize the target trial framework. It is also one of few studies to employ flexible non-linear methods to model the relationship between activity and injury.⁷⁹ Our findings provide insight into the relationship between activity and injury.

Limitations

The studies included in this thesis (Manuscripts 2 and 5) have several limitations that require caution before concluding causal effects and generating training recommendations from our findings.

Measurement of physical activity

Based on the data available, we assessed physical activity as the number of recreational activity sessions per week in Manuscript 2, and the total duration of participation in ice hockey games and practices in Manuscript 5. As we did not assess any measures of intensity, nor did we assess duration of activity in Manuscript 2, there is likely to be heterogeneity in the actual loads experienced by participants. This heterogeneity may have decreased the precision of effect estimates. Further, the study population in Manuscript 2 engaged in a variety of different sports (e.g. soccer, handball, gymnastics). Our results represent an average across different sports, and results may not be generalizable to specific sports or to children outside Denmark who may participate in other types of sports. Finally, because our exposures are not precisely defined (i.e. may vary in activity frequency, duration, intensity, and/or type) and may result in different potential outcomes in the same person, we are likely to violate the consistency assumption of causal inference.^{153,154}

There is the possibility of measurement error in the assessment of physical activity. We made several assumptions regarding physical activity levels. In Manuscript 2, the number of physical activity sessions per week was the sum of physical education classes and parent-reported extracurricular activities. As we did not have data on school attendance, we assumed that participants always attended physical education classes. As such, physical activity may be overestimated in those who were absent from school. If reasons for absences were related to injury risk (e.g. individuals with an injury history did not participate), this could have biased estimates. However, we only assessed the time to first injury in a school year, so it is unlikely that previous injuries would have affected absences. Further, a sensitivity analysis where we included recent musculoskeletal pain as a covariate (accounting for the fact that individuals with pain may have decreased participation in physical education classes) did not affect findings. As such, any measurement error is likely to be non-differential and have resulted in attenuation of effect estimates.¹⁵⁵

In Manuscript 5, individuals who were reported as having partially participated in a game or practice were assigned a participation duration of 50% the session duration. This may have created error in the measurement of previous activity (current activity was based on planned participation). As we restricted eligibility to individuals who were uninjured in the previous 4 weeks, this error is unlikely to be differential by injured status. However, it may have caused non-differential measurement error and resulted in attenuation of effect estimates.¹⁵⁵

Selection bias

Selection affecting internal validity

There are multiple sources of potential selection bias in Manuscript 2. Firstly, we identified a protective association between activity done 11 to 20 weeks ago and injury risk. This association may be partially attributed to survivor bias,¹⁵⁶ where only those who were uninjured for a 20-week interval were observed for the entire 20-week time window. Participants who became injured before the 11- to 20-week interval would not have contributed to the association that was observed in this window. This is a known limitation of survival analyses, and could be mitigated with inverse probability of censoring weighting.^{157,158}

There is also the possibility of selection bias through loss-to-follow up. Participants were able to enter and leave the study at any point. It may be that individuals who experienced symptoms of injury (e.g. pain) were less likely to remain in the study, and were thus censored prior to recording a clinician-diagnosed injury. This could similarly be mitigated with IPCW.

Selection bias is less likely to affect findings from Manuscript 5, as participants were followed throughout the season. Reasons for non-participation were primarily due to the inability to identify a team designate to record participation and injuries,¹⁵⁹ rather than factors related to participation and injury risk.

Selection affecting generalizability

Selection of study participants might also affect generalizability of results. In Manuscript 2, participants were able to enter and leave the study at any point. Parents of children who were heavily involved in extracurricular physical activity may have been enthusiastic about study participation and more likely to be retained in the study, whereas parents of children who did minimal amounts of activity may have been less inclined to participate.

In Manuscript 5, we excluded individuals who were playing in Under-13 leagues because bodychecking is banned in these leagues. As such, our results are not generalizable to younger hockey participants.

Grouping acute and overuse injuries

We combined acute and overuse injuries as the outcome of interest in Manuscripts 2 and 5, although we performed a separate analysis restricted to concussions in Manuscript 5. We made this decision because stakeholders (e.g. athletes, parents, coaches) are concerned about injuries of all types, rather than preventing a certain type of injury. For instance, it would be ill-advised to provide recommendations for decreasing risk of overuse injuries (e.g. stress fractures) if the same recommendations would increase risk of acute injuries (e.g. concussions). Further, the mechanisms behind which unaccustomed activity causes acute and overuse injuries are likely to differ.⁹⁴ Whereas overuse injuries result from tissue fatigue causing decreased load capacity, acute injuries can occur at normal load capacities (although tissue and mental fatigue may play a

role).^{14,15,23,94} To advance our understanding of the causal mechanisms between changes in activity and injury, acute and overuse injuries should be studied separately.

Initial versus subsequent injuries

In Manuscript 2, we only included the first injury occurring for an individual within a given school year in analyses (individuals were censored after the first injury). We did not study subsequent injuries because subsequent injury risk is thought to be influenced by initial injuries,^{19,20} and activity patterns are likely to change after an initial injury (e.g. to give time for recovery). As such, the relationship between cumulative activity and injury risk is likely to differ between initial and subsequent injuries, and our results are not generalizable to subsequent injuries.

In Manuscript 5, we imposed eligibility criteria which required that individuals were uninjured and not recovering from injury in the previous 4 weeks. This was done to restrict analyses to initial injuries. This is similar to imposing a "washout" period to ensure that analyses are not confounded by previous exposures or outcomes.¹⁶⁰ We assumed that any influence of a previous injury on current injury risk would be minimal after having been uninjured (defined as being medically cleared to return to full participation and/or having returned to full participation) for 4 weeks. If this assumption is not valid, our results may be confounded by injury history.

Unmeasured confounding and exchangeability assumption

We did not have data on important indications for changes in activity patterns, including tiredness and soreness,^{23,94} in Manuscripts 2 and 5. These factors may also affect risk of injury, and are sources of unmeasured confounding. Individuals who performed different patterns of activity may not be considered exchangeable (i.e. have the same distribution of injury risk factors) even conditional on measured covariates, and as such we may violate the exchangeability condition of causal inference.¹⁵⁴

Positivity assumption

Under the positivity condition of causal inference, each participant should have a theoretically positive probability of receiving each level of treatment.^{154,161} In the context of Manuscript 5, this

means that each participant should theoretically be able to increase their physical activity across the range of ACWR values. We only assessed injury risk for ACWRs up to 5, based on the distribution of calculated ACWRs in the observed data. However, individuals who already have high amounts of hockey participation may not be able to increase their participation by up to 5fold. For instance, a team with an average past participation duration of 1.5 hours per day (the highest value observed in the dataset) would have to be able to increase their participation up to 7.5 hours per day, which while theoretically possible, may be implausible given other demands during the hockey season (e.g. school).

Accounting for repeated measures

In Manuscripts 2 and 5, we accounted for repeated measures solely using cluster bootstrapping.¹³⁷ This is a computationally inefficient method, but was necessary to derive risk ratios from logistic regression in Manuscript 5.¹³⁷

Although we explored using a random intercept applying GAMMs in the analyses for Manuscript 5 of this thesis, the predicted injury risks were extremely low and implausible. This may have occurred because the random effects could not be well estimated due to the flexibility of the non-parametric fixed effect for activity. Random effect estimation was likely affected by an imbalance of the number of observations per individual and/or the large number of individuals with no injuries. Further, we were unable to use GEE to account for repeated measures in GAMs because the structure of the GAMs did not allow a straightforward computation of robust standard errors using standard statistical packages. Applying mixed effect models and GEE in our analyses could have provided more robust estimates, and provided further insight into the relationship between changes in physical activity and injury risk.

Future Directions

Currently, there is a lack of evidence regarding the causal relationship between changes in activity and injury. Current training recommendations are based largely on statistical associations that are likely to be biased for causal effects, and are unlikely to be generalizable to all sporting contexts. Future research in this area should expand on the work presented in this thesis and employ methods that can be used to estimate causal effects.

In Manuscript 2, we demonstrated that previous physical activity patterns can be represented using flexible WCE methods within a Cox PH model. An extension would be to incorporate flexible WCE methods within generalized linear models (e.g. logistic models) to make inferences about injury risks. For instance, rather than arbitrarily defining change in activity as the ratio of current activity to the unweighted average activity over the previous 4 weeks, change in activity could be represented as the ratio of current activity to a WCE-determined weighted average of previous activity. Flexible WCE methods have been extended to allow the use of weight functions within linear mixed effect models.¹⁶²

In Manuscript 5, we estimated the intention-to-treat effects of relative changes in activity at single timepoints on injury risk. As such, we were not worried about time-varying confounding by previous activity or injury, and avoided introducing immortal time bias as outlined in Manuscript 3. Future research might investigate the per-protocol causal effect of cumulative physical activity as a time-varying treatment on injury risk (e.g. determining the injury risks associated with different week-to-week patterns of activity). This would require adjusting for time-varying confounding by activity and injury history, as well as other important confounders such as fatigue, soreness, or tiredness. This could be done using g-methods, ^{154,163} keeping in mind that extensions may be needed to accommodate continuous exposures such as physical activity.¹⁶⁴ Proper estimation of per-protocol effects also requires obtaining an unbiased measure of an individual's observed activity, as touched on in Manuscripts 3 and 4. This can be done for categorical exposures (e.g. increase in activity by more than 1.5-fold vs. less than 1.5-fold) through cloning and censoring, as discussed in Manuscript 3. However, the only solution we have up to this point for estimating per-protocol effects of continuous exposures of changes in activity is to impose an injury lag period. This obscures variations in activity leading directly up to an injury that are likely to affect injury risk. Finer data on the timing of injury (e.g. the exact minute within a practice session that an injury occurred) is needed to reduce bias in the measurement of exposure. Estimating per-protocol effects in this context remains a future challenge.

We only examined initial injuries in our analyses. However, subsequent injuries account for a considerable proportion of all injuries (10-25% of injuries are subsequent injuries to the same body location).¹⁶⁵ Risk factors for a subsequent injury are likely to differ from that of the initial

injury,¹⁶⁶ and it may be inappropriate to analyze initial and subsequent injuries together as a single outcome. Future research might employ recurrent event methods.^{167,168}

We note that in order to elucidate causal relationships, we must appropriately control for confounders between changes in physical activity and injury risk. This requires data on important confounders (e.g. soreness, tiredness, fatigue). Methods that control for unmeasured confounding such as instrumental variable approaches could also be explored.^{169,170} We previously suggested that an athlete's proposed training schedule might be a valid instrument for their actual activity performed.¹² However, there may be confounders that affect one's proposed training schedule as well as injury risk, such as fatigue. Proposed training schedule is only a valid instrumental variable if these factors are controlled for in analyses.¹²

Conclusion

Assessing the relationship between changes in physical activity and injury among children and adolescents is an important step in developing valid recommendations for injury prevention. In this thesis, I demonstrated how approaches employed in other substantive areas of epidemiology can be used in sport injury epidemiology, including using flexible weighted cumulative exposure methods to represent cumulative physical activity, and employing the target trial framework to minimize biases in observational analyses for causal effects. Although limitations in our data sources and analyses precluded us from definitively concluding causal effects, this thesis is a valuable resource for informing future research in this field.
References

- 1. Warburton DER, Nicol CW, Bredin SSD. Health benefits of physical activity: the evidence. CMAJ. 2006 Mar 14;174(6):801–9.
- 2. Janssen I, LeBlanc AG. Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. Int J Behav Nutr Phys Act. 2010;7(1):40.
- 3. Eime RM, Young JA, Harvey JT, Charity MJ, Payne WR. A systematic review of the psychological and social benefits of participation in sport for children and adolescents: informing development of a conceptual model of health through sport. Int J Behav Nutr Phys Act. 2013 Aug 15;10(1):98.
- Nauta J, Martin-Diener E, Martin BW, van Mechelen W, Verhagen E. Injury Risk During Different Physical Activity Behaviours in Children: A Systematic Review with Bias Assessment. Sports Med. 2015 Mar 1;45(3):327–36.
- Maffulli N, Longo UG, Spiezia F, Denaro V. Sports Injuries in Young Athletes: Long-Term Outcome and Prevention Strategies. The Physician and Sportsmedicine. 2010 Jun 1;38(2):29–34.
- 6. Caine D, Purcell L, Maffulli N. The child and adolescent athlete: a review of three potentially serious injuries. BMC Sports Sci Med Rehabil. 2014 Jun 10;6(1):22.
- Finch CF, Shee AW, Clapperton A. Time to add a new priority target for child injury prevention? The case for an excess burden associated with sport and exercise injury: population-based study. BMJ Open. 2014 Jul 1;4(7):e005043.
- Collard DCM, Verhagen EALM, Mechelen W van, Heymans MW, Chinapaw MJM.
 Economic burden of physical activity–related injuries in Dutch children aged 10–12. Br J Sports Med. 2011 Oct 1;45(13):1058–63.
- Fridman L, Fraser-Thomas JL, McFaull SR, Macpherson AK. Epidemiology of sportsrelated injuries in children and youth presenting to Canadian emergency departments from 2007–2010. BMC Sports Sci Med Rehabil. 2013 Dec 23;5(1):30.
- Parrish AM, Tremblay MS, Carson S, Veldman SLC, Cliff D, Vella S, et al. Comparing and assessing physical activity guidelines for children and adolescents: a systematic literature review and analysis. Int J Behav Nutr Phys Act. 2020 Feb 10;17(1):16.

- Soligard T, Schwellnus M, Alonso JM, Bahr R, Clarsen B, Dijkstra HP, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. Br J Sports Med. 2016 Sep 1;50(17):1030–41.
- 12. Wang C, Vargas JT, Stokes T, Steele R, Shrier I. Analyzing Activity and Injury: Lessons Learned from the Acute:Chronic Workload Ratio. Sports Med. 2020 Jul 1;50(7):1243–54.
- Impellizzeri FM, McCall A, Ward P, Bornn L, Coutts AJ. Training Load and Its Role in Injury Prevention, Part 2: Conceptual and Methodologic Pitfalls. J Athl Train. 2020 Sep;55(9):893–901.
- Bartlett R, Bussey M. Sports Biomechanics: Reducing Injury Risk and Improving Sports Performance [Internet]. Florence, UNITED STATES: Routledge; 2013 [cited 2019 Dec 21]. Available from: http://ebookcentral.proquest.com/lib/mcgill/detail.action?docID=957394
- Bahr R, Alfredson H, Järvinen M, Järvinen T, Khan K, Kjær M, et al. Types and Causes of Injuries. In: The IOC Manual of Sports Injuries [Internet]. John Wiley & Sons, Ltd; 2012 [cited 2023 Feb 15]. p. 1–24. Available from: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118467947.ch1
- Wind H, Gouttebarge V, Kuijer PPFM, Frings-Dresen MHW. Assessment of Functional Capacity of the Musculoskeletal System in the Context of Work, Daily Living, and Sport: A Systematic Review. J Occup Rehabil. 2005 Jun 1;15(2):253–72.
- Wiesinger HP, Kösters A, Müller E, Seynnes OR. Effects of Increased Loading on In Vivo Tendon Properties: A Systematic Review. Med Sci Sports Exerc. 2015 Sep;47(9):1885– 95.
- Bahr R, Cook J, Langberg H, MacAuley D, Matheson G, Mæhlum S. Treating Sports Injuries. In: The IOC Manual of Sports Injuries [Internet]. John Wiley & Sons, Ltd; 2012 [cited 2023 Feb 27]. p. 25–39. Available from: https://onlinelibrary.wiley.com/doi/abs/10.1002/9781118467947.ch2
- Baoge L, Van Den Steen E, Rimbaut S, Philips N, Witvrouw E, Almqvist KF, et al. Treatment of skeletal muscle injury: a review. ISRN Orthop. 2012;2012:689012.
- Toth C. The Epidemiology of Injuries to the Nervous System Resulting from Sport and Recreation. Neurologic Clinics. 2008 Feb 1;26(1):1–31.

- Meaney DF, Smith DH. Biomechanics of Concussion. Clinics in Sports Medicine. 2011 Jan 1;30(1):19–31.
- 22. Bahr R, Clarsen B, Derman W, Dvorak J, Emery CA, Finch CF, et al. International Olympic Committee consensus statement: methods for recording and reporting of epidemiological data on injury and illness in sport 2020 (including STROBE Extension for Sport Injury and Illness Surveillance (STROBE-SIIS)). Br J Sports Med [Internet]. 2020 Feb 18 [cited 2020 Mar 16]; Available from: https://bjsm.bmj.com/content/early/2020/02/28/bjsports-2019-101969
- 23. Bestwick-Stevenson T, Toone R, Neupert E, Edwards K, Kluzek S. Assessment of Fatigue and Recovery in Sport: Narrative Review. Int J Sports Med. 2022 Dec;43(14):1151–62.
- 24. Meeusen R, Van Cutsem J, Roelands B. Endurance exercise-induced and mental fatigue and the brain. Experimental Physiology. 2021;106(12):2294–8.
- 25. Enoka RM, Duchateau J. Muscle fatigue: what, why and how it influences muscle function. J Physiol. 2008 Jan 1;586(Pt 1):11–23.
- Wilkins JC, Valovich McLeod TC, Perrin DH, Gansneder BM. Performance on the Balance Error Scoring System Decreases After Fatigue. J Athl Train. 2004;39(2):156–61.
- Stevens ST, Lassonde M, de Beaumont L, Paul Keenan J. In-Game Fatigue Influences Concussions in National Hockey League Players. Research in Sports Medicine. 2008 Feb 7;16(1):68–74.
- Hiemstra LA, Lo IKY, Fowler PJ. Effect of Fatigue on Knee Proprioception: Implications for Dynamic Stabilization. Journal of Orthopaedic & Sports Physical Therapy. 2001 Oct;31(10):598–605.
- Proske U. Exercise, fatigue and proprioception: a retrospective. Exp Brain Res. 2019 Oct 1;237(10):2447–59.
- 30. Brahms M, Heinzel S, Rapp M, Mückstein M, Hortobágyi T, Stelzel C, et al. The acute effects of mental fatigue on balance performance in healthy young and older adults – A systematic review and meta-analysis. Acta Psychologica. 2022 May 1;225:103540.
- 31. Base NRC (US) SC for the W on WRMITR. Work Factors, Personal Factors, and Internal Loads: Biomechanics of Work Stressors. In: Work-Related Musculoskeletal Disorders: Report, Workshop Summary, and Workshop Papers [Internet]. National Academies Press

(US); 1999 [cited 2024 Mar 25]. Available from: https://www.ncbi.nlm.nih.gov/books/NBK230854/

- 32. Udby CL, Impellizzeri FM, Lind M, Nielsen RØ. How Has Workload Been Defined and How Many Workload-Related Exposures to Injury Are Included in Published Sports Injury Articles? A Scoping Review. Journal of Orthopaedic & Sports Physical Therapy. 2020 Oct;50(10):538–48.
- 33. Impellizzeri FM, Marcora SM, Coutts AJ. Internal and External Training Load: 15 Years
 On. International Journal of Sports Physiology and Performance. 2019 Oct 24;14(2):270–
 3.
- 34. Coutts AJ, Crowcroft S, Kempton T. Developing athlete monitoring systems: Theoretical basis and practical applications. In: Sport, Recovery, and Performance. Routledge; 2017.
- Impellizzeri FM, Shrier I, McLaren SJ, Coutts AJ, McCall A, Slattery K, et al. Understanding Training Load as Exposure and Dose. Sports Med [Internet]. 2023 Apr 6 [cited 2023 Apr 19]; Available from: https://doi.org/10.1007/s40279-023-01833-0
- Barisic A, Leatherdale ST, Kreiger N. Importance of Frequency, Intensity, Time and Type (FITT) in Physical Activity Assessment for Epidemiological Research. Can J Public Health. 2011 May 1;102(3):174–5.
- Westerterp KR. Doubly labelled water assessment of energy expenditure: principle, practice, and promise. Eur J Appl Physiol. 2017 Jul 1;117(7):1277–85.
- Ainslie PN, Reilly T, Westerterp KR. Estimating Human Energy Expenditure. Sports Med. 2003 Aug 1;33(9):683–98.
- Prince SA, Adamo KB, Hamel ME, Hardt J, Gorber SC, Tremblay M. A comparison of direct versus self-report measures for assessing physical activity in adults: a systematic review. Int J Behav Nutr Phys Act. 2008 Nov 6;5(1):56.
- 40. Borresen J, Lambert MI. The Quantification of Training Load, the Training Response and the Effect on Performance. Sports Med. 2009 Sep 1;39(9):779–95.
- 41. Ndahimana D, Kim EK. Measurement Methods for Physical Activity and Energy Expenditure: a Review. Clin Nutr Res. 2017 Apr;6(2):68–80.
- Sirard JR, Pate RR. Physical Activity Assessment in Children and Adolescents. Sports Med. 2001 May 1;31(6):439–54.

- Troiano RP, McClain JJ, Brychta RJ, Chen KY. Evolution of accelerometer methods for physical activity research. Br J Sports Med. 2014 Jul 1;48(13):1019–23.
- 44. Pedišić Ž, Bauman A. Accelerometer-based measures in physical activity surveillance: current practices and issues. Br J Sports Med. 2015 Feb 1;49(4):219–23.
- 45. Chambers R, Gabbett TJ, Cole MH, Beard A. The Use of Wearable Microsensors to Quantify Sport-Specific Movements. Sports Med. 2015 Jul 1;45(7):1065–81.
- 46. Van Hoye A, Nicaise V, Sarrazin P. Self-reported and objective physical activity measurement by active youth. Science & Sports. 2014 Apr 1;29(2):78–87.
- 47. Chen MJ, Fan X, Moe ST. Criterion-related validity of the Borg ratings of perceived exertion scale in healthy individuals: a meta-analysis. J Sports Sci. 2002 Nov;20(11):873–99.
- Abbiss CR, Peiffer JJ, Meeusen R, Skorski S. Role of Ratings of Perceived Exertion during Self-Paced Exercise: What are We Actually Measuring? Sports Med. 2015 Sep 1;45(9):1235–43.
- Haddad M, Padulo J, Chamari K. The Usefulness of Session Rating of Perceived Exertion for Monitoring Training Load Despite Several Influences on Perceived Exertion. International Journal of Sports Physiology and Performance. 2014 Sep 1;9(5):882–3.
- 50. Haddad M, Stylianides G, Djaoui L, Dellal A, Chamari K. Session-RPE Method for Training Load Monitoring: Validity, Ecological Usefulness, and Influencing Factors. Front Neurosci [Internet]. 2017 Nov 2 [cited 2019 Dec 30];11. Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC5673663/
- 51. Jesus RS de, Batista RÉS, Santos VME, Ohara D, Alves E da S, Ribeiro LFP. Exercise Duration Affects Session Ratings of Perceived Exertion as a Function of Exercise Intensity. Percept Mot Skills. 2021 Aug 1;128(4):1730–46.
- 52. Drew MK, Finch CF. The Relationship Between Training Load and Injury, Illness and Soreness: A Systematic and Literature Review. Sports Med. 2016 Jun 1;46(6):861–83.
- Eckard TG, Padua DA, Hearn DW, Pexa BS, Frank BS. The Relationship Between Training Load and Injury in Athletes: A Systematic Review. Sports Med. 2018 Aug 1;48(8):1929–61.
- 54. Stovitz SD, Shrier I. Injury rates in team sport events: tackling challenges in assessing exposure time. Br J Sports Med. 2012 Nov 1;46(14):960–3.

- 55. Bahr R, Krosshaug T. Understanding injury mechanisms: a key component of preventing injuries in sport. British Journal of Sports Medicine. 2005 Jun 1;39(6):324–9.
- Calvert TW, Banister EW, Savage MV, Bach T. A Systems Model of the Effects of Training on Physical Performance. IEEE Transactions on Systems, Man, and Cybernetics. 1976 Feb;SMC-6(2):94–102.
- 57. Banister EW, Calvert TW. Planning for future performance: implications for long term training. Can J Appl Sport Sci. 1980 Sep;5(3):170–6.
- Chiu LZF, Barnes JL. The Fitness-Fatigue Model Revisited: Implications for Planning Short- and Long-Term Training. Strength & Conditioning Journal. 2003 Dec;25(6):42–51.
- Hulin BT, Gabbett TJ, Blanch P, Chapman P, Bailey D, Orchard JW. Spikes in acute workload are associated with increased injury risk in elite cricket fast bowlers. Br J Sports Med. 2014 Apr 1;48(8):708–12.
- 60. Blanch P, Gabbett TJ. Has the athlete trained enough to return to play safely? The acute:chronic workload ratio permits clinicians to quantify a player's risk of subsequent injury. Br J Sports Med. 2016 Apr 1;50(8):471–5.
- 61. Windt J, Gabbett TJ. Is it all for naught? What does mathematical coupling mean for acute:chronic workload ratios? Br J Sports Med. 2018 May 28;bjsports-2017-098925.
- Wang C, Stokes T, Vargas JT, Steele R, Wedderkopp N, Shrier I. Predicting Injury Risk Over Changes in Physical Activity in Children Using the Acute:Chronic Workload Ratio. American Journal of Epidemiology. 2022 Mar 24;191(4):665–73.
- 63. Andrade R, Wik EH, Rebelo-Marques A, Blanch P, Whiteley R, Espregueira-Mendes J, et al. Is the Acute: Chronic Workload Ratio (ACWR) Associated with Risk of Time-Loss Injury in Professional Team Sports? A Systematic Review of Methodology, Variables and Injury Risk in Practical Situations. Sports Med. 2020 Sep 1;50(9):1613–35.
- Griffin A, Kenny IC, Comyns TM, Lyons M. The Association Between the Acute:Chronic Workload Ratio and Injury and its Application in Team Sports: A Systematic Review. Sports Med. 2020 Mar;50(3):561–80.
- Maupin D, Schram B, Canetti E, Orr R. The Relationship Between Acute: Chronic Workload Ratios and Injury Risk in Sports: A Systematic Review. Open Access Journal of Sports Medicine. 2020 Feb 24;11:51–75.

- 66. Fanchini M, Rampinini E, Riggio M, Coutts AJ, Pecci C, McCall A. Despite association, the acute:chronic work load ratio does not predict non-contact injury in elite footballers. Science and Medicine in Football. 2018 Apr 3;2(2):108–14.
- Delecroix B, McCall A, Dawson B, Berthoin S, Dupont G. Workload and non-contact injury incidence in elite football players competing in European leagues. European Journal of Sport Science. 2018 Oct 21;18(9):1280–7.
- McCall A, Dupont G, Ekstrand J. Internal workload and non-contact injury: a one-season study of five teams from the UEFA Elite Club Injury Study. Br J Sports Med. 2018 Dec 1;52(23):1517–22.
- 69. Stares J, Dawson B, Peeling P, Heasman J, Rogalski B, Drew M, et al. Identifying high risk loading conditions for in-season injury in elite Australian football players. Journal of Science and Medicine in Sport. 2018 Jan 1;21(1):46–51.
- Sampson JA, Murray A, Williams S, Halseth T, Hanisch J, Golden G, et al. Injury riskworkload associations in NCAA American college football. Journal of Science and Medicine in Sport. 2018;21(12):1215–20.
- Carey DL, Blanch P, Ong KL, Crossley KM, Crow J, Morris ME. Training loads and injury risk in Australian football—differing acute: chronic workload ratios influence match injury risk. Br J Sports Med. 2017 Aug 1;51(16):1215–20.
- 72. Malone S, Owen A, Mendes B, Hughes B, Collins K, Gabbett TJ. High-speed running and sprinting as an injury risk factor in soccer: Can well-developed physical qualities reduce the risk? Journal of Science and Medicine in Sport. 2018 Mar 1;21(3):257–62.
- 73. Menaspà P. Are rolling averages a good way to assess training load for injury prevention? Br J Sports Med. 2017 Apr 1;51(7):618–9.
- 74. Hawley JA. Adaptations Of Skeletal Muscle To Prolonged, Intense Endurance Training. Clinical and Experimental Pharmacology and Physiology. 2002;29(3):218–22.
- Mujika I, Padilla S. Detraining: Loss of Training-Induced Physiological and Performance Adaptations. Part I. Sports Med. 2000 Aug 1;30(2):79–87.
- Williams S, West S, Cross MJ, Stokes KA. Better way to determine the acute:chronic workload ratio? Br J Sports Med. 2017 Feb 1;51(3):209–10.

- 77. Murray NB, Gabbett TJ, Townshend AD, Blanch P. Calculating acute:chronic workload ratios using exponentially weighted moving averages provides a more sensitive indicator of injury likelihood than rolling averages. Br J Sports Med. 2017 May 1;51(9):749–54.
- 78. Hulin BT, Gabbett TJ, Pickworth NJ, Johnston RD, Jenkins DG. Relationships Among PlayerLoad, High-Intensity Intermittent Running Ability, and Injury Risk in Professional Rugby League Players. International Journal of Sports Physiology and Performance. 2020 Mar 1;15(3):423–9.
- Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Clarsen B, Fagerland MW. Not straightforward: modelling non-linearity in training load and injury research. BMJ Open Sport Exerc Med. 2021 Aug 6;7(3):e001119.
- Bache-Mathiesen LK, Andersen TE, Dalen-Lorentsen T, Clarsen B, Fagerland MW. Assessing the cumulative effect of long-term training load on the risk of injury in team sports. BMJ Open Sport & Exercise Medicine. 2022 May 1;8(2):e001342.
- Windt J, Ardern CL, Gabbett TJ, Khan KM, Cook CE, Sporer BC, et al. Getting the most out of intensive longitudinal data: a methodological review of workload–injury studies. BMJ Open. 2018 Oct 2;8(10):e022626.
- Dijkhuis TB, Otter R, Aiello M, Velthuijsen H, Lemmink K. Increase in the Acute:Chronic Workload Ratio relates to Injury Risk in Competitive Runners. Int J Sports Med. 2020 Oct;41(11):736–43.
- Nakaoka G, Barboza SD, Verhagen E, van Mechelen W, Hespanhol L. The Association Between the Acute:Chronic Workload Ratio and Running-Related Injuries in Dutch Runners: A Prospective Cohort Study. Sports Med. 2021 Nov 1;51(11):2437–47.
- Pollen TR, Ebaugh D, Warren M, Milner CE, Taylor JA, Silfies SP. Workload and Noncontact Musculoskeletal Injury in Collegiate Swimmers: A Prospective Cohort Study. Journal of Athletic Training. 2021 Jun 15;57(5):470–7.
- 85. Moreno-Pérez V, Prieto J, Del Coso J, Lidó-Micó JE, Fragoso M, Penalva FJ, et al. Association of acute and chronic workloads with injury risk in high-performance junior tennis players. European Journal of Sport Science. 2021 Aug 3;21(8):1215–23.
- Zouhal H, Boullosa D, Ramirez-Campillo R, Ali A, Granacher U. Editorial: Acute: Chronic Workload Ratio: Is There Scientific Evidence? Front Physiol. 2021 May 7;12:669687.

- Rico-González M, Oliveira R, González Fernández FT, Clemente FM. Acute:chronic workload ratio and training monotony variations over the season in youth soccer players: A systematic review. International Journal of Sports Science & Coaching. 2023 Aug 1;18(4):1333–41.
- 88. Workload management app, training monitoring software, acute:chronic ratio, sport data analytics, GPS, heart rate, Gabbett, training planning, workout builder, exercise tracking | Athlete Monitoring software, Athlete Management System, Injury tracking software, Electronic medical records [Internet]. [cited 2024 Mar 27]. Available from: https://www.athletemonitoring.com/workload-management/
- 89. ProHockeyStrength.com. Load Monitoring in Hockey: Using Gabbet's Acute-Chronic Workload Ratio [Internet]. 2022 [cited 2024 Mar 27]. Available from: https://www.prohockeystrength.com/public/Load-Monitoring-in-Hockey-Using-Gabbets-AcuteChronic-Workload-Ratio.cfm
- 90. Goom T. Is there an ideal workload ratio to predict and prevent injuries? [Internet]. RunningPhysio. 2022 [cited 2024 Mar 27]. Available from: https://www.runningphysio.com/workload/
- 91. Dalen-Lorentsen T, Andersen TE, Bjørneboe J, Vagle M, Martin KN, Kleppen M, et al. A Cherry, Ripe for Picking: The Relationship Between the Acute-Chronic Workload Ratio and Health Problems. Journal of Orthopaedic & Sports Physical Therapy. 2021 Apr;51(4):162–73.
- 92. Impellizzeri FM, Ward P, Coutts AJ, Bornn L, McCall A. Training Load and Injury Part 2: Questionable Research Practices Hijack the Truth and Mislead Well-Intentioned Clinicians. J Orthop Sports Phys Ther. 2020 Oct;50(10):577–84.
- 93. Nielsen RO, Simonsen NS, Casals M, Stamatakis E, Mansournia MA. Methods matter and the 'too much, too soon' theory (part 2): what is the goal of your sports injury research? Are you describing, predicting or drawing a causal inference? Br J Sports Med. 2020 Nov 1;54(22):1307–9.
- 94. Kalkhoven JT, Watsford ML, Coutts AJ, Edwards WB, Impellizzeri FM. Training Load and Injury: Causal Pathways and Future Directions. Sports Med [Internet]. 2021 Jan 5 [cited 2021 Jan 8]; Available from: https://doi.org/10.1007/s40279-020-01413-6

- 95. Impellizzeri FM, Tenan MS, Kempton T, Novak A, Coutts AJ. Acute:Chronic Workload Ratio: Conceptual Issues and Fundamental Pitfalls. International Journal of Sports Physiology and Performance. 2020 Jun 5;15(6):907–13.
- 96. Atkinson G, Batterham A. The Use of Ratios and Percentage Changes in Sports Medicine: Time for a Rethink? . Int J Sports Med. 2012 Jul;33(07):505–6.
- 97. Lolli L, Batterham AM, Hawkins R, Kelly DM, Strudwick AJ, Thorpe R, et al. Mathematical coupling causes spurious correlation within the conventional acute-tochronic workload ratio calculations. Br J Sports Med. 2017 Nov 3;bjsports-2017-098110.
- Knowles SB, Marshall SW, Guskiewicz KM. Issues in Estimating Risks and Rates in Sports Injury Research. J Athl Train. 2006;41(2):207–15.
- 99. Shrier I, Wang C, Stokes T, Trejovargas J, Stovitz SD, Steele RJ. Causal effects, workload and injury risk: The importance of specifying the research question. Journal of Science and Medicine in Sport. 2022 Jul 1;25(7):574–8.
- Schober P, Vetter TR. Survival Analysis and Interpretation of Time-to-Event Data: The Tortoise and the Hare. Anesth Analg. 2018 Sep;127(3):792–8.
- Ullah S, Gabbett TJ, Finch CF. Statistical modelling for recurrent events: an application to sports injuries. Br J Sports Med. 2014 Sep;48(17):1287–93.
- 102. Hernán MA. The Hazards of Hazard Ratios. Epidemiology. 2010 Jan;21(1):13-5.
- Hopkins WG, Marshall SW, Quarrie KL, Hume PA. Risk Factors and Risk Statistics for Sports Injuries. Clinical Journal of Sport Medicine. 2007 May;17(3):208.
- 104. Caine D, Maffulli N, Caine C. Epidemiology of Injury in Child and Adolescent Sports: Injury Rates, Risk Factors, and Prevention. Clinics in Sports Medicine. 2008 Jan 1;27(1):19–50.
- 105. Ranganathan P, Pramesh CS, Aggarwal R. Common pitfalls in statistical analysis: Intention-to-treat versus per-protocol analysis. Perspectives in Clinical Research. 2016 Sep;7(3):144.
- 106. Strobel A, Wienke A, Kuss O. How hazardous are hazard ratios? An empirical investigation of individual patient data from 27 large randomized clinical trials. Eur J Epidemiol. 2023 Aug 1;38(8):859–67.
- Rothman KJ, Greenland S, Lash TL. Modern Epidemiology. Lippincott Williams & Wilkins; 2008. 776 p.

- 108. Noordzij M, van Diepen M, Caskey FC, Jager KJ. Relative risk versus absolute risk: one cannot be interpreted without the other. Nephrology Dialysis Transplantation. 2017 Apr 1;32(suppl_2):ii13–8.
- 109. GREENLAND S. INTERPRETATION AND CHOICE OF EFFECT MEASURES IN EPIDEMIOLOGIC ANALYSES1. American Journal of Epidemiology. 1987 May 1;125(5):761–8.
- 110. Cummings P. The Relative Merits of Risk Ratios and Odds Ratios. Archives of Pediatrics & Adolescent Medicine. 2009 May 4;163(5):438–45.
- Sjölander A, Dahlqwist E, Zetterqvist J. A Note on the Noncollapsibility of Rate Differences and Rate Ratios. Epidemiology. 2016 May;27(3):356.
- Shrier I, Steele RJ, Hanley J, Rich B. Analyses of Injury Count Data: Some Do's and Don'ts. Am J Epidemiol. 2009 Nov 15;170(10):1307–15.
- 113. Harrell FE. Cox Proportional Hazards Regression Model. In: Harrell Jr Frank E, editor. Regression Modeling Strategies: With Applications to Linear Models, Logistic and Ordinal Regression, and Survival Analysis [Internet]. Cham: Springer International Publishing; 2015 [cited 2024 Mar 29]. p. 475–519. Available from: https://doi.org/10.1007/978-3-319-19425-7_20
- 114. Dobson A, Barnett A. An Introduction to Generalized Linear Models. 3rd ed. New York: Chapman and Hall/CRC; 2008. 320 p.
- Hastie T, Tibshirani R. Generalized additive models for medical research. Stat Methods Med Res. 1995 Sep 1;4(3):187–96.
- 116. McCullagh P. Generalized Linear Models. 2nd ed. New York: Routledge; 2019. 532 p.
- 117. Hastie TJ. Generalized Additive Models. In: Statistical Models in S. Routledge; 1992.
- Carroll KJ. On the use and utility of the Weibull model in the analysis of survival data. Controlled Clinical Trials. 2003 Dec 1;24(6):682–701.
- Zhang Z, Reinikainen J, Adeleke KA, Pieterse ME, Groothuis-Oudshoorn CGM. Timevarying covariates and coefficients in Cox regression models. Ann Transl Med. 2018 Apr;6(7):121.
- 120. Kleinbaum DG, Klein M. The Stratified Cox Procedure. In: Kleinbaum DG, Klein M, editors. Survival Analysis: A Self-Learning Text [Internet]. New York, NY: Springer;

2012 [cited 2024 Apr 9]. p. 201–40. Available from: https://doi.org/10.1007/978-1-4419-6646-9_5

- 121. Aalen OO. Further results on the non-parametric linear regression model in survival analysis. Statistics in Medicine. 1993;12(17):1569–88.
- 122. Aalen OO, Cook RJ, Røysland K. Does Cox analysis of a randomized survival study yield a causal treatment effect? Lifetime Data Anal. 2015 Oct 1;21(4):579–93.
- 123. Harrell FE. Binary Logistic Regression. In: Harrell Jr Frank E, editor. Regression Modeling Strategies: With Applications to Linear Models, Logistic and Ordinal Regression, and Survival Analysis [Internet]. Cham: Springer International Publishing; 2015 [cited 2024 Mar 29]. p. 219–74. Available from: https://doi.org/10.1007/978-3-319-19425-7_10
- Williamson T, Eliasziw M, Fick GH. Log-binomial models: exploring failed convergence. Emerg Themes Epidemiol. 2013 Dec 13;10(1):14.
- Naimi AI, Whitcomb BW. Estimating Risk Ratios and Risk Differences Using Regression. American Journal of Epidemiology. 2020 Jun 1;189(6):508–10.
- 126. Muller CJ, MacLehose RF. Estimating predicted probabilities from logistic regression: different methods correspond to different target populations. Int J Epidemiol. 2014 Jun;43(3):962–70.
- Cornelius AE, Brewer BW, Van Raalte JL. Applications of multilevel modeling in sport injury rehabilitation research. International Journal of Sport and Exercise Psychology. 2007 Jan 1;5(4):387–405.
- 128. Williamson DS, Bangdiwala SI, Marshall SW, Waller AE. Repeated measures analysis of binary outcomes: applications to injury research. Accid Anal Prev. 1996 Sep;28(5):571–9.
- Dean CB, Nielsen JD. Generalized linear mixed models: a review and some extensions. Lifetime Data Anal. 2007 Dec 1;13(4):497–512.
- Chen C. Generalized additive mixed models. Communications in Statistics Theory and Methods. 2000 Jan 1;29(5–6):1257–71.
- 131. Hubbard AE, Ahern J, Fleischer NL, Laan MV der, Lippman SA, Jewell N, et al. To GEE or Not to GEE: Comparing Population Average and Mixed Models for Estimating the Associations Between Neighborhood Risk Factors and Health. Epidemiology. 2010 Jul;21(4):467–74.

- 132. Wood S, Scheipl F. gamm4: Generalized Additive Mixed Models using 'mgcv' and 'lme4' [Internet]. 2020 [cited 2020 Apr 25]. Available from: https://CRAN.Rproject.org/package=gamm4
- Hastie T. gam: Generalized Additive Models [Internet]. 2023 [cited 2024 Mar 30].
 Available from: https://cran.r-project.org/web/packages/gam/index.html
- 134. Demidenko E. Mixed Models: Theory and Applications with R. John Wiley & Sons; 2013.768 p.
- 135. Hanley JA, Negassa A, Edwardes MD deB., Forrester JE. Statistical Analysis of Correlated Data Using Generalized Estimating Equations: An Orientation. American Journal of Epidemiology. 2003 Feb 15;157(4):364–75.
- Højsgaard S, Halekoh U, Yan J. The R Package geepack for Generalized Estimating Equations. Journal of Statistical Software. 2006;15:1–11.
- Huang FL. Using Cluster Bootstrapping to Analyze Nested Data With a Few Clusters. Educ Psychol Meas. 2018 Apr;78(2):297–318.
- Diciccio TJ, Romano JP. A Review of Bootstrap Confidence Intervals. Journal of the Royal Statistical Society: Series B (Methodological). 1988 Jan 1;50(3):338–54.
- Kelly TL, Salter A, Pratt NL. The weighted cumulative exposure method and its application to pharmacoepidemiology: A narrative review. Pharmacoepidemiology and Drug Safety. 2024;33(1):e5701.
- 140. Abrahamowicz M, Bartlett G, Tamblyn R, du Berger R. Modeling cumulative dose and exposure duration provided insights regarding the associations between benzodiazepines and injuries. Journal of Clinical Epidemiology. 2006 Apr 1;59(4):393–403.
- 141. Sylvestre MP, Abrahamowicz M. Flexible modeling of the cumulative effects of timedependent exposures on the hazard. Statistics in Medicine. 2009;28(27):3437–53.
- 142. Abrahamowicz M, Beauchamp M, Sylvestre M. Comparison of alternative models for linking drug exposure with adverse effects. Statistics in Medicine. 2012 May 20;31(11– 12):1014–30.
- 143. Sylvestre MP, Beauchamp ME, Kyle RP, Abrahamowicz M. WCE: Weighted Cumulative Exposure Models [Internet]. 2018 [cited 2023 Aug 24]. Available from: https://cran.rproject.org/web/packages/WCE/index.html

- 144. Sylvestre MP, Abrahamowicz M, Čapek R, Tamblyn R. Assessing the cumulative effects of exposure to selected benzodiazepines on the risk of fall-related injuries in the elderly. International Psychogeriatrics. 2012 Apr;24(4):577–86.
- 145. Shrier I. Understanding Causal Inference: The Future Direction in Sports Injury Prevention. Clinical Journal of Sport Medicine. 2007 May;17(3):220–4.
- 146. Nielsen RO, Bertelsen ML, Møller M, Hulme A, Windt J, Verhagen E, et al. Training load and structure-specific load: applications for sport injury causality and data analyses. Br J Sports Med. 2018 Aug 1;52(16):1016–7.
- 147. West SW, Clubb J, Torres-Ronda L, Howells D, Leng E, Vescovi JD, et al. More than a Metric: How Training Load is Used in Elite Sport for Athlete Management. Int J Sports Med. 2021 Apr;42(4):300–6.
- Hernán MA, Hernández-Díaz S. Beyond the intention to treat in comparative effectiveness research. Clinical trials (London, England). 2012 Feb;9(1):48.
- Platt RW, Hutcheon JA, Suissa S. Immortal Time Bias in Epidemiology. Curr Epidemiol Rep. 2019 Mar 1;6(1):23–7.
- 150. Hernán MA, Sauer BC, Hernández-Díaz S, Platt R, Shrier I. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. Journal of Clinical Epidemiology. 2016 Nov 1;79:70–5.
- Stovitz SD, Verhagen E, Shrier I. Distinguishing between causal and non-causal associations: implications for sports medicine clinicians. Br J Sports Med. 2019 Apr 1;53(7):398–9.
- 152. Hernán MA, Wang W, Leaf DE. Target Trial Emulation: A Framework for Causal Inference From Observational Data. JAMA. 2022 Dec 27;328(24):2446–7.
- 153. Rehkopf DH, Glymour MM, Osypuk TL. The Consistency Assumption for Causal Inference in Social Epidemiology: When a Rose is Not a Rose. Curr Epidemiol Rep. 2016 Mar;3(1):63–71.
- 154. Hernán M, Robins J. Causal Inference: What If [Internet]. Boca Raton: Chapman & Hall/CRC; 2020 [cited 2021 Oct 6]. Available from: https://www.hsph.harvard.edu/miguel-hernan/causal-inference-book/

- 155. Thomas D, Stram D, Dwyer J. Exposure Measurement Error: Influence on Exposure-Disease Relationships and Methods of Correction. Annual Review of Public Health. 1993 May 1;14(Volume 14, 1993):69–93.
- 156. Brown DM, Picciotto S, Costello S, Neophytou AM, Izano MA, Ferguson JM, et al. The Healthy Worker Survivor Effect: Target Parameters and Target Populations. Curr Environ Health Rep. 2017 Sep;4(3):364–72.
- 157. Hernán MA. How to estimate the effect of treatment duration on survival outcomes using observational data. BMJ: British Medical Journal [Internet]. 2018 [cited 2024 Apr 2];360. Available from: https://www.jstor.org/stable/26959303
- 158. Xiao Y, Abrahamowicz M, Moodie EEM, Weber R, Young J. Flexible Marginal Structural Models for Estimating the Cumulative Effect of a Time-Dependent Treatment on the Hazard: Reassessing the Cardiovascular Risks of Didanosine Treatment in the Swiss HIV Cohort Study. Journal of the American Statistical Association. 2014;109(506):455–64.
- 159. Emery C, Palacios-Derflingher L, Black AM, Eliason P, Krolikowski M, Spencer N, et al. Does disallowing body checking in non-elite 13- to 14-year-old ice hockey leagues reduce rates of injury and concussion? A cohort study in two Canadian provinces. Br J Sports Med. 2020 Apr 1;54(7):414–20.
- 160. Roberts AW, Dusetzina SB, Farley JF. Revisiting the washout period in the incident user study design: why 6–12 months may not be sufficient. J Comp Eff Res. 2015 Jan;4(1):27–35.
- 161. Zhu Y, Hubbard RA, Chubak J, Roy J, Mitra N. Core Concepts in Pharmacoepidemiology: Violations of the Positivity Assumption in the Causal Analysis of Observational Data: Consequences and Statistical Approaches. Pharmacoepidemiol Drug Saf. 2021 Nov;30(11):1471–85.
- 162. Danieli C, Sheppard T, Costello R, Dixon WG, Abrahamowicz M. Modeling of cumulative effects of time-varying drug exposures on within-subject changes in a continuous outcome. Stat Methods Med Res. 2020 Sep 1;29(9):2554–68.
- Naimi AI, Cole SR, Kennedy EH. An Introduction to G Methods. Int J Epidemiol. 2016 Dec 30;dyw323.

- 164. Naimi AI, Moodie EEM, Auger N, Kaufman JS. Constructing Inverse Probability Weights for Continuous Exposures: A Comparison of Methods. Epidemiology. 2014 Mar;25(2):292.
- 165. Hamilton G, Meeuwisse W, Emery C, Shrier I. Subsequent Injury Definition, Classification, and Consequence. Clinical Journal of Sport Medicine. 2011 Nov 1;21(6):508–14.
- Stares JJ, Dawson B, Peeling P, Heasman J, Rogalski B, Fahey-Gilmour J, et al.
 Subsequent Injury Risk Is Elevated Above Baseline After Return to Play: A 5-Year
 Prospective Study in Elite Australian Football. Am J Sports Med. 2019 Jul 1;47(9):2225–31.
- Amorim LD, Cai J. Modelling recurrent events: a tutorial for analysis in epidemiology. Int J Epidemiol. 2015 Feb 1;44(1):324–33.
- 168. Su CL, Steele R, Shrier I. Doubly robust estimation and causal inference for recurrent event data. Statistics in Medicine. 2020;39(17):2324–38.
- Greenland S. An introduction to instrumental variables for epidemiologists. Int J Epidemiol. 2000 Aug 1;29(4):722–9.
- 170. Lousdal ML. An introduction to instrumental variable assumptions, validation and estimation. Emerg Themes Epidemiol. 2018 Jan 22;15(1):1.